UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

	Form	10-Q
☑ QU 193		3 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF
	For the quarterly period	d ended March 31, 2008
	C	Or .
□ TR		3 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF
	Commission File	e No.: 001-16753
		RE SERVICES, INC. as Specified in Its Charter)
	Delaware (State or Other Jurisdiction of Incorporation or Organization)	06-1500476 (I.R.S. Employer Identification No.)
	12400 High Bluff Drive, Suite 100 San Diego, California (Address of principal executive offices)	92130 (Zip Code)
	Registrant's Telephone Number, In	cluding Area Code: (866) 871-8519
during the p	ate by check mark whether the registrant: (1) has filed all reports requireceding 12 months (or for such shorter period that the registrant was s for the past 90 days. Yes \boxtimes No \square	nired to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 required to file such reports), and (2) has been subject to such filing
	ate by check mark whether the registrant is a large accelerated filer, a of "large accelerated filer", "accelerated filer" and "smaller reporting	n accelerated filer, a non-accelerated filer, or a smaller reporting company. Secompany" in Rule 12b-2 of the Exchange Act. (Check one):
	Large accelerated filer $oximes$ Accelerated filer $oximes$ N	on-accelerated filer \Box
Indica	ate by check mark whether the registrant is a shell company (as define	ed in Rule 12b-2 of the Exchange Act). Yes \square No \boxtimes
As of	May 7, 2008, there were 33,832,315 shares of common stock, \$0.01	par value, outstanding.
=		

AMN HEALTHCARE SERVICES, INC. TABLE OF CONTENTS

item		Page
	PART I—FINANCIAL INFORMATION	
1.	Condensed Consolidated Financial Statements (unaudited):	3
	Condensed Consolidated Balance Sheets, As of March 31, 2008 and December 31, 2007	3
	Condensed Consolidated Statements of Operations, For the Three Months Ended March 31, 2008 and 2007	4
	Condensed Consolidated Statement of Stockholders' Equity and Comprehensive Income, For the Three Months Ended March 31, 2008	5
	Condensed Consolidated Statements of Cash Flows, For the Three Months Ended March 31, 2008 and 2007	6
	Notes to Condensed Consolidated Financial Statements	7
2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	14
3.	Quantitative and Qualitative Disclosures About Market Risk	25
4.	Controls and Procedures	25
	PART II—OTHER INFORMATION	
6.	<u>Exhibits</u>	26
	Signatures	27

PART I—FINANCIAL INFORMATION

Item 1. Condensed Consolidated Financial Statements

AMN HEALTHCARE SERVICES, INC.

CONDENSED CONSOLIDATED BALANCE SHEETS (Unaudited and in thousands, except par value)

	March 31, 2008	December 31, 2007
ASSETS		
Current assets:		
Cash and cash equivalents	\$ 7,835	\$ 18,495
Accounts receivable, net of allowance of \$4,802 and \$3,605 at March 31, 2008 and December 31, 2007, respectively	199,091	184,741
Prepaid expenses	12,820	9,583
Deferred income taxes, net	28,081	28,084
Other current assets	2,118	2,048
Total current assets	249,945	242,951
Fixed assets, net	25,246	24,600
Deposits and other assets	11,198	11,274
Goodwill, net	252,823	241,266
Intangible assets, net	126,418	113,535
Total assets	\$ 665,630	\$ 633,626
LIABILITIES AND STOCKHOLDERS' EQUITY		
Current liabilities:		
Bank overdraft	\$ 6,617	\$ —
Accounts payable and accrued expenses	28,624	22,231
Accrued compensation and benefits	47,328	43,446
Income taxes payable	3,159	2,925
Revolving credit facility	13,500	_
Current portion of notes payable	15,339	26,616
Deferred revenue	8,322	7,647
Other current liabilities	25,652	25,691
Total current liabilities	148,541	128,556
Notes payable, less current portion	117,628	120,352
Deferred income taxes, net	70,637	71,092
Other long-term liabilities	41,806	37,498
Total liabilities	378,612	357,498
Commitments and contingencies		
Subsequent events (Note 8)		
Stockholders' equity:		
Common stock, \$0.01 par value; 200,000 shares authorized; 45,446 shares issued at each of March 31, 2008 and		
December 31, 2007	454	454
Additional paid-in capital	399,360	397,237
Treasury stock, at cost (11,615 shares at each March 31, 2008 and December 31, 2007)	(201,692)	(201,692)
Retained earnings	90,552	81,036
Accumulated other comprehensive loss	(1,656)	(907)
Total stockholders' equity	287,018	276,128
Total liabilities and stockholders' equity	\$ 665,630	\$ 633,626

AMN HEALTHCARE SERVICES, INC.

CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (Unaudited and in thousands, except per share amounts)

		Three Months En March 31,			
		2008		2007	
Revenue	\$	293,593	\$	283,944	
Cost of revenue	_	216,138		211,439	
Gross profit		77,455		72,505	
Operating expenses:					
Selling, general and administrative		55,103		53,051	
Depreciation and amortization		3,350		2,629	
Total operating expenses		58,453		55,680	
Income from operations		19,002		16,825	
Interest expense, net		2,811		3,334	
Income before income taxes		16,191		13,491	
Income tax expense		6,675		5,297	
Net income	\$	9,516	\$	8,194	
Net income per common share:					
Basic	\$	0.28	\$	0.24	
Diluted	\$	0.28	\$	0.23	
Weighted average common shares outstanding:					
Basic		33,830		34,638	
Diluted	<u> </u>	34,180	_	35,283	

AMN HEALTHCARE SERVICES, INC.

CONDENSED CONSOLIDATED STATEMENT OF STOCKHOLDERS' EQUITY AND COMPREHENSIVE INCOME

Three Months Ended March 31, 2008 (Unaudited and in thousands)

	Common Stock		Additional Paid-in	Treasury	Retained	Accumulated Other Comprehensive	
	Shares	Amount	Capital	Stock	Earnings	Income (Loss)	Total
Balance, December 31, 2007	45,446	\$ 454	\$ 397,237	\$ (201,692)	\$ 81,036	\$ (907)	\$ 276,128
Stock-based compensation	_	_	2,123	_	_	_	2,123
Comprehensive income (loss):							
Foreign currency translation adjustment	_	_	_	_	_	(66)	(66)
Unrealized loss for derivative financial instruments, net of							
tax	_	_	_	_	_	(683)	(683)
Net income	_	_		_	9,516	_	9,516
Total comprehensive income							8,767
Balance, March 31, 2008	45,446	\$ 454	\$ 399,360	\$ (201,692)	\$ 90,552	\$ (1,656)	\$ 287,018

AMN HEALTHCARE SERVICES, INC.

CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited and in thousands)

		Months ded ch 31.
	2008	2007
Cash flows from operating activities:		
Net income	\$ 9,516	\$ 8,194
Adjustments to reconcile net income to net cash provided by operating activities:		
Depreciation and amortization	3,350	2,629
Provision for bad debts	1,435	588
Non-cash interest expense	420	485
Provision for deferred income taxes	_	892
Stock-based compensation	2,123	1,795
Loss on disposal or sale of fixed assets	50	24
Excess tax benefit from stock options exercised		(3,132)
Changes in assets and liabilities, net of effects from acquisition:		
Accounts receivable	(7,559)	4,203
Prepaid expenses and other current assets	(3,092)	(1,541)
Deposits and other assets	(284)	(1,082)
Accounts payable and accrued expenses	5,981	1,681
Accrued compensation and benefits	3,214	(1,222)
Income taxes payable Other liabilities	237	3,522 206
	1,482	
Net cash provided by operating activities	16,873	17,242
Cash flows from investing activities:		
Purchase and development of fixed assets	(2,612)	(2,131)
Cash paid for acquisition	(30,789)	
Net cash used in investing activities	(33,401)	(2,131)
Cash flows from financing activities:		
Capital lease repayments	(163)	(157)
Payments on notes payable	(14,001)	(3,860)
Proceeds from revolving credit facility	25,000	_
Payments on revolving credit facility	(11,500)	_
Proceeds from stock options exercised	_	1,066
Excess tax benefit from stock options exercised	_	3,132
Change in bank overdraft, net of overdraft acquired	6,598	(9,485)
Net cash provided by (used in) financing activities	5,934	(9,304)
Effect of exchange rate changes on cash	(66)	(8)
Net increase (decrease) in cash and cash equivalents	(10,660)	5,799
Cash and cash equivalents at beginning of period	18,495	4,422
Cash and cash equivalents at end of period	\$ 7,835	\$ 10,221
•	4 7,033	\$ 10,221
Supplemental disclosures of cash flow information:		
Cash paid for interest (net of \$59 and \$64 capitalized for the three months ended March 31, 2008 and 2007, respectively)	<u>\$ 2,541</u>	\$ 2,996
Cash paid for income taxes (excluding \$0 and \$170 cash received during the three months ended March 31, 2008 and 2007, respectively)	\$ 6,422	\$ 1,054
Supplemental disclosures of non-cash investing and financing activities:		
Suppensional usussimes of non-cash investing and mainting activities. Fixed assets acquired through capital leases	\$ 26	\$ 46
. • • .		
Net change in foreign currency translation adjustment and unrealized loss on derivative financial instruments, net of tax	<u>\$ (749)</u>	<u>\$ (265)</u>
Fair value of assets acquired in acquisitions, net of cash received	\$ 8,778	\$ —
Goodwill	11,557	
Intangible assets	13,960	_
Liabilities assumed	(1,004)	_
Excess net working capital payable	(166)	_
Holdback provision	(2,336)	
Net cash paid for acquisition	\$ 30,789	<u>\$</u>

AMN HEALTHCARE SERVICES, INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

1. BASIS OF PRESENTATION

The condensed consolidated balance sheets and related condensed consolidated statements of operations, stockholders' equity and comprehensive income and cash flows contained in this Quarterly Report on Form 10-Q, which are unaudited, include the accounts of AMN Healthcare Services, Inc. (the "Company") and its wholly-owned subsidiaries. All significant intercompany balances and transactions have been eliminated in consolidation. In the opinion of management, all entries necessary for a fair presentation of such condensed consolidated financial statements have been included. These entries consist only of normal recurring items. The results of operations for the interim period are not necessarily indicative of the results to be expected for any other interim period or for the entire fiscal year.

The condensed consolidated financial statements do not include all information and notes necessary for a complete presentation of financial position, results of operations and cash flows in conformity with United States generally accepted accounting principles. Please refer to the Company's audited consolidated financial statements and the related notes for the year ended December 31, 2007, contained in the Company's Annual Report on Form 10-K as filed with the Securities and Exchange Commission (the "SEC").

The preparation of financial statements in conformity with United States generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results may differ from those estimates.

2. STOCK-BASED COMPENSATION

The Company adopted the provisions of Financial Accounting Standards Board (FASB) revised Statement of Financial Accounting Standards (SFAS) No. 123 (SFAS No. 123R), *Share-Based Payment*, on January 1, 2006. Under SFAS No. 123R, stock-based compensation cost is measured at the grant date, based on the estimated fair value of the award, and is recognized as expense over the employee's requisite service period.

For the three months ended March 31, 2008, approximately 1,000 stock appreciation rights ("SARs") and approximately 1,000 restricted stock units ("RSUs") were granted to a new director of the Company under the AMN Healthcare Equity Plan ("Equity Plan"). The Company did not issue options during the three months ended March 31, 2008, nor did the Company issue stock options, SARs or RSUs during the three months ended March 31, 2007. The Equity Plan expires on the tenth anniversary of the effective date.

Stock Options and SARs

Stock options entitle the holder to purchase, at the end of a vesting period, a specified number of shares of the Company's common stock at a price per share set at the date of grant. SARs entitle the holder to receive, at the end of a vesting period, shares of the Company's common stock equal to the difference between the exercise price of the SAR, which is set at the date of grant, and the Fair Market Value, as defined in the Company's Equity Plan, of the Company's common stock on the date of exercise. For non-directors, the SARs typically vest ratably over a three year period, with about one third of the awards vesting annually.

Stock-based compensation cost of stock options and SARs is measured based on several criteria including, but not limited to, the valuation model used and associated input factors such as expected term of the award, stock price volatility, dividend rate, risk-free interest rate and award forfeiture rate. The input factors to be used in the valuation model are based on subjective future expectations combined with management judgment. The

Company estimates the fair value of awards granted using the Black-Scholes valuation model. After consideration of both its implied volatility and historical volatility, the Company determined its historical volatility to be the most accurate estimate of future volatility and therefore utilizes this measure. The Company estimates the expected term based on historical exercise patterns, and based the dividend yield assumption on historical dividend payouts, which is zero. The risk-free interest rate assumption is based on observed interest rates appropriate for the expected term of the Company's equity awards. The Company uses historical data to estimate pre-vesting award forfeitures and records stock-based compensation expense only for those awards that are expected to vest. The fair value of awards granted is amortized on a straight-line basis over the requisite service periods of the awards, which are the vesting periods.

Stock-based compensation expense for the three months ended March 31, 2008 for SARs granted was estimated at the date of grant using the Black-Scholes valuation model based on the following assumptions (annualized percentages):

	Three Months Ended March 31, 2008
Expected life	2.0 years
Risk-free interest rate	1.8%
Volatility	30%
Dividend yield	0%

The grant date fair value of the approximately 1,000 SARs granted during the three months ended March 31, 2008 was \$2.74 per SAR. Additionally, there was \$4.8 million of pre-tax total unrecognized compensation cost related to non-vested stock options and SARs at March 31, 2008, which will be adjusted for future changes in forfeitures. The Company expects to recognize such cost over a weighted average period of 1.4 years.

The following table summarizes stock options and SARs activity for the three months ended March 31, 2008:

	Number of Shares (in thousands)	Weighted Average Exercise Price	Weighted Average Remaining Contractual Term (years)	Aggregate Intrinsic Value (in thousands)
Outstanding at January 1, 2008	3,080	\$ 17.22		
Granted	1	\$ 14.95		
Exercised	_	\$ —		
Cancelled/forfeited/expired	(15)	\$ 22.13		
Outstanding at March 31, 2008	3,066	\$ 17.20	6.5	\$ 2,636
Exercisable at March 31, 2008	1,843	\$ 16.40	5.7	\$ 2,202

Restricted Stock Units

RSUs, granted under the Company's Equity Plan, entitle the holder to receive, at the end of a vesting period, a specified number of shares of the Company's common stock. Stock-based compensation cost of RSUs is measured by the market value of the Company's common stock on the date of grant. For non-directors, the RSUs typically vest at the end of a three year vesting period, however, 33% of the awards may vest on the 13th month anniversary of the grant date, and 34% on the 2nd anniversary of the grant date, if certain performance targets are met. The grant date intrinsic value was \$14.95 per RSU for the approximately 1,000 RSUs granted during the three months ended March 31, 2008. The Company uses historical data to estimate pre-vesting award forfeitures and records stock-based compensation expense only for those awards that are expected to vest. The grant date intrinsic value of awards granted is amortized on a straight-line basis over the requisite service periods of the awards, which are the vesting periods.

The following table summarizes RSUs activity for non-vested awards for the three months ended March 31, 2008:

	Number of Shares	Weig	hted Average
	(in thousands)	Grant Da	te Intrinsic Value
Unvested at January 1, 2008	408	\$	21.94
Granted	1	\$	14.95
Vested	0	\$	_
Cancelled/forfeited/expired	(6)	\$	22.73
Unvested at March 31, 2008	403	\$	21.91

As of March 31, 2008, there was \$5.8 million of pre-tax total unrecognized compensation cost related to non-vested RSUs, which will be adjusted for future changes in forfeitures. The Company expects to recognize such cost over a period of 1.8 years.

Stock-Based Compensation under SFAS No. 123R

Compensation cost of stock options and SARs is measured using the Black-Scholes valuation model, and compensation cost of RSUs is measured by the market value of the Company's common stock on the date of grant. The Company uses historical data to estimate pre-vesting award forfeitures and records stock-based compensation expense only for those awards that are expected to vest. The following table shows the total stock-based compensation expense, related to all of the Company's equity awards, recognized for the three month periods ended March 31, 2008 and 2007, in accordance with SFAS No. 123R (in thousands):

	Three Mor Marc	iths Ended ch 31,
	2008	2007
Stock-based employee compensation before tax	\$ 2,123	\$ 1,795
Related income tax benefits	(875)	(705)
Stock-based employee compensation, net of tax	\$ 1,248	\$ 1,090

For the three months ended March 31, 2008 and 2007, exercises of stock-based compensation under SFAS No. 123R resulted in cash flows from financing activities of \$0 and \$3,132,000, respectively, for excess tax benefits related to stock options exercised.

3. NET INCOME PER COMMON SHARE

Basic net income per common share is calculated by dividing net income by the weighted average number of common shares outstanding during the reporting period. Diluted net income per common share reflects the effects of potentially dilutive stock-based awards.

Stock-based awards to purchase 1,205,000 and 2,000 shares of common stock for the three month periods ended March 31, 2008 and 2007, respectively, were not included in the calculations of diluted net income per common share because the effect of these instruments was anti-dilutive. The following table sets forth the computation of basic and diluted net income per common share for the periods ended March 31, 2008 and 2007 (in thousands, except per share amounts):

		nths Ended ch 31,
	2008	2007
Net income	\$ 9,516	\$ 8,194
Weighted average common shares outstanding—basic	33,830	34,638
Plus dilutive equity awards	350	645
Weighted average common shares outstanding—diluted	34,180	35,283
Net income per common share—basic	\$ 0.28	\$ 0.24
Net income per common share—diluted	\$ 0.28	\$ 0.23

4. GOODWILL AND IDENTIFIABLE INTANGIBLE ASSETS

As of March 31, 2008 and December 31, 2007, the Company had the following intangible assets (in thousands):

		March 31, 2008						December 31, 2007						
		Gross Carrying Amount		Carrying		ccumulated nortization	Net Carrying Amount		Gross Carrying Amount		Accumulated Amortization			Net Carrying Amount
Intangible assets subject to amortization:														
Staffing databases	\$	2,430	\$	(877)	\$	1,553	\$	1,890	\$	(752)	\$	1,138		
Customer relationships		36,400		(5,859)		30,541		34,100		(5,166)		28,934		
Tradenames and trademarks		13,511		(715)		12,796		2,711		(565)		2,146		
Noncompete agreements		1,430		(506)		924		1,110		(442)		668		
Acquired technology		800		(138)		662		800		(98)		702		
Online courses		59		(17)		42		59		(12)		47		
	\$	54,630	\$	(8,112)	\$	46,518	\$	40,670	\$	(7,035)	\$	33,635		
Intangible assets not subject to amortization:														
Goodwill (1)	\$	263,275	\$	(10,452)	\$	252,823	\$	251,718	\$	(10,452)	\$	241,266		
Tradenames and trademarks		79,900		_		79,900		79,900		_		79,900		
	\$	343,175	\$	(10,452)	\$	332,723	\$	331,618	\$	(10,452)	\$	321,166		

⁽¹⁾ Goodwill accumulated amortization represents amortization expense recorded prior to the Company's adoption of SFAS No. 142, *Goodwill and Other Intangible Assets*, on January 1, 2002.

The carrying amount of goodwill was \$252,823,000 and \$241,266,000 as of March 31, 2008 and December 31, 2007, respectively. The increase in goodwill was due to the asset acquisition of Platinum Select, L.P. ("Platinum Select Staffing") in February 2008.

The carrying amount of intangible assets was \$126,418,000 and \$113,535,000 as of March 31, 2008 and December 31, 2007, respectively. The increases in intangible assets were due primarily to the asset acquisition of Platinum Select Staffing in February 2008.

Aggregate amortization expense for the intangible assets presented in the above table was \$1,077,000 and \$760,000 for the three months ended March 31, 2008 and 2007, respectively. Estimated future aggregate amortization expense of intangible assets as of March 31, 2008 is as follows (in thousands):

Amount
\$ 3,608
4,801
4,605
3,780
3,433
$\frac{26,291}{$46,518}$
\$46,518

5. SEGMENT INFORMATION

SFAS No. 131, *Disclosures about Segments of an Enterprise and Related Information*, establishes annual and interim reporting standards for an enterprise's operating segments and related disclosures about its products, services, geographic areas and major customers. An operating segment is defined as a component of an enterprise that engages in business activities from which it may earn revenue and incur expenses, and for which discrete financial information is regularly evaluated by the chief operating decision maker in deciding how to allocate resources and assess performance.

The Company provides hospital and healthcare facilities with temporary staffing for physicians, nurses and allied healthcare professionals and physician permanent placement services through the use of several brand names, each having their own marketing and supply distinction. The Company's operating segments are identified in the same manner as they are reported internally and used by the Company's chief operating decision maker for the purposes of evaluating performance and allocating resources.

The Company's management relies on internal management reporting processes that provide revenue and segment operating income for making financial decisions and allocating resources. Segment operating income includes income from operations before depreciation, amortization of intangible assets and amortization of stock compensation expense. Management believes that segment operating income is an appropriate measure of evaluating the operational performance of the Company's segments. However, this measure should be considered in addition to, not as a substitute for, or superior to, income from operations or other measures of financial performance prepared in accordance with generally accepted accounting principles. The Company's management does not evaluate, manage or measure performance of segments using asset information; accordingly, asset information by segment is not prepared or disclosed. The information in the following table is derived from the segment's internal financial information as used for corporate management purposes.

The following table provides a reconciliation of revenue and segment operating income by reportable segment to consolidated results (in thousands):

		onths Ended rch 31,
	2008	2007
Revenue:		
Nurse and allied healthcare staffing	\$ 203,985	\$ 199,975
Locum tenens staffing	76,353	71,405
Physician permanent placement services	13,255	12,564
	\$ 293,593	\$ 283,944
Segment operating income:		
Nurse and allied healthcare staffing	\$ 15,481	\$ 13,731
Locum tenens staffing	5,655	4,018
Physician permanent placement services	3,339	3,500
	24,475	21,249
Depreciation and amortization	3,350	2,629
Stock-based compensation	2,123	1,795
Interest expense, net	2,811	3,334
Income before income taxes	\$ 16,191	\$ 13,491

6. ACQUISITION

On February 15, 2008, the Company acquired substantially all of the assets of Platinum Select Staffing, a national travel allied staffing firm. The strategic combination has further broadened the type of services the Company offers. The total initial purchase price was \$33.3 million, which included \$30.4 million cash consideration, \$2.3 million cash holdback, \$0.4 million of direct transaction costs and \$0.2 million excess net working capital payable. The acquisition agreement also provides for an annual performance-based contingent earn-out payment up to \$17 million to be paid in 2009. The allocation of the purchase price consisted of \$7.8 million of net assets acquired, \$11.5 million of goodwill and \$14.0 million of identified intangible assets. The intangible assets included the fair value of trademarks, tradenames, customer relationships, non-compete agreements and staffing databases. The weighted average useful life of the acquired intangible assets subject to amortization is approximately 17 years. The goodwill was allocated to the Company's nurse and allied healthcare staffing segment. The acquisition was recorded using the purchase method of accounting. The results of operations of Platinum Select Staffing are included in the nurse and allied healthcare staffing segment in the Company's consolidated financial statements from the acquisition date.

7. FAIR VALUE MEASUREMENT

In September 2006, the Financial Accounting Standards Board (FASB) issued Statement of Financial Accounting Standards No. 157, *Fair Value Measurements* (SFAS 157). SFAS 157 introduces a framework for measuring fair value and expands required disclosure about fair value measurements of assets and liabilities. SFAS 157 for financial assets and liabilities is effective for fiscal years beginning after November 15, 2007. The Company has adopted the standard for those assets and liabilities as of January 1, 2008 and the impact of adoption was not significant. In November 2007, the FASB placed a one year deferral for the implementation of SFAS 157 for nonfinancial assets and liabilities. Accordingly, the Company will adopt the methods of fair value described in SFAS 157 for nonfinancial assets and liabilities on January 1, 2009.

SFAS 157 defines fair value as the exchange price that would be received for an asset or paid to transfer a liability (an exit price) in the principal or most advantageous market for the asset or liability in an orderly

transaction between market participants on the measurement date. SFAS 157 also establishes a fair value hierarchy which requires an entity to maximize the use of observable inputs and minimize the use of unobservable inputs when measuring fair value. The standard describes three levels of inputs that may be used to measure fair value:

Level 1—Quoted prices in active markets for identical assets or liabilities.

Level 2—Observable inputs other than Level 1 prices such as quoted prices for similar assets or liabilities; quoted prices in markets that are not active; or other inputs that are observable or can be corroborated by observable market data for substantially the full term of the assets or liabilities.

Level 3—Unobservable inputs that are supported by little or no market activity and that are significant to the fair value of the assets or liabilities.

The Company utilizes the market approach to measure fair value for its financial assets and liabilities. The market approach uses prices and other relevant information generated by market transactions involving identical or comparable assets or liabilities. As of March 31, 2008, the Company held certain assets and liabilities that are required to be measured at fair value on a recurring basis. These included the Company's investments associated with the Company's Executive Nonqualified Excess Plan ("Excess Benefit Plan"), and interest rate swaps. The Company's investments associated with its Excess Benefit Plan consist of mutual funds that are publicly traded and for which market prices are readily available. The Company's interest rate swaps are valued using commonly quoted intervals from observable market. In addition, the Company discounts the derivative liabilities to reflect the potential credit risk to lenders by using current interest rates available to the Company which were obtained directly from the Company's third-party lender.

Financial assets and liabilities measured at fair value on a recurring basis are summarized below (in thousands):

		Fair Value Measuremer	nts as of March 31, 2008	
		Quoted Prices in Active Markets for Identical	Significant Other Observable	Significant Unobservable
	Total	Assets (Level 1)	Inputs (Level 2)	Inputs (Level 3)
Financial Assets: Trading securities investment	\$1,480	\$ 1,480	\$ —	\$ —
Financial Liabilities: Interest rate swaps	\$2.212	\$ —	\$ 2,212	\$ —

8. SUBSEQUENT EVENTS

Effective May 2, 2008, the Company entered into a Third Amendment (the "Third Amendment") to its Second Amended and Restated Credit Agreement (the "Credit Agreement") dated November 2, 2005. The Third Amendment provides that the Company may repurchase its common stock in an aggregate cash amount not to exceed \$50 million, provided that no default or event of default exists either before or after giving effect to such repurchase payment and such repurchase occurs on or before March 31, 2009. The Third Amendment also adds a \$15 million secured debt basket to the permitted indebtedness exceptions (subject to certain conditions) and increases the annual capital expenditure basket to \$15 million.

On May 7, 2008, the Company announced that it intends to repurchase up to \$38 million of its outstanding common stock in the open market through March 31, 2009. Under the repurchase program, share purchases may be made from time to time beginning in May 2008 depending on prevailing market conditions and other considerations. The Company expects to use available cash on hand and its revolver to fund the purchase of shares, and repurchased shares will be held in treasury.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following discussion should be read in conjunction with, and is qualified in its entirety by, our consolidated financial statements and the notes thereto and other financial information included elsewhere herein and in our Annual Report on Form 10-K for the year ended December 31, 2007. Certain statements in this "Management's Discussion and Analysis of Financial Condition and Results of Operations" are "forward-looking statements." See "Special Note Regarding Forward-Looking Statements." We undertake no obligation to update the forward-looking statements in this filing. References in this filing to "AMN Healthcare," the "Company," "we," "us" and "our" refer to AMN Healthcare Services, Inc. and its wholly owned subsidiaries.

Overview

We are the largest healthcare staffing company in the United States. As the largest nationwide provider of travel nurse and allied staffing services, locum tenens (temporary physician staffing) and physician permanent placement services, we recruit physicians, nurses, and allied healthcare professionals, our "healthcare professionals", nationally and internationally and place them on assignments of variable lengths and in permanent positions at acute-care hospitals, physician practice groups and other healthcare settings, including rehabilitation centers, radiology imaging facilities, dialysis clinics and pharmacies throughout the United States.

For the three months ended March 31, 2008, we recorded revenue of \$293.6 million, as compared to revenue of \$283.9 million for the three months ended March 31, 2007. We recorded net income of \$9.5 million for the three months ended March 31, 2008, as compared to net income of \$8.2 million for the three months ended March 31, 2007.

Our staffing services are marketed to two distinct customer groups: (1) healthcare professionals and (2) hospitals, physician practice groups and other healthcare centers. We use distinct brands to market our differentiated services throughout the healthcare staffing spectrum.

We use a multi-brand recruiting strategy to enhance our ability to successfully attract healthcare professionals in the United States and internationally. We market our staffing opportunities to healthcare professionals under recruitment brands including American Mobile Healthcare®, Medical ExpressSM, NurseChoice®, NursesRx®, Preferred Healthcare Staffing®, Med TravelersSM, Platinum Select StaffingSM, Rx Pro HealthSM, RN Demand®, O'Grady Peyton International®, Staff Care® and Merritt, Hawkins & Associates®. Each brand has a distinct clinician focus, market strengths and brand reputation.

At the end of March 31, 2008, we had healthcare professionals on assignment at over 2,000 different healthcare facility clients. We provide staffing services mainly to acute care and sub-acute healthcare facilities, physician groups, dialysis centers, clinics, radiology imaging facilities and retail and mail-order pharmacies. We market our travel nursing services to hospitals and healthcare facilities generally under one brand, AMN Healthcare®, as a single staffing provider with access to healthcare professionals from several nurse recruitment brands. We market our locum tenens staffing and physician permanent placement services under the brand names Staff Care and Merritt, Hawkins & Associates, respectively, and market our allied healthcare staffing services to hospitals and healthcare facilities generally under the brand names Med Travelers, Platinum Select Staffing and Rx Pro Health. Rx Pro Health is a pharmacy staffing company that we acquired in May 2007, and Platinum Select Staffing is an allied healthcare staffing company, whose assets we acquired in February 2008.

Physicians, nurses and allied healthcare professionals join us on temporary assignments for a wide variety of reasons that include: seeking flexible work opportunities, exploring different areas of the country and diverse practice settings, building clinical skills and experience by working at prestigious healthcare facilities, avoiding the demands and political environment of working as permanent staff, working through life and career transitions, and as a means of access into a permanent staff position with our clients.

Our large number of hospital, healthcare facility and other clients provides us with the opportunity to offer temporary positions typically in all 50 states and in a variety of work environments and clinical settings. In addition, we provide our temporary healthcare professionals with an attractive benefits package that may include free or subsidized housing, free or reimbursed travel, competitive wages, professional development opportunities, professional liability insurance, and, for employed professionals, a 401(k) plan and health insurance. We believe that we attract temporary healthcare professionals due to our long-standing reputation for providing a high level of service, our numerous job opportunities, our benefit packages, our innovative marketing programs and word-of-mouth referrals from our thousands of current and former healthcare professionals.

Our clients include hospitals and healthcare systems such as Georgetown University Hospital, HCA, NYU Medical Center, Stanford Health Care, UCLA Medical Center, The University of Chicago Hospitals, and Mayo Health System. We also provide services to government facilities, sub-acute healthcare facilities, physician practice groups, dialysis centers, clinics, imaging facilities and pharmacies.

Our hospital and healthcare facility clients utilize our staffing services to cost-effectively manage both short and long-term shortages in their staff due to a variety of circumstances such as a lack of qualified, specialized local healthcare professionals, attrition, leave schedules, new unit openings, seasonal patient census variations and to identify candidates for permanent positions. In addition to providing continuity of care and quality patient care, we believe hospitals, healthcare facilities and other clients contract with us due to our access to a large national network of quality temporary and permanent healthcare professionals, our ability to meet their specific staffing needs, our flexible staffing assignment lengths and our reliable and superior customer service.

The physician permanent placement market has strong growth potential due to client demand for physicians that enable our clients to generate revenue. Physicians are significant drivers of revenue, influencing many hospitals, healthcare facilities and physician practice groups to devote the resources necessary to recruit them. While this demand creates significant competition for a limited supply of physicians, our national reach and brand recognition positions us well in this environment.

The strong demand and significant overlap of clients seeking physicians for both temporary and permanent placement assignments provide us an opportunity to cross-sell these complementary physician staffing services.

Recent Trends

The U. S. temporary healthcare staffing industry revenues grew by 6% in 2007 to \$11.2 billion according to self-reported growth estimates supplied by staffing companies to Staffing Industry Analysts, which was published in March 2008. Industry revenue estimates for the markets in which we provide temporary healthcare staffing services were \$2.4 billion for travel nursing, \$1.6 billion for locum tenens and \$3.2 billion for allied healthcare. The balance of the industry revenue estimate was attributable to per diem staffing, which is a market in which we do not participate.

Our supply of healthcare professionals continues to be constrained relative to the demands of our healthcare facility clients. We have experienced a shortage in healthcare professionals in the markets in which we provide temporary healthcare staffing service in that we have more demand than we can fill in each of these markets.

Our nurse and allied healthcare staffing segment, the Company's largest reporting segment, experienced a strong demand and tight supply environment during 2005 and early 2006. During the last quarter of 2006 and throughout 2007, the nurse and allied healthcare staffing industry generally experienced a lessening of demand in certain key states, such as California, which was partially offset by increased demand in other parts of the country. The lower demand in certain states was driven by several factors, such as relatively flat hospital admission levels, the aggressive hiring of new graduates and some improvement in hospitals' ability to retain permanent labor. More recently, we have seen significant demand improvement to levels similar to what we experienced last year at this time.

We recruit a small percentage of nurses from international channels through our O'Grady Peyton International (OGP) brand to meet our facility clients' long-term staffing needs. Internationally trained nurses have represented an important supply channel, comprising roughly 15% of newly licensed nurses in the United States since 2005. Revenue from OGP comprises less than 5% of our consolidated revenue. Our continued utilization of this international supply channel is reliant upon an increase or exemption from current permanent immigrant visa quotas. Since late 2006, the number of permanent immigrant visas available for nurses overseas has been limited as a result of these quotas, which has constrained our ability to place new international nurses on assignment. Recently, federal legislation has been introduced that would have the effect of increasing the permanent visas available for nurses.

In addition to these demand and supply dynamics, we have experienced fluctuations in direct costs, primarily for increased housing and related costs, such as utilities, associated with the placement of temporary nurse and allied professionals which have impacted our gross margins in this segment.

Locum tenens physicians are used by our hospital, healthcare facility and physician practice group clients to fill temporary vacancies due to vacations and leave schedules, and increasingly, to bridge staffing gaps while these clients seek permanent physician candidates. Staffing Industry Analysts estimated the locum tenens market to be the fastest-growing segment of the healthcare staffing industry in 2007. We believe that the demand environment for locum tenens services continues to be strong, and socioeconomic factors, such as increased physician emphasis on job and lifestyle flexibility, suggest a growing number of physicians are embracing locum tenens as a practice style.

The market for physician permanent placement services also continues to experience a steady demand environment that has contributed to moderate growth in this segment of our business over the past year. As with our locum tenens business, the supply of candidates is insufficient to meet the demands of our clients seeking a physician to join their practice or staff.

Critical Accounting Principles and Estimates

We have identified the following critical accounting policies that affect the more significant judgments and estimates used in the preparation of our unaudited condensed consolidated financial statements. The preparation of our financial statements in conformity with United States generally accepted accounting principles requires us to make estimates and judgments that affect our reported amounts of assets and liabilities, revenue and expenses, and related disclosures of contingent assets and liabilities. On an on-going basis we evaluate our estimates, including those related to asset impairment, accruals for self-insurance and compensation and related benefits, accounts receivable, contingencies and litigation, and valuation and recognition of share-based payments. We state these accounting policies in the notes to the audited financial statements for the year ended December 31, 2007, contained in our Annual Report on Form 10-K as filed with the Securities and Exchange Commission, and in relevant sections in this discussion and analysis. These estimates are based on the information that is currently available to us and on various other assumptions that we believe to be reasonable under the circumstances. Actual results could vary from these estimates under different assumptions or conditions.

We believe that the following critical accounting policies affect the more significant judgments and estimates used in the preparation of our unaudited condensed consolidated financial statements:

Goodwill and Intangible Assets

We have recorded goodwill and other indefinite lived intangible assets resulting from our past acquisitions. Commencing with the adoption of SFAS No. 142, *Goodwill and Other Intangible Assets*, on January 1, 2002, we ceased amortizing goodwill and have performed annual impairment analyses each year to assess the recoverability of the goodwill and other indefinite lived intangible assets, in accordance with the provisions of

SFAS No. 142. We performed the most recent annual impairment test on October 31, 2007 and determined that there was no impairment. No events have occurred subsequent to October 31, 2007 that would indicate impairment may have occurred.

SFAS No. 142 also requires that intangible assets with estimable useful lives continue to be amortized over their respective estimated useful lives and reviewed for impairment in accordance with SFAS No. 144, *Accounting for the Impairment or Disposal of Long-Lived Assets*. In accordance with the provisions of SFAS No. 144, we analyze our amortizable intangible assets for impairment annually as well as whenever events or changes in circumstances indicate that the carrying amount may not be recoverable.

If we are required to record an impairment charge in the future, it could have an adverse impact on our results of operations.

Professional Liability Reserve

We maintain an accrual for professional liability self-insured retention limits, which is included in accounts payable and accrued expenses and other long term liabilities in our consolidated balance sheets. We determine the adequacy of this accrual by evaluating our historical experience and trends, loss reserves established by our insurance carriers, management and third party administrators, as well as through the use of independent actuarial studies. For the nurse and allied healthcare staffing segment, we obtain updated actuarial studies on a regular basis that use our historical claims data and industry data to determine the appropriate reserves for incurred, but not reported, professional liability claims for each year, with reserves for reported claims based upon loss reserves established by management and our third party administrators. For the locum tenens staffing segment, we obtain updated actuarial studies on a regular basis that utilize industry data to determine the appropriate reserves for both reported claims and incurred, but not reported, claims.

Self-Insured Health Insurance Claims Reserve

We maintain an accrual for self-insured health benefits we provide to our corporate employees and temporary nurse and allied healthcare professionals, which is included in accrued compensation and benefits in our consolidated balance sheets. We determine the adequacy of this accrual by evaluating our historical experience and trends related to both health insurance claims and payments, information provided to us by our insurance broker and third party administrator and industry experience and trends. If such information indicates that our accruals are overstated or understated, we reduce or provide for additional accruals. Our accrual is based on (i) a monthly average of our actual historical health insurance claim amounts and (ii) the average period of time from the date the claim is incurred to the date that it is reported to us and paid. We believe this is the best estimate of the amount of incurred, but not reported, self-insured health benefit claims at year-end. Historically, our accrual for health insurance has been adequate to provide for incurred claims and has fluctuated with increases or decreases in the average number of plan participants, changes in our claims experience and changes in the reporting and payment processing time for claims.

Workers Compensation Reserve

We maintain an accrual for workers compensation self-insured retention limits, which is included in accrued compensation and benefits and other long term liabilities in our consolidated balance sheets. We determine the adequacy of these accruals by evaluating our historical experience and trends, loss reserves established by our insurance carriers and third party administrators, as well as through the use of independent actuarial studies. We obtain updated actuarial studies on a semi-annual basis that use our payroll and actual claims data, as well as industry data, to determine the appropriate reserve both for reported claims and incurred, but not reported, claims for each policy year. The actuarial study for workers compensation provides us with the estimated losses for prior policy years and an estimated percentage of payroll compensation to be accrued for the current year. We record our accruals based on the amounts provided in the actuarial study, and we believe this is the best estimate of our liability for reported claims and incurred, but not reported, claims.

Accounts Receivable

Accounts receivable are recorded at the invoiced amount and are non-interest bearing. We maintain an allowance for doubtful accounts for estimated credit losses resulting from collection risks, including the inability of our customers to make required payments under their contractual agreements. This results in a provision for bad debt expense. The allowance for doubtful accounts is reported as a reduction of accounts receivable in our consolidated balance sheets. We determine the adequacy of this allowance by evaluating historical delinquency and write-off trends, the financial condition and credit risk and histories of each customer, historical payment trends and current economic conditions. If the financial condition of our customers deteriorates, resulting in an impairment of their ability to make payments, additional allowances would be provided.

Contingent Liabilities

We are subject to various claims and legal actions in the ordinary course of our business. Some of these matters include tax, payroll and employee-related matters and investigations by governmental agencies regarding our employment practices. We are the subject of an assessment by the California Employment Development Department (EDD) with respect to the payment of certain payroll related taxes, in connection with our locum tenens staffing segment, for the period between April 1, 2001 and March 31, 2004. We have appealed the EDD's assessment. Although we have not received an assessment for the subsequent periods, we are the subject of an audit by the EDD with respect to these matters for the subsequent periods. As we become aware of such claims and legal actions, we provide accruals if the exposures are probable and estimable. If an adverse outcome of such claims and legal actions is reasonably possible, we assess materiality and provide disclosure, as appropriate. We may also become subject to claims, governmental inquiries and investigations and legal actions relating to services provided by our temporary healthcare professionals, and we maintain accruals for these matters if the amounts are probable and estimable. We are not currently aware of any such pending or threatened litigation that would be considered reasonably likely to have a material adverse effect on our consolidated financial position, results of operations or liquidity.

Share-Based Payments

Effective January 1, 2006, we adopted the provisions of revised SFAS No. 123 (SFAS No. 123R), *Share-Based Payment*, which established accounting for share-based awards exchanged for employee services and requires companies to expense the estimated fair value of these awards over the requisite employee service period. Accordingly, stock-based compensation cost is measured at grant date, based on the fair value of the award, and is recognized as expense over the employee's requisite service period. The measurement of stock based compensation cost is based on several criteria including, but not limited to, the valuation model used and associated input factors such as expected term of the award, our stock price volatility, dividend rate, risk free interest rate, and award forfeiture rate. The input factors to use in the valuation model are based on subjective future expectations combined with management judgment. We estimate the fair value of stock options and stock appreciation rights granted using the Black-Scholes valuation model and the assumptions shown in Note 2 to the accompanying consolidated financial statements. We use historical data to estimate pre-vesting equity award forfeitures and record stock-based compensation expense only for those awards that are expected to vest. We estimate the expected term, dividend yield, and risk-free interest rate assumptions consistent with the methodologies used prior to January 1, 2006. After consideration of both our implied volatility and historical volatility, we determined our historical volatility to be the most accurate estimate of future volatility and therefore utilize this measure. The fair value of equity awards granted is amortized on a straight-line basis over the requisite service periods of the awards, which are the vesting periods. If factors change, we may decide to use different assumptions under the Black-Scholes valuation model, or a different valuation model, in the future, which could materially affect our net income and earnings per share.

Results of Operations

The following table sets forth, for the periods indicated, certain statements of operations data as a percentage of revenue. Our results of operations include three reportable segments: (1) nurse and allied healthcare staffing; (2) locum tenens staffing; and (3) physician permanent placement services. Our historical results are not necessarily indicative of our results of operations to be expected in the future.

Three Months Ended

	I nree Monti March	
	2008	2007
Condensed Consolidated Statements of Operations:		
Revenue	100.0%	100.0%
Cost of revenue	73.6	74.5
Gross profit	26.4	25.5
Selling, general and administrative	18.8	18.7
Depreciation and amortization expense	1.1	0.9
Income from operations	6.5	5.9
Interest expense, net	1.0	1.1
Income before income taxes	5.5	4.8
Income tax expense	2.3	1.9
Net income	3.2%	2.9%

Comparison of Results for the Three Months Ended March 31, 2008 to the Three Months Ended March 31, 2007

Revenue. Revenue increased 3%, to \$293.6 million for the three months ended March 31, 2008 from \$283.9 million for the same period in 2007, primarily due to pricing increases in the locum tenens staffing segment and the acquisitions of Rx Pro Health in May 2007 and Platinum Select Staffing in February 2008 in the nurse and allied healthcare staffing segment.

Nurse and allied healthcare staffing segment revenue increased 2%, to \$204.0 million for the three months ended March 31, 2008 from \$200.0 million for the same period in 2007. Of the \$4.0 million increase, \$3.8 million was attributable to the increase in revenue generated per temporary healthcare professional due primarily to an increase in the average bill rates charged to hospital and healthcare facility clients, \$2.5 million was attributable to a combination of an extra billing day during the period and a shift in the mix of temporary healthcare professionals working on flat rate contracts to hours and days worked contracts, \$2.3 million was attributable to the acquisition of Rx Pro Health in May 2007 and \$7.1 million was attributable to the acquisition of Platinum Select Staffing in February 2008. These increases were partially offset by a \$11.7 million decrease due to a decrease in the average number of temporary healthcare professionals on assignment.

Locum tenens staffing segment revenue increased 7%, to \$76.3 million for the three months ended March 31, 2008 from \$71.4 million for the same period in 2007. Of the \$4.9 million increase, \$4.6 million was attributable to bill rate increases, \$0.5 million was attributable to an increase in the number of days filled by healthcare professionals during the three months ended March 31, 2008. These increases were partially offset by a \$0.2 million decrease due to a shift in the mix of healthcare specialties worked.

Physician permanent placement services segment revenue increased 6%, to \$13.3 million for the three months ended March 31, 2008 from \$12.6 million for the same period in 2007.

Cost of Revenue. Cost of revenue increased 2%, to \$216.1 million for the three months ended March 31, 2008 from \$211.4 million for the same period in 2007, primarily due to an increase in the compensation and housing costs provided to the healthcare professionals.

Nurse and allied healthcare staffing segment cost of revenue increased 1%, to \$155.0 million for the three months ended March 31, 2008 from \$153.4 million for the same period in 2007. Of the \$1.6 million increase, \$1.6 million was attributable to an extra billing day during the period, \$1.9 million was attributable to net increases in compensation, primarily related to wages and housing costs provided to our temporary healthcare professionals, \$0.2 million was attributable to a shift in the mix of temporary healthcare professionals working on flat rate contracts to hours and days worked contracts, \$1.3 million was attributable to the acquisition of Rx Pro Health in May 2007 and \$5.6 million was attributable to the acquisition of Platinum Select Staffing in February 2008. These increases were partially offset by a \$9.0 million decrease attributable to the decrease in the average number of temporary healthcare professionals on assignment.

Locum tenens staffing segment cost of revenue increased 5%, to \$56.0 million for the three months ended March 31, 2008 from \$53.4 million for the same period in 2007. Of the \$2.6 million increase, \$0.4 million was attributable to an increase in the number of days filled by healthcare professionals and \$2.2 million was attributable to an increase in compensation provided to our healthcare professionals.

Physician permanent placement services segment cost of revenue increased 11%, to \$5.1 million for the three months ended March 31, 2008 from \$4.6 million for the same period in 2007.

Gross Profit. Gross profit increased 7%, to \$77.5 million for the three months ended March 31, 2008 from \$72.5 million for the same period in 2007, representing gross margins of 26.4% and 25.5%, respectively. Gross margin by reportable segment for the three months ended March 31, 2008 and 2007 was 24.0% and 23.3% for nurse and allied healthcare staffing, 26.7% and 25.2% for locum tenens staffing and 61.6% and 63.2% for physician permanent placement services, respectively. The increase in the nurse and allied healthcare staffing segment gross margin was primarily attributable to a decrease in health insurance claims, and the increase in the locum tenens staffing segment gross margin was primarily attributable to widening of the pay-to-bill spread.

Selling, General and Administrative Expenses. Selling, general and administrative expenses increased 4%, to \$55.1 million for the three months ended March 31, 2008 from \$53.1 million for the same period in 2007. The increase was primarily attributable to the acquisitions of Rx Pro Health in May 2007 and Platinum Select Staffing in February 2008. These increases were partially offset by decreases in corporate insurance costs and travel related costs. Selling, general and administrative expenses by reportable segment for the three months ended March 31, 2008 and 2007, respectively, were \$35.3 million and \$34.5 million for nurse and allied healthcare staffing, \$14.8 million and \$14.1 million for locum tenens staffing and \$5.0 million and \$4.5 million for physician permanent placement services.

Depreciation and Amortization Expense. Amortization expense increased 38%, to \$1.1 million for the three months ended March 31, 2008 from \$0.8 million for the same period in 2007. The increase was attributable to amortization of intangibles acquired from the Rx Pro Health acquisition in May 2007 and the Platinum Select Staffing acquisition in February 2008. Depreciation expense increased 28%, to \$2.3 million for the three months ended March 31, 2008 from \$1.8 million for the same period in 2007. The increase was primarily attributable to additional internally developed software.

Interest Expense, Net. Interest expense, net, was \$2.8 million for the three months ended March 31, 2008 as compared to \$3.3 million for the same period in 2007. The \$0.5 million decrease was primarily attributable to the \$23.1 million decrease in debt outstanding from March 31, 2007 to March 31, 2008.

Income Tax Expense. Income tax expense increased to \$6.7 million for the three months ended March 31, 2008 from \$5.3 million for the same period in 2007, reflecting effective income tax rates of 41.2% and 39.3% for these periods, respectively. The increase in the effective income tax rate was primarily attributable to an increase in the projected uncertain tax position liability.

Liquidity and Capital Resources

Historically, our primary liquidity requirements have been for acquisitions, working capital requirements and debt service under our credit facility. We have funded these requirements through internally generated cash flow and funds borrowed under our credit facility. At March 31, 2008, \$146.5 million was outstanding under our credit facility. We believe that cash generated from operations and available borrowings under our revolving credit facility will be sufficient to fund our operations for the next 12 months. We expect to be able to finance future acquisitions either with cash provided from operations, borrowings under our revolving credit facility, bank loans, debt or equity offerings, or some combination of the foregoing. The following discussion provides further details of our liquidity and capital resources.

Operating Activities:

Historically, our principal working capital need has been for accounts receivable. At March 31, 2008, our days sales outstanding ("DSO") was 62 days. Excluding the impact of the acquisition of Platinum Select Staffing, DSO was 60 days at March 31, 2008. At March 31, 2007, our DSO was 60 days and at December 31, 2007, our DSO was 59 days. The increase in DSO compared to March 31, 2007 and December 31, 2007 was due in part to the acquisition of Platinum Select Staffing in February 2008, which only contributed revenue for half of the first quarter, but increased accounts receivable at March 31, 2008. Our principal sources of cash to fund our working capital needs are cash generated from operating activities and borrowings under our revolving credit facility. Net cash provided by operations decreased \$0.3 million from \$17.2 million for the three months ended March 31, 2007 to \$16.9 million for the three months ended March 31, 2008. This decrease in net cash provided by operations was primarily driven by increases in accounts receivable and prepaid expenses during the three months ended March 31, 2008. These increases were partially offset by an increase in accrued compensation and benefits, accounts payable and other accrued expenses during the three months ended March 31, 2008.

Investing Activities:

We continue to have relatively low capital investment requirements. Capital expenditures were \$2.6 million and \$2.1 million for the three months ended March 31, 2008 and 2007, respectively. For the first three months of 2008, our primary capital expenditures were \$1.9 million for purchased and internally developed software and \$0.7 million for computers, leasehold improvements, furniture and equipment and other expenditures. We expect our future capital expenditure requirements to be similar to those during the three months ended March 31, 2008.

Our business acquisition expenditures were \$30.8 million for the three months ended March 31, 2008. The \$30.8 million of expenditures for three months ended March 31, 2008 was the cash consideration for the acquisition of Platinum Select Staffing in February 2008. We financed the acquisition using \$18.0 million of cash on hand at the purchase date and the balance from our revolving credit facility. We will continue to evaluate and look for acquisition opportunities that may require additional funding.

Financing Activities:

Our credit facility, the Second Amended and Restated Credit Agreement ("Credit Agreement"), provides for, among other things, a \$75.0 million secured revolving credit facility, a \$30.0 million letter of credit sub-facility, a \$15.0 million swing-line sub-facility, all maturing in November 2010, and a \$235.0 million secured term loan facility maturing in November 2011. The secured term loan facility was used primarily to fund our acquisition of The MHA Group, Inc. ("MHA") in November 2005, pay off then-existing borrowings, and repurchase shares of our common stock in May 2006.

The revolving credit facility portion of our Credit Agreement carries an unused fee of between 0.375% and 0.5% per annum based on our then current leverage ratio, and there are no mandatory reductions in the revolving commitment under the revolving credit facility. Borrowings under this revolving credit facility bear interest at floating rates based upon either a LIBOR or a prime interest rate option selected by us, plus a spread of 1.50% to 2.25% and 0.50% to 1.25%, respectively, to be determined based on our then current leverage ratio. Amounts

available under our revolving credit facility may be used for working capital, capital expenditures, permitted acquisitions and general corporate purposes, subject to various limitations. At March 31, 2008, there was \$13.5 million outstanding under our revolving credit facility.

The term loan portion of our Credit Agreement is subject to quarterly amortization of principal (in equal installments), with an amount equal to 1.25% of the initial aggregate principal amount of the facility payable quarterly through September 30, 2007 (except in the case of the initial quarterly payment on June 30, 2006 of 2.5%) and 2.5% of the initial aggregate principal amount of the facility payable quarterly from December 31, 2007 through September 30, 2010 with any remaining amounts payable in 2011. The term loan portion of our credit facility bears interest, at our option, at either LIBOR plus 1.75% or the Base Rate plus 0.75%. Voluntary prepayments of the term loan portion of the credit facility are applied as we may elect, including ratably to the remaining quarterly amortization payments.

We are required to make additional mandatory prepayments on the term loan with the proceeds of asset dispositions, extraordinary receipts, debt issuances and certain equity issuances. We also are required to make mandatory prepayments on the term loan within ninety days after the end of each fiscal year, which commenced in the fiscal year ended December 31, 2006, in an amount equal to 50% of our excess cash flow (as defined in the Credit Agreement), less any voluntary prepayments made during the fiscal year. These mandatory prepayment amounts, if any, are applied ratably to the remaining quarterly amortization payments. On February 5, 2008, we made a total payment of \$14.0 million, among which \$10.1 million was used for the 2007 excess cash flow requirement, \$3.8 million was applied to the quarterly principal amortization payments for three months ended March 31, 2008, and the remaining \$0.1 million to be applied to a future quarterly principal amortization payment. The voluntary prepayments made during 2007 and the \$10.1 million payment made in February 2008 satisfied the excess cash flow prepayment requirement for 2007. In connection with the voluntary payment, we wrote off \$0.1 million of deferred financing costs, which was recorded in interest expense. At March 31, 2008, total term loan outstanding (including both the current and long term portions) was \$133.0 million.

We are required to maintain a maximum leverage ratio, based on EBITDA and funded indebtedness as defined in the Credit Agreement, as of the end of each fiscal quarter of not more than 3.00 to 1.00 for the fiscal quarter ending March 31, 2008, decreasing throughout the term of the agreement to ultimately arrive at a ratio of 2.00 to 1.00 for the fiscal quarter ending March 31, 2009 and thereafter. We are also required to maintain a minimum fixed charge coverage ratio, based on EBITDA and debt and interest payments as defined in the Credit Agreement, as of the end of each fiscal quarter of not less than 1.50 to 1.00 for the fiscal quarter ending March 31, 2008, decreasing throughout the term of the agreement to ultimately arrive at a ratio of 1.25 to 1.00 for the fiscal quarter ending September 30, 2008 and thereafter. We are also subject to limitations on the amount of our annual capital expenditures and on the amount of consolidated total assets and consolidated EBITDA that may be owned or attributable to our foreign subsidiaries. We were in compliance with these requirements at March 31, 2008.

Under our Credit Agreement, our subsidiaries are not permitted to pay dividends or distributions to us, except for certain permitted dividends and distributions, including those related to taxes, certain reporting obligations under federal and state law and certain other ordinary course operating expenses, subject to the limitations contained in our Credit Agreement.

We are also required to maintain interest rate protection on at least 50% of the term loan portion of our Credit Agreement beginning November 2006 until May 2009. As of March 31, 2008, we maintained three interest rate swap agreements with notional amounts totaling \$75 million. We pay fixed rates ranging from 4.87% to 4.95% under these agreements and receive a floating three-month LIBOR. The agreements expire beginning June 2008 through December 2009, and no initial investments were made to enter into these agreements.

At March 31, 2008 and December 31, 2007, the interest rate swap agreements had a fair value of \$(2.2) million and \$(1.1) million, respectively, which is included in other liabilities in the accompanying consolidated

balance sheets. The Company's interest rate swaps are valued using commonly quoted intervals from observable market. In addition, the Company discounts its derivative liabilities to reflect the potential credit risk to lenders. We have formally documented the hedging relationships and account for these arrangements as cash flow hedges.

As of March 31, 2008 and December 31, 2007, our credit facility also served to collateralize certain letters of credit aggregating \$20.0 million for both periods, issued by us in the normal course of business.

Potential Fluctuations in Quarterly Results and Seasonality

Due to the regional and seasonal fluctuations in the hospital patient census and healthcare staffing needs of our hospital and healthcare facility clients and due to seasonal preferences for destinations of our temporary healthcare professionals, revenue, earnings and the number of temporary healthcare professionals on assignment are subject to moderate seasonal fluctuations. Many of our hospital and healthcare facility clients are located in areas that experience seasonal fluctuations in population during the winter and summer months. These facilities adjust their staffing levels to accommodate the change in this seasonal demand and many of these facilities utilize temporary healthcare professionals to satisfy these seasonal staffing needs. This historical seasonality of revenue and earnings may vary due to a variety of factors and the results of any one quarter are not necessarily indicative of the results to be expected for any other quarter or for any year.

Recent Accounting Pronouncements

In September 2006, the FASB issued Statement of Financial Accounting Standards No. 157 (SFAS No. 157), *Fair Value Measurements*. SFAS No. 157 defines fair value, establishes a framework for measuring fair value in generally accepted accounting principles, and expands disclosures related to fair value measurements. SFAS No. 157 does not require any new fair value measurements, but provides guidance on how to measure fair value by providing a fair value hierarchy used to classify the source of the information. SFAS No. 157 was effective for fiscal years beginning after November 15, 2007. In February 2008, the FASB issued FASB Staff Position No. 157-2, "Effective Date of FASB Statement No. 157", which permits the deferral of the effective date of SFAS No. 157 to fiscal years beginning after November 15, 2008 for all nonfinancial assets and liabilities, except those that are recognized or disclosed at fair value in the financial statements on a recurring basis. We adopted SFAS No. 157 beginning January 1, 2008 except as it applies to those nonfinancial assets and nonfinancial liabilities, and the adoption did not have a material effect on our consolidated financial statements.

In December 2007, the FASB issued Statement of Financial Accounting Standards No. 141R (SFAS No. 141R), *Business Combinations*. This statement establishes principles and requirements for the reporting entity in a business combination, including recognition and measurement in the financial statements of the identifiable assets acquired, the liabilities assumed, and any noncontrolling interest in the acquiree. This statement also establishes disclosure requirements to enable financial statement users to evaluate the nature and financial effects of the business combination. SFAS No. 141R applies prospectively to business combinations for which the acquisition date is on or after the beginning of the first annual reporting period beginning on or after December 15, 2008, and interim periods within those fiscal years. We are required to adopt SFAS No. 141R beginning January 1, 2009.

In March 2008, the FASB issued Statement of Financial Accounting Standards No. 161 (SFAS No. 161), *Disclosures about Derivative Instruments and Hedging Activities—an amendment of FASB Statement No. 133*. This statement requires entities that utilize derivative instruments to provide qualitative disclosures about their objectives and strategies for using such instruments, as well as any details of credit-risk-related contingent features contained within derivatives. SFAS No. 161 also requires entities to disclose additional information about the amounts and location of derivatives located within the financial statements, how the provisions of SFAS 133 has been applied, and the impact that hedges have on an entity's financial position, financial performance, and cash flows. Statement 161 is effective for fiscal years and interim periods beginning after November 15, 2008, with early application encouraged. We are required to adopt SFAS No. 161 beginning January 1, 2009, and we do not expect the adoption will have a material effect on our consolidated financial statements.

Special Note Regarding Forward-Looking Statements

This Quarterly Report on Form 10-Q contains "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. We based these forward-looking statements on our current expectations and projections about future events. Our actual results could differ materially from those discussed in, or implied by, these forward-looking statements. Forward-looking statements are identified by words such as "believe," "anticipate," "expect," "intend," "plan," "will," "may" and other similar expressions. In addition, any statements that refer to expectations, projections or other characterizations of future events or circumstances are forward-looking statements. The following factors could cause our actual results to differ materially from those implied by the forward-looking statements in this Quarterly Report:

- our ability to continue to recruit qualified temporary and permanent healthcare professionals at reasonable costs;
- · our ability to retain qualified temporary healthcare professionals for multiple assignments at reasonable costs;
- our ability to attract and retain sales and operational personnel;
- our ability to enter into contracts with hospitals, healthcare facility clients, affiliated healthcare networks and physician practice groups on terms attractive to us and to secure orders related to those contracts;
- our ability to demonstrate the value of our services to our healthcare and facility clients;
- our ability to maintain and enhance the brand identities we have developed, at reasonable costs;
- changes in the timing of hospital, healthcare facility and physician practice group clients' orders for temporary healthcare professionals;
- the general level of patient occupancy and utilization of services at our hospital and healthcare facility clients' facilities, including the potential impact on such utilization caused by adoption of alternative modes of healthcare delivery, which utilization may influence demand for our services;
- · the overall level of demand for services offered by temporary and permanent healthcare staffing providers;
- · the ability of our hospital, healthcare facility and physician practice group clients to retain and increase the productivity of their permanent staff;
- the variation in pricing of the healthcare facility contracts under which we place temporary healthcare professionals;
- our ability to successfully design our strategic growth, acquisition and integration strategies and to implement those strategies, including integration of acquired companies' accounting, management information, human resource and other administrative systems, and implementation or remediation of controls, procedures and policies at acquired companies;
- our ability to leverage our cost structure;
- access to and undisrupted performance of our management information and communication systems, including use of the Internet, and our candidate and client databases and payroll and billing software systems;
- our ability to keep our web sites operational at a reasonable cost and without service interruptions;
- the effect of existing or future government legislation and regulation;
- our ability to grow and operate our business in compliance with legislation and regulations, including regulations that may affect our clients and, in turn, affect demand for our services;

- the challenge to the classification of certain of our healthcare professionals as independent contractors;
- the impact of medical malpractice and other claims asserted against us;
- the disruption or adverse impact to our business as a result of a terrorist attack or breach of security of our data systems;
- our ability to carry out our business strategy and maintain sufficient cash flow and capital structure to support our business;
- · the loss of key officers and management personnel that could adversely affect our ability to remain competitive;
- · the effect of recognition by us of an impairment to goodwill; and
- the effect of adjustments by us to accruals for self-insured retentions.

Other factors that could cause actual results to differ from those implied by the forward-looking statements in this Quarterly Report on Form 10-Q are set forth in our Annual Report on Form 10-K for the year ended December 31, 2007.

Item 3. Quantitative and Qualitative Disclosures about Market Risk

Market risk is the risk of loss arising from adverse changes in market rates and prices, such as interest rates, foreign currency exchange rates and commodity prices. We do not believe that we have any material market risk exposure with respect to derivative or other financial instruments.

During 2008 and 2007, our primary exposure to market risk was interest rate risk associated with our debt instruments. See "Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations" for further description of our debt instruments. Excluding the effect of our interest rate swap arrangements, a 1% change in interest rates on our variable rate debt would have resulted in interest expense fluctuating approximately \$0.4 million and \$0.5 million during the three months ended March 31, 2008 and 2007, respectively. Considering the effect of our interest rate swap arrangements, a 1% change in interest rates on our variable rate debt would have resulted in interest expense fluctuating approximately \$0.2 million for both the three months ended March 31, 2008 and 2007.

Our international operations create exposure to foreign currency exchange rate risks. We believe that our foreign currency risk is immaterial.

Item 4. Controls and Procedures

We carried out an evaluation, under the supervision and with the participation of our management, including our Chief Executive Officer and Chief Financial Officer, of the effectiveness of our disclosure controls and procedures, as defined in Rules 13a-15(e) and 15d-15(e) of the Securities Exchange Act of 1934. Based on that evaluation, our Chief Executive Officer and Chief Financial Officer have concluded that our disclosure controls and procedures as of March 31, 2008 were effective to ensure that information required to be disclosed by us in reports that we file or submit under the Securities Exchange Act of 1934 is recorded, processed, summarized and reported within the time periods specified in Securities and Exchange Commission rules and forms.

There were no changes in our internal control over financial reporting that occurred during the quarter ended March 31, 2008 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II—OTHER INFORMATION

Item 6.	Exhibits
Exhibit No.	Description of Document
2.1	Asset Purchase Agreement, dated as of February 15, 2008, by and among AMN Healthcare Services, Inc., Platinum Select Healthcare Staffing, Inc., Platinum Select, L.P., Platinum Select Management, L.L.C., Patrick Aunkst, Kristi Bomar, Robert Quigley, Stephanie Houston and Lyle Seedig.*
4.1	Third Amendment to the Second Amended And Restated Credit Agreement, dated as of May 5, 2008, by and among AMN Healthcare, Inc., as borrower, AMN Healthcare Services, Inc., AMN Services, Inc., O'Grady-Peyton International (USA), Inc., International Healthcare Recruiters, Inc., AMN Staffing Services, Inc., The MHA Group Inc., Merritt, Hawkins & Associates, Med Travelers, Inc., RN Demand, Inc., Staff Care, Inc., MHA Allied Consulting, Inc., Med Travelers, LLC, Lifework, Inc., Pharmacy Choice, Inc., and Rx Pro Health, Inc., Platinum Select Healthcare Staffing, Inc. as guarantors, the lenders identified on the signature pages thereto and Bank of America, N.A., as administrative agent.*
10.1	AMN Healthcare Equity Plan (incorporated by reference to Exhibit 1 of AMN Healthcare Services, Inc.'s Definitive Proxy Statement on Schedule 14A filed on March 14, 2007)
31.1	Certification by Susan R. Nowakowski pursuant to Rule 13a-14(a) of the Securities Exchange Act of 1934*
31.2	Certification by David C. Dreyer pursuant to Rule 13a-14(a) of the Securities Exchange Act of 1934*
32.1	Certification by Susan R. Nowakowski pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002*
32.2	Certification by David C. Dreyer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002*

^{*} Filed herewith.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Date: May 9, 2008

AMN HEALTHCARE SERVICES, INC.

SUSAN R. NOWAKOWSKI Name: Susan R. Nowakowski Title: President and Chief Executive Officer

(Principal Executive Officer)

Date: May 9, 2008

/s/ DAVID C. DREYER

Name: Title:

David C. Dreyer Chief Accounting Officer, Chief Financial Officer and Treasurer (Principal Accounting and Financial Officer)

ASSET PURCHASE AGREEMENT

dated as of

February 15, 2008

by and among

PLATINUM SELECT HEALTHCARE STAFFING, INC.,

AMN HEALTHCARE, INC. (as to certain provisions),

PLATINUM SELECT, L.P.,

and

the PARTNERS OF PLATINUM SELECT, L.P.

EXECUTION COPY

TABLE OF CONTENTS

		Page
ARTICLE	E 1 PURCHASE AND SALE OF ASSETS	7
1.1	<u>Transfer of Assets</u>	7
1.2	Assets Not Transferred	3
1.3	Assumption of Certain Liabilities	S
1.4	Other Liabilities Not Assumed	g
ARTICLE	2 CLOSING DATE/PURCHASE PRICE	10
2.1	Closing Date	10
2.2	Purchase Price	10
2.3	Net Working Capital Adjustment	11
2.4	Earnout.	12
2.5	Certain Tax Matters.	14
ARTICLE	3 REPRESENTATIONS AND WARRANTIES OF SELLER AND EACH PARTNER	15
3.1	Organization, Power and Authority	15
3.2	Subsidiaries	16
3.3	Authorization and Approval of Agreements.	16
3.4	Effect of Agreement	16
3.5	<u>Approvals</u>	16
3.6	Capitalization.	17
3.7	<u>Intellectual Property.</u>	17
3.8	<u>Material Contracts</u>	22
3.9	Tax Matters.	23
3.10	Seller Financial Statements.	24
3.11		25
3.12	<u>Title to Properties</u>	25
3.13		25
3.14		27
3.15		27
3.16	<u> </u>	28
3.17	Employees and Other Compliance.	28
3.18		32
3.19		32
3.20	No Existing Discussions	32
3.21		32
3.22		33
3.23		33
3.24	Accuracy of Information	33
3 25	Fair Consideration: No Fraudulent Conveyance	3/

EXECUTION COPY

ARTICLE	4 REPRESENTATIONS AND WARRANTIES OF BUYER AND GUARANTOR	34
4.1	Organization, Corporate Power and Authority	34
4.2	Authorization of Agreement	34
4.3	Effect of Agreement	35
4.4	<u>Approvals</u>	35
4.5	Sufficient Funds	35
ARTICLE	5 ADDITIONAL COVENANTS	35
5.1	<u>Confidentiality</u>	35
5.2	Audit Cooperation	35
5.3	All Reasonable Efforts	36
5.4	Post-Closing Cooperation Relating to Assets and Excluded Assets.	36
5.5	No Post-Closing Retention of Copies	37
5.6	Post-Closing Tax Matters	37
5.7	Continued Existence of Seller	37
5.8	<u>Use of Purchase Price; Employee Matters</u>	37
5.9	No Continuing Business Operations; Noncompetition	39
5.10	Continuing Employees Confidentiality, Non-Competition and Non-Solicitation Agreements	39
5.11	Release of Guarantees	39
5.12	Nonassignable Contracts	39
5.13	<u>Professional Liability Insurance</u>	40
ARTICLE	6 THE CLOSING – SELLER AND PARTNER DELIVERABLES AND ACTIONS	40
6.1	Performance	40
6.2	Employees.	40
6.3	Bank of Texas Pay-Off	40
6.4	<u>Liabilities Schedule</u>	40
6.5	Seller Closing Debt	41
6.6	Partner Independent Consultant Agreements	41
6.7	Other Independent Consultant Agreements	41
6.8	Insurance; Insurance Tail	41
6.9	Transfer of A/R Bank Accounts	41
6.10	Seller Contracts Workbook	41
6.11	Additional Closing Deliverables and Actions	41
ARTICLE	7 THE CLOSING – BUYER DELIVERABLES AND ACTIONS	42
7.1	<u>Performance</u>	42
7.2	Additional Closing Deliverables and Actions	42
ARTICLE	8 SURVIVAL	43
8.1	Survival of Seller's and the Partners' Representations and Warranties	43
8.2	Survival of Buyer's and Guarantor's Representations and Warranties	43
8.3	Survival of Post-Closing Covenants	44
ARTICLE	9 INDEMNIFICATION; HOLDBACK	44
9.1	Obligations of Seller and the Partners	44
	-	

EALCOHON COL	EXEC	UTION	I COP
--------------	------	-------	-------

9.2	Obligations of Buyer	45
9.3	<u>Limitations</u>	45
9.4	Notice of Claim	47
9.5	<u>Defense of Third-Party Claims</u>	48
9.6	Contents of Notice of Claim	48
9.7	Resolution of Notice of Claim	49
9.8	Payment of Indemnification Claims	50
9.9	Release of Remaining Holdback Amount	50
9.10	Tax Consequences of Indemnification Payments	50
9.11	<u>Appointment of Seller as Attorney-In-Fact</u>	50
9.12	<u>Insurance Proceeds</u>	51
RTICLE	10 TERMINATION	52
10.1	Termination by Mutual Consent	52
10.2	Procedure and Effect of Termination	52
RTICLE	11 GENERAL	52
11.1	Amendments; Waivers	52
11.2	Schedules; Exhibits; Integration	52
11.3	Governing Law; Submission to Jurisdiction	52
11.4	No Assignment	53
11.5	<u>Headings</u>	53
11.6	<u>Counterparts</u>	53
11.7	Publicity and Reports	53
11.8	<u>Remedies</u>	53
11.9	<u>Parties in Interest</u>	53
11.10	<u>Notices</u>	53
11.11	Expenses and Attorneys Fees	55
11.12	<u>Specific Performance</u>	55
11.13	<u>Guaranty</u>	55
RTICLE	12 DEFINITIONS	55

List of Exhibits

Operating and Integration Plan

Exhibit A

Form of Partner Independent Consultant Agreement
Form of Independent Consultant Agreement between Seller and each of Barrett Alley, Wil Batchler, Carlos Mendez and Verve Communications
Legal Opinion of Baker & McKenzie LLP
Consent of Spouse
Assignment of Intellectual Property
Assignment and Assumption of Office Lease
Assignment and Assumption of Office Sublease
Assignment and Assumption of Temporary Storage Agreement
Bill of Sale and Assignment of Contract Rights
Confidentiality, Non-Competition and Non-Solicitation Agreement

List of Required Schedules

Transferable Permits and Approvals Schedule 1.1(d) Schedule 1.1(g) Deposits and Prepaid Expenses

Schedule 1.1(h) Contracts

Schedule 1.3 Assumed Liabilities Schedule 2.2(b) Liability Payments **Earnout Calculation** Schedule 2.4

Schedule 3.10(d) Debt

Healthcare Employee Payroll Amount and Undetermined Commissions Schedule 5.8(b)

Schedule 5.11 Partner Guarantees Schedule 5.12 Nonassignable Contracts

Schedule 6.10(m) Marital Status of the Limited Partners Schedule 12-1 Assumed Working Capital Obligations

Schedule 12-2 Logo

Schedule 12-3 Permitted Encumbrances

Disclosure Schedule

Jurisdictions Where Seller is Qualified to do Business
Effect of Agreement
Approvals
Capitalization of Seller

Section 3.7(a) Intellectual Property Section 3.7(b)

Information Regarding Seller Licenses

Section 3.7(e) Software Licenses Section 3.7(f) Source Code in Escrow

Section 3.7(i) Certain Communications and Rights

Section 3.7(k) Owned Intellectual Property Not Developed Internally Section 3.7(l) Proprietary Information and Assignment of Inventions

Section 3.7(n) Open Source Materials

Section 3.7(o) Servers and Internet Information

Section 3.8 Material Contracts Section 3.9(a) Tax Returns Liability For Taxes Section 3.9(b)

Section 3.9(c) Payment and Withholding of Taxes

Claims that Seller may be Subject to Taxation in Additional Jurisdictions Section 3.9(e)

Section 3.9(f)

Nonqualified Deferred Compensation Plans Section 3.9(i)

Financial Statements Section 3.10(a) Alleged Deficiencies Section 3.10(b) **Bank Account Information** Section 3.10(c) Section 3.11 Legal Proceedings

Section 3.12	Real and Personal Property
Section 3.13	Absence of Certain Changes
Section 3.14(a)	Defaults
Section 3.15(a)	Compliance With Applicable
Section 3.15(b)	Disclosures
2 4 2 4 2 4 2 4 3	To 1. 1.4 1

e Law

Section 3.15(c) Permits and Approvals Section 3.17(a) Seller Benefit Arrangements Section 3.17(d) Contributions and Payments

Participation in Seller Benefit Arrangements Section 3.17(f)

Section 3.17(j) Certain Contracts and Benefits

Section 3.17(k) Seller Employees

Section 3.17(l) Disputes and Labor Issues

Insurance Section 3.18 Privacy Policy Section 3.21 Section 3.22 Accounts Receivable

Section 3.23 Legal and Accounting Advisors

Seller Financial Statements Attachment 3.10(a)-1

Attachment 3.11-1 Workers Compensation Claims Log Attachment 3.12-1 Fixed Assets included in Seller Balance Sheet

Attachment 3.17(k)-1 Corporate Employees

Attachment 3.17(k)-2 Healthcare Employees

Attachment 3.17(k)-3 Platinum Select Work Order End Date Report Attachment 3.17(k)-4 Corporate Employee Bonus Arrangements Accounts Receivable as of December 31, 2007 Attachment 3.22-1 Attachment 3.22-2 Accounts Receivable as of February 11, 2008

ASSET PURCHASE AGREEMENT

This Asset Purchase Agreement (this "Agreement") is entered into as of February 15, 2008, among Platinum Select Healthcare Staffing, Inc., a Texas corporation ("<u>Buyer</u>"), Platinum Select, L.P., a Texas limited partnership ("<u>Seller</u>"), Platinum Select Management, L.L.C., a Texas limited liability company and general partner of Seller ("<u>General Partner</u>"), Patrick Aunkst ("<u>Aunkst</u>"), Kristi Bomar ("<u>Bomar</u>"), Robert Quigley ("<u>Quigley</u>"), Stephanie Houston ("<u>Houston</u>") and Lyle Seedig ("<u>Seedig</u>") (each of Aunkst, Bomar, Quigley, Houston and Seedig, a "<u>Limited Partner</u>" and, together, the "<u>Limited Partners</u>," and, together with General Partner, the "<u>Partners</u>"), and, with respect to Article 4, Section 7.2(f), Section 8.2 and Section 11.13 only, AMN Healthcare, Inc., a Nevada corporation ("<u>Guarantor</u>"). Capitalized terms used herein that are not otherwise defined shall have the meanings set forth in Article 12.

RECITALS

WHEREAS, Seller is engaged in the Business.

WHEREAS, the Partners are the only partners of Seller and collectively own 100% of the rights and interest in Seller, including rights to receive distributions of revenues, allocations of income and loss and distribution of liquidation proceeds and all management rights, voting rights and rights to consent.

WHEREAS, Buyer desires to purchase and acquire certain assets, properties and contractual rights of Seller used in connection with the Business, and Seller desires to sell, transfer, assign and convey such assets, properties and contractual rights to Buyer, and in conjunction therewith, Buyer desires to assume certain identified obligations and liabilities associated with such assets so purchased, all on the terms and subject to the conditions described herein.

WHEREAS, concurrently with the execution and delivery of this Agreement, and as a condition and material inducement to Buyer's willingness to enter into this Agreement, Landry Seedig is executing and delivering to Buyer an employment offer letter (including the non-competition and non-solicitation covenants set forth therein and the related Confidentiality and Assignment of Inventions Agreement attached as an exhibit thereto) (the "Employment Agreement"), which Employment Agreement shall become effective only at the Closing.

WHEREAS, as a further condition and inducement for Buyer to enter into this Agreement, concurrent with the execution and delivery of this Agreement, the Partners have, by resolutions duly adopted by the Partners, approved and adopted the terms of this Agreement and the transactions contemplated hereby.

AGREEMENT

In consideration of the mutual promises contained herein and intending to be legally bound, the parties agree as follows:

ARTICLE 1 PURCHASE AND SALE OF ASSETS

The transactions contemplated by this Agreement are as follows:

- 1.1 <u>Transfer of Assets</u>. Subject to the terms and conditions of this Agreement, Seller hereby sells, transfers, irrevocably assigns and delivers to Buyer, and Buyer hereby purchases from Seller, free and clear of any Encumbrances, all of Seller's right, title and interest in and to all of Seller's property and assets, real, personal or mixed, tangible and intangible, of every kind and description, wherever located, including, without limitation, the following (but excluding the Excluded Assets):
- (a) all intellectual property which is used in the Business or is necessary for the operation of the Business as presently conducted or as contemplated to be conducted, including any or all of the following and all rights in, arising out of, or associated therewith, including, without limitation, all such rights used in the operation of the Business: (i) any and all patents and/or patent applications which may be granted on or claim priority of (or serve as the basis for a priority claim by) and all reissues, divisions, renewals, extensions, provisionals, continuations and continuations-in-part thereof, together with all priority rights and counterpart applications under any existing or future international patent conventions, agreements or treaties; (ii) all inventions (whether patentable or not), invention disclosures, improvements, trade secrets, proprietary information, know-how, technology, technical information, data and Customer lists, engineering procedures and all documentation relating to any of the foregoing; (iii) all Trademarks, Domain Names, Trade Secrets and all other rights corresponding thereto throughout the world; (iv) all copyrights, copyrights registrations and applications therefor, and all other rights corresponding thereto throughout the world; (iv) all software, including, without limitation, source code (but excluding "shrink wrapped" software which is generally available to the public); (vi) all works of authorship, databases and data collections and all rights therein throughout the world; and (vii) any similar or equivalent rights to any of the foregoing anywhere in the world, together with the goodwill and the business appurtenant thereto and the right to sue third parties for infringement (including, without limitation, damages and injunctive relief) of any of the foregoing based on activities occurring prior to and including the Closing Date, and any current or future right to receive royalties based on any of the foregoing, whether choate or inchoate, known or unknown, contingen
 - (b) all existing documentation relating to the development, use, maintenance and operation of the Intellectual Property;
- (c) all Tangible Personal Property, including those items described in <u>Section 3.12</u> of the Disclosure Schedule, and all rights and interests in, or benefits under, any leases of Tangible Personal Property;
- (d) all permits, licenses, franchises, consents and other approvals and operating rights relating to the Assets that are assignable or transferable, including those permits, licenses, franchises, consents and other approvals and operating rights described in <u>Schedule 1.1(d)</u>;

- (e) all data and records in Seller's or any of the Partner's possession related to the Assets or the operation of the Business, including financial and accounting records, creative materials, advertising materials, promotional materials, studies, reports, correspondence and other similar documents;
- (f) all claims of Seller against third parties relating to the Assets or the operation of the Business, whether choate or inchoate, known or unknown, contingent or noncontingent;
- (g) all rights of Seller's relating to deposits and prepaid expenses, claims for refunds and rights to offset in respect thereof relating to the Assets or the operation of the Business, including those rights described in <u>Schedule 1.1(g)</u>;
- (h) all rights and interests in and to the Contracts of Seller, including, without limitation, the Contracts of Seller listed in Schedule 1.1(h) and the Seller Contracts Workbook;
 - (i) all other tangible and intangible assets of Seller used in the Business which are not identified as "Excluded Assets" pursuant to Section 1.2;
- (j) all data and information that is collected from, or on behalf of, visitors or users of Seller's website (or the services provided by Seller's website), including, without limitation, user names, passwords, account recovery detail, aggregate information, web analytics, all information collected, processed and/or stored through use of TKO software, and any other information necessary for operating the services provided by Seller's website following the Closing as such website was running as of the date hereof and immediately prior to the Closing;
- (k) all accounts receivable related to the Business and relating to services rendered (i) by Seller prior to the Closing or (ii) by Buyer after the Closing pursuant to any Contract not assigned to Buyer at the Closing by operation of Section 5.12 hereto;
 - (l) all A/R Bank Accounts; and
- (m) all routing and billing information and components used in connection with the Business, including, without limitation, all routing and billing computer software and programs containing any Customer information.
 - All of the property and assets to be transferred to Buyer hereunder are herein referred to collectively as the "Assets".
- 1.2 <u>Assets Not Transferred</u>. Notwithstanding anything to the contrary contained in Section 1.1 or elsewhere in this Agreement, the following assets of Seller (collectively, the "<u>Excluded Assets</u>") are not part of the sale and purchase contemplated in Section 1.1, are excluded from the Assets and shall remain the property of Seller after the Closing:
 - (a) all minute books, stock records, ownership records, corporate seals, partnership seals and tax returns;

- (b) all personnel records and other records that Seller is required by Law to retain in its possession; <u>provided</u>, <u>however</u>, that Buyer has been provided copies of all personnel records for all Continuing Employees;
- (c) all purchased tickets for the 2007-2008 National Basketball Association season to Dallas Mavericks basketball games (both regular and post-regular season games) and all rights to purchase tickets for future National Basketball Association seasons;
 - (d) all insurance policies of Seller other than those included in the Assumed Seller Benefits Arrangements;
- (e) the laptop computers used by the Limited Partners; <u>provided</u>, <u>however</u>, that there shall be no more than five (5) of such computers and only one (1) computer per Limited Partner, and <u>provided</u>, <u>further</u>, that all data, records and other information related to the Assets shall be entirely and permanently deleted from each Limited Partner's computer at or before the later of (i) the Closing and (ii) the termination of any consulting or employment arrangement between such Limited Partner and Buyer; and
 - (f) all rights of Seller and the Partners under the Transaction Documents.
- 1.3 <u>Assumption of Certain Liabilities</u>. Subject to the terms and conditions of this Agreement, Buyer hereby assumes, takes the Assets subject to and shall be liable for, only those liabilities and obligations identified on <u>Schedule 1.3</u> (collectively, the "<u>Assumed Liabilities</u>").
- 1.4 Other Liabilities Not Assumed. Except for the Assumed Liabilities, Buyer does not and shall not be deemed to assume any other liabilities or obligations of Seller in connection with the Transactions. All liabilities and obligations of Seller other than the Assumed Liabilities are referred to herein as the "Excluded Liabilities". The Excluded Liabilities will include, without limitation, the following (but, for avoidance of doubt, will not include any Assumed Liabilities):
- (a) any liabilities or obligations (whether assessed or unassessed) of Seller for any Taxes, any Transfer Taxes (whether imposed on Seller or Buyer) and any Taxes of the Business for any period (or portion thereof) ending on or prior to, the Closing Date;
- (b) any and all liabilities or obligations of Seller or the Partners payable to any Person identified on <u>Schedule 2.2(b)</u> hereto, whether or not such liabilities are set forth on <u>Schedule 2.2(b)</u> and whether or not such liabilities are accrued in connection with, or with respect to, any Asset or the Business;
- (c) except for the Healthcare Employee Payroll Amount and any Undetermined Commissions set forth on Schedule 5.8(b) hereto, any liabilities or obligations arising out of or related to (i) the employment or other service relationship with Seller, and (ii) the termination or compensation by Seller of any Seller Employee or other current or former service provider of Seller, including, without limitation, any Taxes (including, without limitation, any employer withholding and similar

obligations), compensation claims, workers' compensation benefits or 401(k) Matching Payments, whether or not such liabilities are accrued in connection with, or with respect to, the Business or the sale of the Assets pursuant hereto; <u>provided</u> that any liability or obligation under or relating to any Assumed Seller Benefit Arrangement (other than any liability or obligation related to Seller's 401(k) plan, including the 401(k) Matching Payments) with respect to an event or circumstance first occurring or arising after the Closing Date shall not be an Excluded Liability;

- (d) any liabilities or obligations of Seller or the Partners incurred, arising from or out of or in connection with this Agreement or the events or negotiations leading up to this Agreement;
- (e) all liabilities or obligations of Seller or the Partners in respect of any of the Excluded Assets, including, without limitation, liability for any breaches or nonfulfillments of representations, warranties and covenants (including support requirements and indemnification provisions);
- (f) any liabilities or obligations under or relating to any Seller Benefit Arrangement, whether or not such liability or obligation arises prior to or after the Closing Date; provided that any liability or obligation under or relating to any Assumed Seller Benefit Arrangement (other than any liability or obligation related to Seller's 401(k) plan, including the 401(k) Matching Payments) with respect to an event or circumstance first occurring or arising after the Closing Date shall not be an Excluded Liability; and
- (g) any other liability or obligation to which a Seller Employee becomes entitled under any compensatory, benefit or severance policy, plan, agreement, arrangement or program which exists or arises, or may be deemed to exist or arise, under any Law or otherwise, as a result of, or in connection with, the sale of the Business hereunder.

ARTICLE 2 CLOSING DATE/PURCHASE PRICE

- 2.1 <u>Closing Date</u>. The closing of the transactions contemplated under this Agreement (the "<u>Closing</u>") shall take place at the offices of Heller Ehrman LLP, 275 Middlefield Road, Menlo Park, California 94025 concurrently with the execution and delivery of this Agreement by all of the parties hereto (the "<u>Closing Date</u>"). All transactions shall be deemed to take place simultaneously and none of them shall be deemed to have occurred until all shall have occurred.
- 2.2 <u>Purchase Price</u>. As consideration for the sale to Buyer of the Assets, subject to the provisions of Section 2.4 hereof, Buyer shall pay to Seller cash in the amount of (i) thirty two million, eight hundred eighty four thousand, six hundred dollars (\$32,884,600), <u>less</u> (ii) the amounts, if any, set forth on <u>Schedule 2.2(b)</u> hereto, <u>plus</u> (iii) the Net Working Capital Increase, if any, <u>less</u> (iv) the Net Working Capital Decrease, if any ("<u>Purchase Price</u>"). The Purchase Price shall be paid as follows:
- (a) An amount in cash (the "Initial Cash Payment") equal to (i) thirty two million, eight hundred eighty four thousand, six hundred dollars (\$32,884,600), less (ii) the amounts, if any, set forth on Schedule 2.2(b) hereto, less (iii) the Holdback Amount to be paid to Seller at Closing in cash by wire transfer of immediately available funds;

- (b) The Transaction Expenses and Seller Closing Debt (including, without limitation, any Debt outstanding and payable to Bank of Texas and set forth in the Pay-Off Letter) set forth on <u>Schedule 2.2(b)</u> which shall be paid to those Persons identified on <u>Schedule 2.2(b)</u> hereto (which <u>Schedule 2.2(b)</u> hereto shall be delivered to Buyer immediately prior to the Closing) at Closing in cash by wire transfers of immediately available funds (the "<u>Liability Payments</u>");
- (c) Two million five hundred thousand dollars (\$2,500,000) (the "Holdback Amount") to be paid to Seller, if at all, subject to the provisions of Article 9 hereof; and
 - (d) The Net Working Capital Increase, if any, to be paid to Seller, if at all, subject to the provisions of Section 2.3 hereof.
- 2.3 Net Working Capital Adjustment. Within ten (10) Business Days after the Closing Date, Seller shall deliver to Buyer an unaudited balance sheet of Seller as of the close of business on the Closing Date (the "Closing Balance Sheet"), together with a certificate signed by Aunkst and Bomar (i) certifying that the Closing Balance Sheet fairly presents, on a good faith basis and using Seller's commercially reasonable efforts, the financial position of Seller as of the Closing Date, (ii) setting forth the amount of Seller's Net Working Capital as of the Closing Date, (iii) setting forth an accurate and complete list of Seller's accounts receivable as of the Closing Date which indicates the aging of such accounts receivable and (iv) including Seller's calculation of the Net Working Capital (the "Closing Balance Sheet Certificate"). Following the delivery by Seller of the Closing Balance Sheet Certificate, Buyer and its representatives shall be given all such access as they may reasonably require to those books and records of Seller in the possession of, and/or under the control of, Seller and/or the Partners, and access to such current or former Seller Employees or representatives of Seller as they may reasonably require for the purposes of resolving any disputes or responding to any matters or inquiries raised concerning the Closing Balance Sheet and/or the contents of the Closing Balance Sheet Certificate. Buyer shall have ten (10) Business Days following receipt of the Closing Balance Sheet Certificate to notify Seller in writing either that (i) the Net Working Capital as proposed by Seller is acceptable ("Confirmation Certificate") or (ii) the Net Working Capital as proposed by Seller is not acceptable (a "Balance Sheet Dispute Notice"). If a Confirmation Certificate is delivered by Buyer, then the Net Working Capital set forth in the Closing Balance Sheet Certificate shall be the Net Working Capital for purposes of this Agreement. If a Balance Sheet Dispute Notice is delivered by Buyer, the Net Working Capital shall be determined by negotiation between Buyer and Seller. If Buyer and Seller shall be unable to reach agreement with respect to such calculation within thirty (30) days after delivery of a Balance Sheet Dispute Notice to Seller by Buyer, the determination of such amount shall be submitted to the San Diego office of Ernst & Young LLP, independent certified public accountants (the "Independent Accountant"), for review and such review by the Independent Accountant shall be limited to (a) such items and calculations as were addressed in the written notice of objection of Buyer that have not been resolved by the parties and (b) any factual or mathematical errors contained in the information provided by Seller. In the event that Buyer and Seller shall submit any dispute to the Independent Accountant, each such party may submit a "position paper" to the Independent Accountant setting forth the position of such party with respect to such dispute, to be considered

by such Independent Accountant as it deems fit. The parties shall cause the Independent Accountant to review, subject to limitations of the previous sentence, as promptly as practicable, the calculation of Net Working Capital and to make, subject to the limitations of the previous sentence, such corrections thereto as it deems appropriate consistent with the terms of this Agreement. The Independent Accountant shall issue a written report of its review, setting forth in reasonable detail its calculation of Net Working Capital, which calculation shall be conclusive and binding on the parties to this Agreement and shall be the sole and exclusive means of resolving such disputes. If requested by the Independent Accountant, Buyer and Seller (on its own behalf and on behalf of the General Partner and the Limited Partners) shall execute an engagement letter with the Independent Accountant setting forth the terms and conditions of such engagement, which shall be consistent with the terms of this Section 2.3. If the Independent Accountant is engaged pursuant to this Section 2.3, Buyer and Seller shall use commercially reasonable efforts to cause the Independent Accountant to complete its review and written report as soon as reasonably practicable following such engagement. All fees and expenses relating to the engagement of the Independent Accountant shall be borne equally by Buyer and Seller, and the portion of such fees and expenses payable by Seller shall be "Seller Expenses" which Buyer may, in its sole discretion, elect to pay to the Independent Accountant, in which case the amount of Seller Expenses shall be "Losses" for purposes of Article 9 without regard to the Basket and shall, among other remedies available to Buyer, be subject to Buyer's Rights of Set-Off. Following the determination of the Net Working Capital in accordance with this Section 2.3, (i) the Net Working Capital Increase, if any, shall be paid to Seller on the fifth (5th) Business Day following the date that the determination of the Net Working Capital shall become conclusive and binding on the parties to this Agreement in accordance with this Section 2.3 or (ii) the Net Working Capital Decrease, if any, shall, at the sole election of Buyer, be "Losses" for purposes of Article 9 without regard to the Basket and be deducted from the Holdback Amount and/or be subject to Buyer's Rights of Set-Off.

2.4 Earnout.

(a) <u>Earnout Payment</u>. In addition to the Purchase Price payable to Seller pursuant to Section 2.2 hereof, Seller shall, subject to and contingent upon achievement of the EBITDA and revenue threshold (for the Business as conducted by Buyer following the Closing) set forth on <u>Schedule 2.4</u> hereto and measured for the 12 months ending at the close of business on December 31, 2008 (the "<u>Earnout Period</u>"), be entitled to receive an amount of cash equal to the Applicable Earnout Amount set forth on <u>Schedule 2.4</u>. The provisions of this Section 2.4(a) are subject to the provisions of Section 2.4(c) (regarding Buyer's Rights of Set-Off) below.

(b) Time for Determination.

(i) As soon as reasonably practicable following December 31, 2008 and no earlier than the completion of Buyer's annual audit, Buyer shall determine (A) the Earnout Period EBITDA, (B) the Earnout Period Total Revenue and (C) the resulting Applicable Earnout Amount. A copy of Buyer's computation of the Earnout Period EBITDA, the Earnout Period Total Revenue, and the resulting Applicable Earnout Amount, in each case as set forth in, and (in the case of the Applicable Earnout Amount) calculated in accordance with, Schedule 2.4 hereto, shall be delivered to Seller on or prior to March 3, 2009 and, unless Seller shall object to such computation within ten (10) Business Days after receipt thereof in accordance with the provisions of Section 2.4(b)(ii) below, the computations shall be binding and conclusive on the parties to this Agreement for purposes of this Agreement.

(ii) If Seller shall object to Buyer's computations by notifying Buyer in writing within ten (10) Business Days after receipt of Buyer's computations pursuant to Section 2.4(b)(i), the amount of Earnout Period EBITDA or Earnout Period Total Revenue and/or the resulting Applicable Earnout Amount shall be determined by negotiation between Buyer and Seller. If Buyer and Seller shall be unable to reach agreement with respect to such computations within thirty (30) days after such notification to Buyer by Seller, the determination of such amounts shall be submitted to the Independent Accountant for review, and such review by the Independent Accountant shall be limited to (A) such items and calculations as were addressed in the written notice of objection of Seller that have not been resolved by the parties and (B) any factual or mathematical errors contained in the information provided to or by Buyer. The parties shall cause the Independent Accountant to review, subject to limitations of the previous sentence, as promptly as practicable, the calculation of the Earnout Period EBITDA and the Earnout Period Total Revenue and to make, subject to the limitations of the previous sentence, such corrections thereto as it deems appropriate consistent with the terms of this Agreement. The Independent Accountant shall issue a written report of its review, setting forth in reasonable detail its calculation of the Earnout Period EBITDA and the Earnout Period Total Revenue, which calculation with respect to matters solely involving the application of accounting principles or Tax rules shall be conclusive and binding on the parties to this Agreement and shall be the sole and exclusive means of resolving such disputes. If requested by the Independent Accountant, Buyer and Seller (on its own behalf and on behalf of the General Partner and the Limited Partners) shall execute an engagement letter with the Independent Accountant setting forth the terms and conditions of such engagement, which shall be consistent with the terms of this Section 2.4. If the Independent Accountant is engaged pursuant to this Section 2.4(b)(ii), Buyer and Seller shall use commercially reasonable efforts to cause the Independent Accountant to complete its review and written report as soon as reasonably practicable following such engagement. All fees and expenses relating to the engagement of the Independent Accountant shall be borne equally by Buyer and Seller. Buyer may, in its sole discretion, elect to (but under no circumstances shall be obligated to) pay to the Independent Accountant the fees and expenses for which Seller is responsible pursuant to this Section 2.4(b)(ii) (the "Seller" Expenses"), in which case Buyer shall be reimbursed for such payment by deduction of the amount of the Seller Expenses from the Rights of Set-Off or the amounts due pursuant to this Section 2.4, if any, as "Losses" for purposes of Article 9; provided, however, that if and only if the Independent Accountant enters into an engagement letter with Buyer and Seller which provides that Buyer and Seller are severally and not jointly liable for equal portions of all of the Independent Accountant's fees and expenses in excess of \$30,000, Buyer will not deduct from the amounts due pursuant to Section 2.4 pursuant to Buyer's Rights of Set-Off any amount of Seller Expenses which exceed \$15,000 without Seller's prior written consent, which shall not be unreasonably withheld.

(iii) The Applicable Earnout Amount, if any, shall be paid to Seller on the later of (A) March 17, 2009 and (B) the fifth (5th) Business Day following the date that the determination of the Applicable Earnout Amount shall become binding and conclusive on the parties to this Agreement in accordance with Section 2.4(b)(i) or 2.4(b)(ii) above, as the case may be.

- (c) <u>Right of Set-Off</u>. Buyer shall have the right to withhold and set off against any amount otherwise due to be paid pursuant to this Section 2.4 the amount of (i) any Net Working Capital Decrease and (ii) any Losses to which any Buyer Indemnified Party may be entitled under Article 9 of this Agreement or any other agreement entered into pursuant to this Agreement (the "<u>Rights of Set-Off</u>").
- (d) <u>Conduct of the Business</u>. For the period commencing on the Business Day immediately following the Closing Date and ending on the close of business on December 31, 2008, Buyer shall use commercially reasonable efforts to (i) conduct the Business in accordance with the operating and integration plan (the "<u>Operating and Integration Plan</u>"), which Operating and Integration Plan has been agreed upon by Buyer and Seller and is attached hereto as <u>Exhibit A</u>, and (ii) for each quarterly period of fiscal 2008, provide to the Partners financial information for the Business consisting of an income statement, a cash flow statement, a statement of budget versus actual results and a trial balance. Notwithstanding the foregoing, following the Closing Date, Buyer and Seller may, but shall not be obligated to, amend the Operating and Integration Plan, which amendment shall be binding on the parties to this Agreement and the Partners.
- (e) Acknowledgement of Seller and the Partners. Seller and each Partner acknowledges that (i) upon the Closing and except as expressly provided to the contrary in Section 2.4(d) of this Agreement, Buyer has the right to operate the Business and Buyer's other businesses in any way that Buyer deems appropriate in Buyer's sole discretion, (ii) Buyer has no obligation to operate the Business in order to achieve any Applicable Earnout Amount or to maximize the amount of any Applicable Earnout Amount, (iii) there is no assurance that Seller will receive any Earnout Amount and Buyer has not promised or projected any Earnout Amount, (iv) Buyer has no fiduciary duty or express or implied duty to Seller or any Partner, including any implied duty of good faith and fair dealing, (v) the parties solely intend the express provisions of this Agreement to govern their contractual relationship, (vi) the right of Seller to payment of the Earnout Amount, if any, shall not bear any interest and (vii) the right of Seller to a portion of the Earnout Amount, if any, shall not be represented by a certificate or other instrument, shall not represent an ownership interest in Buyer or the Business and shall not entitle Seller to any rights common to any holder of any equity security of Buyer. Each of Seller and each Partner hereby waives any fiduciary duty or express or implied duty of Buyer to Seller or such Partner, including any implied duty of good faith and fair dealing, with respect to the matters contemplated by this Section 2.4.

2.5 Certain Tax Matters.

(a) Seller shall be responsible for paying, shall promptly discharge when due, and shall reimburse, indemnify and hold harmless Buyer and any Buyer subsidiary from, any sales or use, transfer, real property gains, excise, stamp or other similar Taxes resulting from the sale of Assets and the assumption of the Assumed Liabilities contemplated by this Agreement (the "<u>Transfer Taxes</u>"). Buyer and Seller agree to cooperate to the extent commercially reasonable and legally permitted to minimize any Transfer Taxes.

- (b) For purposes of complying with the requirements of Section 1060 of the Code, the Purchase Price and any other amounts that may be treated as part of the total consideration for the Assets, as determined for Tax purposes, shall be allocated among the Assets in accordance with fair market value of the Assets as provided in the purchase price allocation schedule to be prepared by Buyer in its reasonable discretion after consultation with Seller (the "Purchase Price Allocation Schedule"). Each of Buyer and Seller agrees to prepare its federal, state and foreign income Tax Returns for all current and future tax reporting periods and file Form 8594 (and corresponding state forms) with respect to transfer of the Assets to Buyer in a manner consistent with such allocation. If any state, federal or foreign taxing authority challenges such allocation, the party receiving notice of such challenge shall give the other prompt written notice of such challenge, and the parties shall cooperate in good faith in responding to it in order to preserve the effectiveness of the allocation.
- (c) Buyer shall be entitled to deduct and withhold from any payment made under this Agreement such amounts as may be required to be deducted and withheld under the Tax Laws of the United States, any state, or any political subdivision of either. To the extent amounts are so withheld and paid to any appropriate taxing authority, Buyer shall be treated as though it had paid that amount directly to the party entitled to receive such payment.

ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF SELLER AND EACH PARTNER

Except as otherwise indicated in the Disclosure Schedule attached hereto (it being understood that Disclosure Schedule statements and information referenced in the Disclosure Schedule shall be organized under sections and subsections that correspond to those contained in this Agreement and that a statement or reference under a particular section or subsection of the Disclosure Schedule shall be deemed an exception only to (or, as applicable, a disclosure for purposes of only) (a) the representations and warranties of Seller and/or the Partners, as applicable, that are contained in the corresponding section or subsection of this Agreement and (b) any other representations and warranties of Seller and/or the Partners that are contained in this Agreement, but only, in each case, if the relevance of that statement or reference as an exception to (or disclosure for purposes of) such other representations and warranties is reasonably apparent on its face), Seller hereby represents and warrants to Buyer, and each Partner (severally and not jointly) hereby represents and warrants to Buyer, as follows:

3.1 <u>Organization, Power and Authority.</u> Seller is a limited partnership duly organized, validly existing and in good standing under the laws of the State of Texas. <u>Section 3.1</u> of the Disclosure Schedule contains a complete and accurate list of any other jurisdiction in which Seller is qualified to do business. Seller is duly qualified to do business and is in good standing under the laws of each jurisdiction in which either ownership or use of the properties owned or used by it, or the nature of the activities conducted by it, requires such qualification. Seller has all requisite partnership power and authority to own, operate and lease the Assets and to conduct the Business, to execute and deliver the Transaction Documents to which it is a party and to perform its obligations thereunder. Seller is not in violation of its certificate of limited partnership or the Limited Partnership Agreement. Seller has provided Buyer with true, correct and complete copies of its certificate of limited partnership and the Limited Partnership Agreement is in full force and effect, has not been amended since the date thereof and is the only agreement in effect governing the respective rights and obligations of the Partners with respect to Seller.

3.2 <u>Subsidiaries</u>. Seller does not have any subsidiaries or any equity or ownership interest (or any interest convertible or exchangeable for any equity or ownership interest), whether direct or indirect, in any Person. Seller is not obligated to make nor is it bound by any agreement or obligation to make any investment in or capital contribution in or on behalf of any other Person.

3.3 Authorization and Approval of Agreements.

- (a) The execution, delivery and performance by Seller of the Transaction Documents, and the consummation by it of the Transactions, has been duly authorized by Seller and the Partners. This Agreement has been, and each other Transaction Document will be at the Closing, duly executed and delivered by Seller and constitute, or will constitute, when delivered, the legal, valid and binding obligation of Seller, enforceable against Seller in accordance with their respective terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium, and other similar laws relating to or limiting creditors' rights generally and to the principles of equity (regardless of whether considered in a proceeding in equity or an action at law).
- (b) This Agreement has been, and each other Transaction Document to which any Partner is a party will be at the Closing, duly executed and delivered by each such Partner and constitute, or will, when delivered, constitute the legal, valid and binding obligation of such Partner, enforceable against such Partner in accordance with their respective terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium, and other similar laws relating to or limiting creditors' rights generally and to the principles of equity (regardless of whether considered in a proceeding in equity or an action at law).
- 3.4 Effect of Agreement. The execution and delivery of the Transaction Documents to which Seller or any Partner is a party and each instrument required hereby to be executed and delivered by Seller or any Partner and the performance and consummation by Seller or any Partner of the Transactions will not (a) conflict with or violate the Limited Partnership Agreement or any Law to which Seller or any Partner is subject, or any judgment, award or decree applicable to Seller or any of the Assets, (b) conflict with, result in a breach of, constitute (with or without due notice or lapse of time or both) a default under, result in the acceleration of, create in any party the right to accelerate, terminate, modify or cancel, or require any notice, consent or waiver under, or result in any loss of any benefit to which Seller is entitled under, any Material Contract or Permit, or (c) result in the creation or imposition of any Encumbrance upon the Business or any Asset.
- 3.5 <u>Approvals</u>. No Approval or Order or action of or filing with any Governmental Entity or other Person is required to be obtained by Seller or any Partner for the execution and delivery by Seller or any Partner of the Transaction Documents to which Seller or any Partner is a party or the consummation of the Transactions, including, without limitation, the sale, transfer, assignment and conveyance of all of Seller's right, title and interests in and to the Assets to Buyer.

3.6 Capitalization.

- (a) <u>Section 3.6(a)</u> of the Disclosure Schedule sets forth the name, address, and percentage ownership interest of each Partner of Seller. All outstanding limited partnership ownership interests in Seller have been duly authorized and validly issued, are fully paid and nonassessable, were not issued in violation of and are not subject to any right of rescission, right of first refusal or preemptive right, have been offered, issued and sold and delivered by Seller in compliance with all requirements of applicable Law and all requirements set forth in applicable contracts and Seller has received all consideration due to it in connection with the sale and issuance of such limited partnership ownership interests.
- (b) Except as set forth in Section 3.6(a) of the Disclosure Schedule, there are no ownership appreciation rights, options, warrants, calls, rights, commitments, conversion privileges or preemptive or other rights or Contracts outstanding to purchase or otherwise acquire any ownership interests of Seller or any securities or debt convertible into or exchangeable for Seller ownership interests or obligating Seller to grant, extend or enter into any such option, warrant, call, right, commitment, conversion privilege or preemptive or other right or Contract. There are no voting agreements, registration rights, rights of first refusal, preemptive rights, co-sale rights or other restrictions applicable to any outstanding ownership interests or securities of Seller.

3.7 Intellectual Property.

- (a) <u>Section 3.7(a)</u> of the Disclosure Schedule lists all of the Intellectual Property owned, used or held by or licensed by or to Seller, including, without limitation:
- (i) all patents and pending patent applications owned by Seller, including for each such patent or patent application the patent or serial number, country, filing date and title;
- (ii) all registered trademarks and service marks of Seller, and all pending applications for registration by Seller of trademarks, including for each such trademark or service mark, the registration or application number, country, filing and expiration date;
 - (iii) to the Knowledge of Seller, all worldwide common law trademarks and service marks of Seller;
- (iv) all registered copyrights of Seller and all applications by Seller for registration of copyrights, including the registration number, country and filing and expiration date of each such copyright;
 - (v) all Domain Names owned or used by Seller;
 - (vi) to the Knowledge of Seller, all unregistered Intellectual Property owned or used by Seller;
 - (vii) all material Software used in connection with the Business;

(viii) all licenses or sublicenses by Seller to any person or entity of (A) any of the rights identified in subparagraphs (i) through (vii) above, and (B) any of Seller's other proprietary information;

- (ix) all licenses or sublicenses by any other person or entity to Seller of any other material Intellectual Property; and
- (x) any other instrument or other Contract governing any Intellectual Property Right.
- (b) Each license, sublicense or other agreement identified in Section 3.7(a)(viii) and Section 3.7(a)(ix) of the Disclosure Schedule (each a "Seller License") is a valid and binding obligation of Seller, enforceable in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium, and other similar laws relating to or limiting creditors' rights generally and to the principles of equity (regardless of whether considered in a proceeding in equity or an action at law). With respect to each Seller License, there is no default (or event that with the giving of notice or passage of time would constitute a default) by Seller or, to the Knowledge of Seller, the other party thereto. Seller has not received any written notice (and Seller has no knowledge) of claims asserted by any person related to the use of any patents, trademarks, service marks, trade names, copyrights, technology, know-how or processes licensed by or to Seller or challenging or questioning the validity or effectiveness of any Seller License. Neither the execution, delivery or performance of this Agreement or the Transaction Documents to which Seller is a party nor the consummation of the Transactions (i) constitute a material breach of or default under any Seller License; (ii) cause the forfeiture or termination of, or give rise to a right of forfeiture or termination of, any Intellectual Property Right; or (iii) materially impair the right of Seller to use, make, market, license, sell, copy, distribute or dispose of any Intellectual Property Right or portion thereof. There are no royalties, honoraria, fees or other payments payable by Seller to any third Person (other than salaries payable to employees and independent contractors not contingent on or related to use of their work product) as a result of the ownership, use, manufacture, marketing, license-in, sale, copying, distribution or disposition of any Intellectual Property Right by Seller, and none shall become payable as a result of the consummation
- (c) Seller has good and valid title to, or otherwise possesses adequate rights to use pursuant to the licenses listed in Section 3.7(a)(ix) of the Disclosure Schedule, all Intellectual Property necessary to permit Seller to conduct the business and operations of Seller in substantially the same manner as it had been conducted during the last fiscal year and immediately prior to the date hereof and the Assets include all such Intellectual Property.
- (d) With respect to any Trademark used by Seller in the Business (each, a "<u>Seller Mark</u>"), to the Knowledge of Seller, (i) other than with respect to the Primary Marks, Seller has made prior use in commerce throughout the Territory of any Seller Mark as to any third party, (ii) other than with respect to the Primary Marks, such use by Seller has been continuous throughout the Territory since the initial use through to the present, (iii) each Seller Mark is valid, subsisting and enforceable, and (iv) Seller's rights to each Seller Mark are superior to and enforceable against the rights of any third party. With respect to the Primary Marks, (A) Seller has made continuous use in commerce throughout the Territory of Primary Mark "A" since January 9, 2002 through to the present,

- (B) Seller has made continuous use in commerce throughout the Territory of Primary Mark "B" since January 22, 2003 through to the present, and (C) Seller has made continuous use in commerce throughout the Territory of the Logo since May 20, 2004 through to the present.
- (e) <u>Section 3.7(a)(ix)</u> of the Disclosure Schedule lists all licenses by any other person or entity to Seller of any Intellectual Property which is used in or related to the Business or is necessary for the operation of the Business as presently conducted or as proposed to be conducted (including, without limitation, any "shrink wrapped" software) and such licenses are assignable to Buyer without third party consent and shall be assigned to Buyer as of the Closing, except the "shrink wrapped" software, which is generally available to the public.
- (f) Except as set forth in Section 3.7(f) of the Disclosure Schedule, neither Seller nor any other party authorized to act on its behalf has disclosed or delivered to any party, or permitted the disclosure or delivery to any escrow agent or other party of, any Seller Source Code. No event has occurred, and no circumstance or condition exists, that (with or without notice or lapse of time, or both) shall, or would reasonably be expected to, result in the disclosure or delivery by Seller or any other party authorized to act on its behalf to any party of any Seller Source Code. Section 3.7(f) of the Disclosure Schedule identifies each Contract pursuant to which Seller has deposited, or is or may be required to deposit, with an escrow agent or other party, any Seller Source Code and further describes whether the execution of this Agreement or the Transaction Documents or the consummation of the Transactions, in and of itself, would reasonably be expected to result in the release from escrow of any Seller Source Code. As used in this Section 3.7, "Seller Source Code" means, collectively, any human readable software source code, or any material portion or aspect of the software source code, or any material proprietary information or algorithm contained in or relating to any software source code, that constitutes Seller-owned Intellectual Property Rights or any other Seller product or service marketed or currently proposed to be marketed by Seller.
- (g) The Seller Employees have taken all actions necessary, appropriate or desirable to irrevocably assign or otherwise transfer to Seller all of their respective right, title and interest in and to any Intellectual Property Rights owned by them which are necessary to permit Seller to conduct the business and operations of Seller in substantially the same manner as it has been conducted during the last fiscal year and immediately prior to the date hereof. To the Knowledge of Seller, no current or former Seller Employee: (i) is in material violation of any term or covenant of any employment contract, patent disclosure agreement, invention assignment agreement, nondisclosure agreement, noncompetition agreement or any other Contract with any other party by virtue of such Seller Employee's being employed by, or performing services for, Seller or using trade secrets or proprietary information of others without permission; (ii) has developed any technology, Software, or other copyrightable, patentable or otherwise proprietary work for Seller that (A) is subject to any Contract (written or oral) under which such Seller Employee has assigned or otherwise granted (or may be required to assign or grant) to any third party any rights (including Intellectual Property Rights) in or to such technology, Software or copyrightable, patentable or otherwise proprietary work, (B) was developed during the hours for which such Seller Employee was legally, contractually or otherwise required to be working for any third party, (C) was developed on any computers or other equipment owned by any third party or (D) is substantially similar or related to any technology, Software, or other

copyrightable, patentable or otherwise proprietary work of any third party which has employed or hired such Seller Employee. To the Knowledge of Seller, neither the employment of any employee of Seller, nor the use by Seller of the services of any consultant or Partner subjects Seller to any liability to any third party (including for improperly soliciting such employee or consultant to work for Seller). Any Partner who was employed by another party during the time when such Partner provided services to Seller represents that he or she has informed and received approval from such other party with respect to such Partner's involvement with Seller.

- (h) Neither the operation of the Business nor the use, development, manufacture, marketing, license, sale, distribution or furnishing of any Seller product or service currently used, developed, made, marketed, licensed, sold, distributed or furnished by Seller (i) violates any Contract between Seller and any third party, or (ii) to the Knowledge of Seller, infringes or misappropriates any third party rights, including without limitation, property rights, intellectual property rights, moral rights and privacy rights. Neither the use, development, manufacture, marketing, license, sale, distribution or furnishing of any Seller product or service currently under development by Seller (i) violates any Contract between Seller and any third party, or (ii) infringes or misappropriates any third party rights, including without limitation, property rights, intellectual property rights, moral rights and privacy rights.
- (i) Seller has not received any written threat, demand or notice of any claim from any person or entity asserting that Seller's use of any of the Intellectual Property or the registration thereof constitutes any infringement, interference, violation, misappropriation, breach or wrongful use of the intellectual property rights of any other person or entity and Seller is not a party to any proceeding or outstanding Order, agreement or stipulation restricting in any manner the use, transfer, sale, assignment and/or licensing by Seller of any Intellectual Property, or which may affect the validity, use or enforceability of such Intellectual Property by Seller. All Intellectual Property used by Seller in the Business is valid and enforceable. Seller has received no communications from any third parties raising the possibility of licensing discussions for intellectual property rights held by such third party, which Seller reasonably suspects or believes are made due to the third party's belief that Seller has infringed or misappropriated such party's intellectual property rights.
- (j) The disclosure by Seller, and the use and further disclosure by Buyer to Affiliates of Buyer or other Persons, of the Intellectual Property assigned to Buyer hereunder will not violate any privacy policy, data-sharing agreement, confidentiality agreement or non-disclosure agreement of which Seller is a party, or infringe or violate the privacy rights of any party.
- (k) Except as set forth in $\underbrace{Section\ 3.7(k)}$ of the Disclosure Schedule, all of the Intellectual Property, other than the licensed Intellectual Property identified in $\underbrace{Section\ 3.7(a)(ix)}$ of the Disclosure Schedule, was entirely internally-developed by employees of Seller or validly transferred by such employees to Seller, and all inventorship, ownership, authorship, patent, copyright, trademark, trade secret and other rights relating to such Intellectual Property are owned by Seller free and clear of any obligation or agreement to any other person and will not, as a result of the transactions contemplated by this Agreement, be the subject of any obligation or agreement to any third party. Seller has not transferred, sold, and/or assigned any of its ownership rights in such Intellectual Property to any person or entity.

- (I) Seller has taken all steps necessary, appropriate or desirable (including, without limitation, entering into appropriate confidentiality, non-disclosure and non-competition agreements with all officers, employees and consultants of Seller with access to or knowledge of Seller's patents, Trademarks, copyrights, inventions, Trade Secrets, Software licenses and other proprietary information) to safeguard and maintain the secrecy and confidentiality of, and its proprietary rights in, all of Seller's Intellectual Property. All current and former Seller Employees having access to proprietary information of Seller, its Customers or business partners and inventions owned by Seller have executed and delivered to Seller an agreement regarding the protection of such proprietary information and the assignment of inventions to Seller (in the case of proprietary information of Seller's Customers and business partners, to the extent required by such Customers and business partners); and copies of all such agreements have been delivered to Buyer and, to the Knowledge of Seller, no such individual is in breach of such agreement. Section 3.7(!) of the Disclosure Schedule lists all agreements relating to the protection of such proprietary information and the assignment of inventions. Seller has secured valid written assignments from all current and former Seller Employees who were involved in, or who contributed to, the creation or development of any Intellectual Property Rights, of the rights to such contributions that may be owned by such persons or that Seller does not already own by operation of law. No current or former Seller Employee has any right, license, claim or interest whatsoever in or with respect to any Intellectual Property Rights. No current or former Seller Employee has (i) excluded works or inventions made prior to his or her employment with Seller from his or her assignment of inventions agreement or (ii) failed to affirmatively indicate in such proprietary information and inventions agreement with Seller
- (m) Seller has not disposed of or abandoned or permitted to lapse any Intellectual Property Rights or disclosed to any person other than (i) representatives of Buyer or (ii) third parties subject to confidentiality or non-disclosure agreements, copies of which have been delivered to Buyer, any Trade Secret, proprietary information, formula, process or know-how of Seller not theretofore a matter of public knowledge.
- (n) Section 3.7(n) of the Disclosure Schedule lists all software, documentation or other material that is distributed as "free software", "open source software" or under a similar licensing or distribution model (including, without limitation, the GNU General Public License (GPL), GNU Lesser General Public License (LGPL), GNU Free Documentation License (FDL), Mozilla Public License (MPL), BSD licenses, the Artistic License, the Netscape Public License, the Sun Community Source License (SCSL) the Sun Industry Standards License (SISL), the Apache License, or any Creative Commons License) (collectively, "Open Source Materials") used by Seller in any way, and describes the manner in which such Open Source Materials were used (such description shall include, without limitation, whether (and, if so, how) the Open Source Materials were modified and/or distributed by Seller). Seller has not (i) incorporated Open Source Materials into, or combined Open Source Materials with, any Intellectual Property; (ii) distributed Open Source Materials in conjunction with or for use with any Intellectual Property; (iii) acquired any software from a third party that was or has become or that includes, as a result of such acquisition, Open Source Materials; or (iv) incorporated into Intellectual Property,

combined or distributed with Intellectual Property, acquired or used any Open Source Materials that create, or purport to create, obligations for Seller with respect to Intellectual Property, or grant, or purport to grant, to any third party, any rights or immunities under Intellectual Property (including, without limitation, having incorporated into, combined or distributed with, acquired or used any Open Source Materials that require, as a condition of use, modification and/or distribution of such Open Source Materials or other software incorporated into, derived from or distributed with such Open Source Materials, that it be (x) disclosed or distributed in source code form, (y) be licensed for the purpose of making derivative works, or (z) be redistributable at no charge or with any restriction on the consideration charged therefor).

- (o) Section 3.7(o) of the Disclosure Schedule sets out the physical location of the computer servers that are currently hosting Seller websites. Such servers are validly owned or leased by Seller. Section 3.7(o) of the Disclosure Schedule also sets out any applicable Internet hosting Contract including the term of the Contract, associated costs, corporate information of the host and amount of bandwidth to which the server is connected to the Internet. In addition, Section 3.7(o) of the Disclosure Schedule sets out the name and IP address of Seller's Internet web homepage and lists all similar names and addresses owned by Seller, the date the homepage was granted and the date of the next annual payment. Seller's websites contain all legal disclaimers that, in accordance with industry practice, are customarily contained on websites similar to Seller's websites.
- (p) Seller has entered into appropriate agreements, provided legally enforceable disclaimers and has otherwise conducted the Business and taken all actions necessary to eliminate its liability for software or other content developed by third parties that is posted on its website ("<u>Uploaded Materials</u>"). No claim has been made against, and no communication has been received by, Seller from any use of the Seller services or content alleging that Seller is liable to such user for any Uploaded Materials under any theory of liability, from third parties alleging that Seller is liable for infringement or misappropriation of its Intellectual Property Rights.
- 3.8 Material Contracts. Section 3.8 of the Disclosure Schedule lists each Contract to which the Assets are subject, by which the Assets are bound or to which Seller is a party which involves the expenditure or receipt by Seller, (a) with respect to any such Contract with a healthcare facility Customer, of \$50,000 or more annually (each, a "Facility Contract"), or (b) with respect to all other such Contracts, of \$10,000 or more annually (each such Contract, including any Facility Contract, a "Material Contract"). Each Material Contract is valid and subsisting; Seller has duly performed all its obligations thereunder to the extent that such obligations to perform have accrued; and, subject to the receipt of those third party waivers and/or consents identified on Schedule 5.12 hereto, no breach or default, alleged breach or default, or event which would (with the passage of time, notice or both) constitute a breach or default thereunder by Seller, or, to the Knowledge of Seller, any other party or obligor with respect thereto, has occurred or as a result of the Transactions will occur. True copies of all Material Contracts listed in Section 3.8 of the Disclosure Schedule, including all amendments and supplements thereto but excluding Facility Contracts to which Seller is a party which involve the receipt by Seller of less than \$100,000 annually, have been delivered to Buyer. All Material Contracts are in written form. Other than the Material Contracts identified in Section 3.8 of the Disclosure Schedule, there are no Material Contracts to which the Assets are subject, by which the Assets are bound or to which Seller is a party.

3.9 Tax Matters.

- (a) All Tax Returns required to be filed by or on behalf of Seller have been duly and timely filed with the appropriate Tax authority in all jurisdictions in which such Tax Returns are required to be filed (after giving effect to any valid extensions of time in which to make such filings), and all such Tax Returns are true, complete and correct in all material respects. All Taxes payable by or on behalf of Seller (whether or not shown on a Tax Return) have been fully and timely paid. With respect to any period for which such Tax Returns have not yet been filed or for which such Taxes are not yet due or owing, Seller has made due and sufficient accruals for such Taxes on the Seller Balance Sheet. All required estimated Tax payments have been timely made by or on behalf of Seller.
- (b) As of the Balance Sheet Date, Seller has no liability for unpaid Taxes, other than Taxes for which adequate reserves have been established. Since the Balance Sheet Date, Seller has not incurred any liability for Taxes arising from extraordinary gains or losses, as that term is used in GAAP, outside the ordinary course of business consistent with past custom and practice.
- (c) Seller has complied with all applicable Laws relating to the payment and withholding of Taxes and has duly and timely withheld and paid over to the appropriate Tax authority all amounts required to be so withheld and paid under all applicable Legal Requirements.
- (d) Buyer has received complete copies of (i) all material Tax Returns of or including Seller for Tax periods ending after December 31, 2002, and (ii) any audit, report or other similar correspondence issued relating to Taxes of Seller for such periods.
- (e) No claim has been made by a Tax authority in a jurisdiction where Seller does not file a Tax Return that it is or may be subject to taxation in that jurisdiction.
- (f) No Tax Return of Seller has ever been audited by the Internal Revenue Service or any other Tax authority, no such audit is in progress and Seller has not been notified of any request for such an audit or other examination. No deficiencies for any Tax have been threatened, claimed, proposed or assessed against Seller which have not been settled or paid. No federal, state, local or foreign audits, examinations, investigations or other administrative proceedings or court proceedings are presently pending with regard to any Taxes or Tax Returns filed by or on behalf of Seller. No issue has been raised by any Tax authority in any prior examination of Seller which, by application of the same or similar principles, could reasonably be expected to result in a proposed deficiency for any subsequent Tax period. No adjustment relating to any Tax Returns filed by Seller has been proposed in writing by a Tax authority to Seller (or any representative thereof).
- (g) There is not in effect any waiver by Seller of any statute of limitations with respect to any Taxes. Seller is not a foreign person within the meaning of Section 1445(f)(3) of the Code.

- (h) There are no liens (other than Permitted Encumbrances) as a result of any unpaid Taxes upon any of the assets of Seller.
- (i) There are no "nonqualified deferred compensation plans" (within the meaning of Section 409A of the Code) to which Seller is a party and that Buyer will assume or become obligated to make payments under as a result of the Transaction. No event has occurred that would be treated by Section 409A(b) as a transfer of property for purposes of Section 83 of the Code.

3.10 Seller Financial Statements.

- (a) Section 3.10(a) of the Disclosure Schedule includes the Seller Financial Statements. The Seller Financial Statements: (i) are derived from and are in accordance with the books and records of Seller; and (ii) fairly present the financial condition of Seller at the dates therein indicated and the results of operations and cash flows of Seller for the periods therein specified. The reserves, if any, reflected on the Seller Financial Statements are adequate in light of the contingencies with respect to which they were made and, without limiting the foregoing, Seller has established adequate reserves for medical malpractice liability and tax liability on the Seller Balance Sheet. Seller has no liabilities, except for (A) liabilities included in the Seller Balance Sheet, (B) liabilities that were incurred after the Balance Sheet Date in the ordinary course of Seller's business consistent with its past practices, and (C) liabilities which are set forth in Section 3.10(a) of the Disclosure Schedule.
- (b) Seller has established and maintains a system of internal accounting controls that is appropriate for a privately-held corporation similarly situated with Seller in size and scope of operations. Neither Seller, nor to the Knowledge of Seller, any Seller Employee, has identified or been made aware of any fraud, whether or not material, that involved Seller's management or other Seller Employees who have a role in the preparation of the financial statements or the internal accounting controls utilized by Seller, or any claim or allegation regarding any of the foregoing. Neither Seller nor, to the Knowledge of Seller, any Seller Employee, auditor, accountant or representative of Seller has received or otherwise had or obtained knowledge of any material complaint, allegation, assertion or claim, whether written or oral, in each case, regarding deficient accounting or auditing practices, procedures, methodologies or methods of Seller or its internal accounting controls or any material inaccuracy in Seller's financial statements. No attorney representing Seller, whether or not employed by Seller, has reported to the management of Seller or any Partner or any committee thereof evidence of a material violation of securities laws, breach of fiduciary duty or similar violation by Seller or any of its respective Partners, officers, managers, employees or agents. At the Balance Sheet Date, there were no material loss contingencies (as such term is used in Statement of Financial Accounting Standards No. 5 ("Statement No. 5") issued by the Financial Accounting Standards Board in March 1975) that are not adequately provided for in the Seller Balance Sheet as required by Statement No. 5. There has been no change in the Seller accounting policies since the Balance Sheet Date.
- (c) Section 3.10(c) of the Disclosure Schedule sets forth the names and locations of all banks, trust companies, savings and loan associations and other financial institutions or third parties including related parties at which Seller maintains accounts of any nature and the names of all persons authorized to draw thereon or make withdrawals therefrom.

- (d) <u>Schedule 3.10(d)</u> of this Agreement lists all Debt of Seller, including, for each item of Debt, the Contract governing the Debt and the interest rate, maturity date and any assets or properties securing such Debt. All Debt may be prepaid at or prior to the Closing without penalty under the terms of the Contracts governing such Debt.
- 3.11 <u>Legal Proceedings</u>. Except as set forth in <u>Section 3.11</u> of the Disclosure Schedule, there is no Order or Action pending, or, to the Knowledge of Seller, threatened, against or affecting Seller, the Assets or the Business. <u>Section 3.11</u> of the Disclosure Schedule lists each Order and each Action that involves a claim or potential claim against, or that enjoins or seeks to enjoin, prevent, delay, make illegal or impose limitations or conditions or otherwise interfere with any activity or action by Seller or any Affiliate, including, without limitation, execution and delivery of the Transaction Documents and performance of Seller's or the Partners' obligations thereunder. There is no matter as to which Seller or any Affiliate has received any notice, claim or assertion, or which otherwise has been threatened or is reasonably expected to be threatened or initiated, against or affecting Seller, the Assets or the Business.
- 3.12 <u>Title to Properties</u>. Seller has good and marketable title to all of the Assets (including those shown on the Seller Balance Sheet) free and clear of all Encumbrances. Such Assets are sufficient for the continued operation of the Business. <u>Section 3.12</u> of the Disclosure Schedule sets forth a list of each fixed asset of Seller included in the Seller Balance Sheet. All properties used in the operations of the Business are reflected on the Seller Balance Sheet. All Tangible Personal Property owned or leased by Seller or used in the Business is in good condition and repair, normal wear and tear excepted. All leases of real and personal property to which Seller is a party are fully effective and afford Seller a valid leasehold possession of the real or personal property that is the subject of the lease. <u>Section 3.12</u> of the Disclosure Schedule also sets forth a complete and accurate list of Contracts governing the lease of any personal property by Seller, including the title, date of and the parties to each such Contract and, if reasonably necessary for understanding, a brief description of the item(s) of personal property leased thereby. <u>Section 3.12</u> of the Disclosure Schedule also identifies each parcel of real property leased or sub-leased by Seller and the Contract pursuant to which such real property is leased or sub-leased. Seller has adequate ingress and egress into any real property used in the operation of the Business. A true and complete copy of each Contract under which Seller uses or occupies or has the right to use or occupy, now or in the future, any real property or facility, has been delivered to Buyer. Each such Contract is in written form. Seller does not own any real property. Following the Closing and upon delivery of the Bill of Sale and Assignment of Contract Rights, Buyer will have good and marketable title to, free of Encumbrances, all of the Assets.
 - 3.13 <u>Absence of Certain Changes</u>. Since the Balance Sheet Date, there has not been with respect to Seller any:
- (a) Material Adverse Change or any change, event, circumstance, condition or effect that would reasonably be expected to result in a Material Adverse Change;

- (b) amendment to or change in its certificate of limited partnership or the Limited Partnership Agreement;
- (c) incurrence, creation or assumption of (i) any Encumbrance on any of its assets or properties, (ii) any liability for borrowed money other than Seller Closing Debt identified on <u>Schedule 2.2(b)</u>, or (iii) any liability as a guarantor or surety with respect to the obligations of others;
- (d) payment or discharge of any Encumbrance on any of its assets or properties, or payment or discharge of any of its liabilities, in each case that was not shown on the Seller Balance Sheet;
- (e) purchase, license, sale, grant, assignment or other disposition or transfer, or any agreement or other arrangement for the purchase, license, sale, assignment or other disposition or transfer, of any of its assets (including Seller's right in its Intellectual Property and other intangible assets), properties or goodwill other than the sale or non-exclusive license of its products or services to its Customers in the ordinary course of its business consistent with its past practices;
 - (f) material damage, destruction or loss of property or assets, whether or not covered by insurance;
- (g) change or increase in the compensation payable or to become payable to any current employee of Seller or any Partner, or in any bonus, pension, severance, retention, insurance or other benefit payment or arrangement made to or with any employee of Seller or any Partner, other than such change or increase made with respect to a non-officer employee of Seller in the ordinary course of Seller's business consistent with its past practices;
- (h) change with respect to its management, supervisory or other key personnel, any termination of employment of a material number of employees, or any labor dispute or claim of unfair labor practices;
- (i) liability incurred by it to any Seller Employee, except for normal and customary compensation and expense allowances payable to Seller Employees in the ordinary course of its business consistent with its past practices;
- (j) making by it of any loan, advance or capital contribution to, or any investment in, any Seller Employee or any firm or business enterprise in which any such person had a direct or indirect material interest at the time of such loan, advance, capital contribution or investment;
- (k) entering into, amendment of, relinquishment, termination or nonrenewal by it of any Contract (or any other right or obligation), any default by it under such Contract (or other right or obligation), or any written or oral indication or assertion by the other party thereto of any material problems with its services or performance under such Contract (or other right or obligation) or such other party's desire to so amend, relinquish, terminate or not renew any such Contract (or other right or obligation);

- (1) change in the manner in which it extends discounts, credits or warranties to Customers or otherwise deals with its Customers;
- (m) any labor trouble or adverse relations with Seller Employees;
- (n) any deferral of the payment of any accounts payable;
- (o) any change which would adversely affect the validity, use, priority or enforceability of Seller's rights to any Seller Mark; or
- (p) announcement of, any negotiation with respect to or any entry into any Contract to do any of the things described in the preceding clauses (a) through (o) (other than negotiations and agreements with Buyer and its representatives regarding the transactions contemplated by this Agreement).

3.14 No Default; No Restrictions.

- (a) Each of the Material Contracts is in full force and effect. There exists no material default or event of default or event, occurrence, condition or act, with respect to Seller or, to the Knowledge of Seller, with respect to any other contracting party, which, with the giving of notice, the lapse of time or the happening of any other event or conditions, would reasonably be expected to (i) become a default or event of default under any Material Contract or (ii) give any third party (1) the right to declare a default or exercise any remedy under any Material Contract, (2) the right to a rebate, chargeback, refund, credit, penalty or change in delivery schedule under any Material Contract, (3) the right to accelerate the maturity or performance of any obligation of Seller under any Material Contract, or (4) the right to cancel, terminate or modify any Material Contract. Seller has not received any written or oral notice or other communication regarding any actual or possible violation or breach of or default under any Material Contract. Seller has no liability for renegotiation of government contracts or subcontracts.
- (b) Seller is not a party to, and no asset or property of Seller is bound or affected by, any judgment, injunction, order or decree, that restricts or prohibits Seller or, following the Closing Date, will restrict or prohibit Buyer from freely engaging in the Business or from competing anywhere in the world (including any judgments, injunctions, orders or decrees, restricting the geographic area in which Seller or, following the Closing Date, Buyer may sell, license, market, distribute or support any products or technology or provide services or restricting the markets, customers or industries that Seller or, following the Closing Date, Buyer may address in operating the Business or restricting the prices which Seller or, following the Closing Date, Buyer may charge for its products, technology or services (including most favored customer pricing provisions)), or includes any grants by Seller of exclusive rights or licenses, rights of refusal, rights of first negotiation or similar rights.

3.15 Compliance with Laws.

(a) Seller has complied in all material respects, and is now in material compliance, with all applicable Law.

- (b) All materials, products and services distributed or marketed by Seller have at all times made all material disclosures to users or Customers required by applicable Law, and none of such disclosures made or contained in any such materials have been inaccurate, misleading or deceptive in any material respect.
- (c) Seller holds all material Permits and Approvals, and has made all material filings with, Governmental Entities that are necessary and/or legally required to be held or made by it to conduct the Business without any violation of applicable Law, and all such Permits and Approvals are valid and in full force and effect and are listed in Section 3.15(c) of the Disclosure Schedule. Seller has not received any notice or other communication from any Governmental Entity regarding (i) any actual or possible violation of law or any Permit or Approval or any failure to comply with any term or requirement of any Permit or Approval or (ii) any actual or possible revocation, withdrawal, suspension, cancellation, termination or modification of any Permit or Approval.
- (d) Neither Seller nor any Seller Employee has, for or on behalf of Seller, (i) used any funds for unlawful contributions, gifts, entertainment or other unlawful expenses relating to political activity, (ii) made any unlawful payment to foreign or domestic government officials or employees or to foreign or domestic political parties or campaigns or violated any provision of the Foreign Corrupt Practices Act of 1977, as amended, or (iii) knowingly made any other payment in violation of applicable Law.
- 3.16 <u>Certain Transactions and Agreements</u>. None of the Partners or any immediate family member of any Partner, has a direct ownership interest of more than 1% of the equity ownership of any firm or corporation that competes with, or does business with, or has any contractual arrangement with, Seller. None of the Partners or any immediate family member of any Partner, is a party to, or otherwise directly or indirectly interested in any Material Contract.

3.17 Employees and Other Compliance.

- (a) Section 3.17(a) of the Disclosure Schedule lists each Seller Benefit Arrangement. A copy of each document reflecting each Seller Benefit Arrangement has been provided to Buyer or Buyer's counsel, including any summary plan descriptions, prospectuses, trust agreements, form agreements and other documents related to such plans and arrangements that have been provided or made available to any Seller Employee. The terms and conditions of any Seller Benefit Arrangement that is not in writing are set forth in Section 3.17(a) of the Disclosure Schedule.
- (b) All Seller Benefit Arrangements that are intended to be subject to Code Section 401(a) and any trust agreement that is intended to be tax exempt under Code Section 501(a) have been determined by the Internal Revenue Service to be qualified under Code Section 401(a) and exempt from taxation under Code Section 501(a), and, to the Knowledge of Seller, nothing has occurred that would adversely affect the qualification of any such plan. Each Seller Benefit Arrangement and any related trust subject to ERISA complies in all material respects with and has been administered in substantial compliance with, (A) the provisions of ERISA, (B) all provisions of the Code, (C) all other applicable Laws, and (D) its terms and the terms of any collective bargaining or collective labor agreement. Seller has not received any written notice from any Governmental Entity questioning or challenging such compliance.

There are no unresolved claims or disputes under the terms of, or in connection with, the Seller Benefit Arrangements other than claims for benefits which are payable in the ordinary course. There has not been any non-exempt "prohibited transaction" (within the meaning of Section 406 of ERISA or Section 4975 of the Code) with respect to any Seller Benefit Arrangement. No litigation has been commenced with respect to any Seller Benefit Arrangement and, to the Knowledge of Seller, no such litigation is threatened (other than routine claims for benefits in the ordinary course). There are no governmental audits or investigations pending or, to the Knowledge of Seller, threatened in connection with any Seller Benefit Arrangement. To the Knowledge of Seller, there are not any facts that could give rise to any material liability in the event of any governmental audit or investigation.

- (c) Seller has timely filed and delivered or made available to Buyer and its legal counsel the three most recent annual reports (Form 5500) and all schedules attached thereto for each Seller Benefit Arrangement that will be assumed by Buyer and for which such report was required as well as copies of any material correspondence with the IRS, the U.S. Department of Labor, or any other Governmental Entity relating to any such Seller Benefit Arrangement.
- (d) All contributions due from Seller with respect to any of the Seller Benefit Arrangements have been made or have been accrued on Seller's financial statements (including the Seller Financial Statements), and, except as set forth in Section 3.17(d) of the Disclosure Schedule, no further contributions from Seller shall be due or shall have accrued thereunder as of the Closing Date (other than contributions accrued in the ordinary course of business, consistent with past practices, after the Balance Sheet Date as a result of the operations of Seller after the Balance Sheet Date). All claims as of the Closing Date made under any self-insured Seller Benefit Arrangement that is an "employee welfare benefit plan" as defined in Section 3(1) of ERISA have been timely paid or, if not yet paid, will be paid in a timely manner by Seller.
- (e) Neither Seller nor any ERISA Affiliate of Seller (i) has an "obligation to contribute" (as defined in ERISA Section 4212) to a Seller Benefit Arrangement that is a "multiemployer plan" (as defined in ERISA Sections 4001(a)(3) and 3(37)(A)); (ii) sponsors, maintains or contributes to any plan, program or arrangement that provides for post-retirement or other post-employment welfare benefits (other than health care continuation coverage as required by applicable Law); or (iii) maintains or sponsors or has ever maintained or sponsored any compensation or benefit plan subject to the law or applicable custom or rule of any jurisdiction outside of the United States. Neither Seller nor any ERISA Affiliate has ever maintained, established, sponsored, participated in, or contributed to, any defined benefit plan (as defined in ERISA Section 3(35)) subject to Part 3 of Subtitle B of Title I of ERISA, Title IV of ERISA or a pension plan (as defined in Section 3(2) of ERISA) subject to Section 412 of the Code.
- (f) All individuals who, pursuant to the terms of any Seller Benefit Arrangement, are entitled to participate in any Seller Benefit Arrangement, are currently participating in such Seller Benefit Arrangement or have been offered an opportunity to do so and have declined in writing.
 - (g) Seller is not a party to any agreement or arrangement whereby Seller Employees are "leased" from a third party.

- (h) There has been no amendment to, written interpretation or announcement (whether or not written) by Seller relating to, or change in employee participation or coverage under, any Seller Benefit Arrangement that would increase materially the expense of maintaining such Seller Benefit Arrangement above the level of the expense incurred in respect thereof during Seller's fiscal year ending December 31, 2007, except any such amendments that are required under applicable Law.
- (i) Each Seller Benefit Arrangement, to the extent applicable, is in compliance, in all material respects, with the continuation coverage requirements of Section 4980B of the Code, Sections 601 through 608 of ERISA, the Americans with Disabilities Act of 1990, as amended, and the regulations thereunder, the Health Insurance Portability and Accountability Act of 1996, as amended, the Women's Health and Cancer Rights Act of 1998, and the Family Medical Leave Act of 1993, as amended, and the regulations thereunder, and all similar state laws, as such requirements affect Seller and its employees. There are no outstanding, uncorrected violations under the Consolidation Omnibus Budget Reconciliation Act of 1985, as amended ("COBRA"), (or similar state law) with respect to any of the Seller Benefit Arrangements, covered employees or qualified beneficiaries that would be reasonably likely to result in a Material Adverse Effect on Seller.
- (j) No benefit payable or that may become payable by Seller pursuant to any Seller Benefit Arrangement or as a result of, in connection with or arising under this Agreement or pursuant to the Transactions shall constitute a "parachute payment" (as defined in Section 280G(b)(2) of the Code) that is subject to the imposition of an excise tax under Section 4999 of the Code or that would not be deductible by reason of Section 280G of the Code. Unless otherwise indicated in Section 3.17(j) of the Disclosure Schedule, Seller is not a party to any: (i) Contract with any executive officer or other key employee thereof (A) the benefits of which are contingent, or the terms of which are materially altered, upon the occurrence of a transaction involving Seller in the nature of the Transactions or any Transaction Agreement, (B) providing any term of employment or compensation guarantee, or (C) providing severance benefits or other benefits after the termination of employment of such employee regardless of the reason for such termination of employment other than as required by COBRA (or similar state laws), vacation pay cash-outs or other arrangements governed by ERISA; or (ii) Contract or plan, including any stock option plan, stock appreciation rights plan or stock purchase plan, any of the benefits of which shall be increased, or the vesting of benefits of which shall be accelerated, by the occurrence of the Transaction or any of the other transactions contemplated by this Agreement, or any event subsequent to the Transaction such as the termination of employment of any person, or the value of any of the benefits of which shall be calculated on the basis of any of the transactions contemplated by this Agreement. Seller does not have any obligation to pay any material amount or provide any material benefit to any former Seller Employee, other than obligations (i) for which Seller has established a reserve for such amount on the Seller Balance Sheet and (ii) pursuant to Contracts entered into after the Balance Sheet Date a
- (k) Seller is in compliance in all material respects with all applicable Law and Contracts relating to employment, employment practices, immigration, wages, hours, and terms and conditions of employment, including employee compensation matters, and has correctly classified employees of Seller as exempt employees and nonexempt employees under the Fair Labor

Standards Act. A complete list of all Seller Employees and their current title and/or job description and compensation (base compensation and bonuses) is set forth in Section 3.17(k) of the Disclosure Schedule. To the Knowledge of Seller, all employees of Seller are legally permitted to be employed by Seller in the jurisdiction in which such employee is employed in their current job capacities for the maximum period allowed under applicable Law. All allied healthcare professionals, including, without limitation, rehabilitation therapists, radiology technicians and physical therapists and other staff performing work for Seller have been properly classified as employees or independent contractors, as applicable, for purposes of federal and applicable state tax laws, laws applicable to employee benefits and other applicable Law. Seller has no employment or consulting Contracts currently in effect that are not terminable at will (other than agreements with the sole purpose of providing for the confidentiality of proprietary information or assignment of inventions or for providing for the noncompetition covenant of the employee or consultant party thereto).

- (l) To the Knowledge of Seller, Seller is not now, nor has ever been, subject to a union organizing effort. Seller is not subject to any collective bargaining agreement with respect to any of its employees, subject to any other Contract with any trade or labor union, employees' association or similar organization, and subject to any current labor disputes. Seller has good labor relations, and Seller has no knowledge of any facts indicating that the consummation of the Transactions shall have a material adverse effect on such labor relations, and has no knowledge that any of the key employees of Seller intends to leave their employ with Seller.
- (m) To the Knowledge of Seller, no employee or consultant of Seller is in material violation of (i) any term of any employment or consulting Contract or (ii) any term of any other Contract or any restrictive covenant relating to the right of any such employee or consultant to be employed by Seller or to use trade secrets or proprietary information of others. To the Knowledge of Seller, the employment of any employee or consultant by Seller does not subject it to any liability to any third party other than liabilities with respect to employer payroll tax and employee tax withholding.
- (n) Seller has not established any compensation and benefit plan that is maintained or is required to be maintained or contributed to by the law or applicable custom or rule of the relevant jurisdiction, outside of the United States.
- (o) Seller is in compliance in all material respects with the Worker Adjustment Retraining Notification Act of 1988, as amended ("WARN Act"), or any similar state or local law. In the past two years, (i) Seller has not effectuated a "plant closing" (as defined in the WARN Act) affecting any site of employment or one or more facilities or operating unit within any site of employment or facility of its business; (ii) there has not occurred a "mass layoff" (as defined in the WARN Act) affecting any site of employment or facility of Seller; and (iii) Seller has not been affected by any transaction or engaged in layoffs or employment terminations sufficient in number to trigger application of any similar state, local or foreign law or regulation. Seller has not caused any of its employees to suffer an "employment loss" (as defined in the WARN Act) during the ninety (90) day period prior to the date of this Agreement.

- (p) All 401(k) Matching Payments due and payable as of the Closing Date have been contributed by Seller to Seller's 401(k) plan and all 401(k) Matching Payments accrued (but that are not yet due and payable) as of the Closing Date will be contributed in a timely manner by Seller to Seller's 401(k) plan and allocated to the accounts of the applicable Seller Employees following the Closing Date in accordance with Seller's past practice.
- 3.18 <u>Insurance</u>. Seller maintains the policies of insurance and bonds set forth in <u>Section 3.18</u> of the Disclosure Schedule, including, without limitation, (i) professional liability insurance and errors and omissions insurance for Seller's healthcare professionals, and (ii) all legally required workers' compensation and other legally required insurance. <u>Section 3.18</u> of the Disclosure Schedule sets forth the name of the insurer under each such policy and bond, the type of policy or bond, and the coverage amount and any applicable deductible. There is no material claim pending under any of such policies or bonds as to which coverage has been questioned, denied or disputed by the underwriters of such policies or bonds. All premiums due and payable under all such policies and bonds have been timely paid, and Seller is otherwise in compliance with the terms of such policies and bonds. All such policies and bonds remain in full force and effect, and Seller has no knowledge of any threatened termination of, or material premium increase with respect to, any of such policies or bonds. Seller has delivered to Buyer correct and complete copies of all such insurance binders or policies of insurance and bonds issued at the request, or for the benefit, of Seller.
- 3.19 Environmental Matters. Seller is not in violation of any applicable statute, law or regulation relating to the environment or occupational health and safety, and to its knowledge, no material expenditures are or will be required in order to comply with any such existing statute, law or regulation. No Hazardous Materials (as defined below) are used or have been used, stored, or disposed of by Seller or, to the Knowledge of Seller, by any other person or entity on any property owned, leased or used by Seller. For the purposes of the preceding sentence, "Hazardous Materials" shall mean materials which are listed or otherwise defined as "hazardous" or "toxic" under any applicable local, state, federal and/or foreign laws and regulations that govern the existence and/or remediation of contamination on property, the protection of the environment from contamination, the control of hazardous wastes, or other activities involving hazardous substances, including building materials or any petroleum products or nuclear materials.
- 3.20 No Existing Discussions. Neither Seller, any Partner nor, to the Knowledge of Seller, any Seller Employee or agent of Seller (or any investment banker, broker, finder or similar party) of Seller is engaged, directly or indirectly, in any discussions or negotiations with any third party (other than Buyer) regarding the merger or other business combination of Seller with or into any corporation or other entity, the sale, conveyance or disposition of all or substantially all of the assets of Seller or a transaction or series of related transactions in which more than fifty percent (50%) of the voting power of Seller would be disposed of, or any other form of liquidation, dissolution or winding up of Seller.
- 3.21 <u>Privacy</u>. Seller has provided notice of its privacy practices on its websites and these notices have not contained any material omissions of Seller's privacy practices since inception. The privacy policy or policies providing this notice and the periods that each policy has been in effect are set forth in <u>Section 3.21</u> of the Disclosure Schedule. Seller's privacy practices conform, and at all times

that such policies have been in effect have conformed, in all respects to its respective privacy policies. Seller has complied with all applicable Law relating to (a) the privacy of users of Seller's products or services and Seller websites, and (b) the collection, use, storage and disclosure of any personally identifiable information collected by Seller or by third parties acting on Seller's behalf or having authorized access to Seller's records. Seller's privacy policies conform, and at all times have conformed, to all of Seller's contractual commitments to its Customers and the viewers of Seller's websites. Each Seller website and all materials distributed or marketed by Seller have at all times made all disclosures to users or Customers required by applicable Law, and none of such disclosures made or contained in Seller's website or in any such materials have been inaccurate, misleading or deceptive or in violation of any applicable Law. No claims have been asserted or, to the Knowledge of Seller, are threatened against Seller by any person or entity alleging a violation of any person's or entity's privacy, personal or confidentiality rights under the privacy policies of Seller. With respect to all personal and user information collected by Seller in connection with Seller's websites, Seller has at all times taken all steps reasonably necessary (including, without limitation, implementing and monitoring compliance with adequate measures with respect to technical and physical security) to ensure that the information is protected against loss and against unauthorized access, use, modification, disclosure or other misuse. To the Knowledge of Seller, there has been no unauthorized access to or other misuse of that information. To the Knowledge of Seller, except as specified in Section 3.21 of the Disclosure Schedule, Seller has not collected or obtained any personally identifiable information from any third party not otherwise registered as a user of Seller's website. The execution, delivery and performance of

- 3.22 Accounts Receivable. Section 3.22 of the Disclosure Schedule sets forth an accurate and complete list of Seller's accounts receivable as of December 31, 2007 and as of February 11, 2008 in the aggregate and by Customer, and indicates the aging of such accounts receivable and the amounts of allowances for doubtful accounts. In addition, Section 3.22 of the Disclosure Schedule sets forth such amounts of accounts receivable of Seller as of December 31, 2007 and as of February 11, 2008 which were or are, as applicable, subject to claims asserted by Customers and reasonably detailed information regarding claims made within the preceding twelve months, including the type and amount of each such claim. All accounts receivable as of the Closing Date reflected in Section 3.22 of the Disclosure Schedule are collectible in full following the date of this Agreement in the ordinary course of the Business.
- 3.23 <u>Transaction Expenses</u>. Neither Seller, the Partners nor any of their respective Affiliates is obligated for the payment of any fees or expenses of any investment banker, broker, finder or similar party in connection with the origin, negotiation or execution of this Agreement or in connection with any of the Transactions. The legal and accounting advisors, and any other persons, to whom Seller or the Partners currently expects to owe fees and expenses that will constitute Transaction Expenses are set forth in <u>Section 3.23</u> of the Disclosure Schedule, and, other than the Transaction Expenses that will be due to the entities set forth in Section 3.23 of the Disclosure Schedule, there are no Transaction Expenses.
- 3.24 <u>Accuracy of Information</u>. OTHER THAN THE REPRESENTATIONS AND WARRANTIES MADE BY SELLER AND THE PARTNERS IN THIS AGREEMENT, THE OTHER TRANSACTION DOCUMENTS AND THE CERTIFICATES DELIVERED BY SELLER PURSUANT TO ARTICLE 2 AND/OR ARTICLE 6 OF THIS AGREEMENT, SELLER AND THE

PARTNERS MAKE NO REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED (INCLUDING THOSE REFERRED TO IN THE UNIFORM COMMERCIAL CODE OR IN ANY STATUTE OR RULE OF LAW THAT CAN BE LIMITED OR WAIVED AND WOULD OTHERWISE BE APPLICABLE TO REAL PROPERTY), AND THE ASSETS AND BUSINESS OF SELLER SHALL BE DEEMED TO BE "AS IS, WHERE IS" ON THE CLOSING DATE, AND IN THE THEN-PRESENT CONDITION.

3.25 <u>Fair Consideration</u>; <u>No Fraudulent Conveyance</u>. Seller is not now and Seller will not be rendered insolvent by the sale, transfer and assignment of the Assets pursuant to the terms of this Agreement or the Transactions. Seller has no intention to file for bankruptcy, and, to the Knowledge of Seller, no insolvency proceedings of any character including, without limitation, bankruptcy, receivership, reorganization, composition or arrangement with creditors, voluntary or involuntary, affecting Seller or any of the Assets or Assumed Liabilities are pending or threatened. Seller and the Partners are not entering into this Agreement and the Transactions with the intent to defraud, delay or hinder their creditors and the consummation of the transactions contemplated by this Agreement and the Transactions will not have any such effect. The Transactions will not constitute a fraudulent conveyance, or otherwise give rise to any right of any creditor of Seller or the Partners whatsoever to any of the Assets after the Closing.

ARTICLE 4 REPRESENTATIONS AND WARRANTIES OF BUYER AND GUARANTOR

Buyer and Guarantor, jointly and severally, represent and warrant to Seller and the Partners as follows:

- 4.1 <u>Organization, Corporate Power and Authority</u>. Each of Buyer and Guarantor is a corporation duly organized, validly existing and in good standing under the laws of the state in which it is incorporated and Buyer is duly qualified to do business as a foreign corporation in the jurisdictions in which it conducts its business, except where the failure to so qualify will not have a Material Adverse Effect on Buyer's ability to perform its obligations under the Transaction Documents. Each of Buyer and Guarantor has all requisite corporate power and authority to execute and deliver the Transaction Documents to which it is a party and to perform its obligations thereunder.
- 4.2 <u>Authorization of Agreement</u>. The execution, delivery and performance by each of Buyer and Guarantor of the Transaction Documents to which it is a party, and the consummation by it of the Transactions, have been duly authorized by all necessary corporate action by each of Buyer and Guarantor. This Agreement has been, and each other Transaction Document to which Buyer or Guarantor, respectively, is a party will be at the Closing, duly executed and delivered by Buyer and Guarantor, as applicable, and constitute, or will, when delivered, constitute, the legal, valid and binding obligations of Buyer and Guarantor, as applicable, enforceable against Buyer and Guarantor, as applicable, in accordance with their respective terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar laws relating to or limiting creditors' rights generally and to the principles of equity (regardless of whether considered in a proceeding in equity or an action at law).

- 4.3 Effect of Agreement. The execution and delivery of the Transaction Documents and each instrument required hereby to be executed and delivered by each of Buyer and Guarantor and the performance and consummation by each of Buyer and Guarantor, as applicable, of the Transactions will not (a) conflict with or violate the certificate or articles of incorporation or formation, as applicable, and the bylaws of Buyer or Guarantor or any Law to which Buyer or Guarantor is subject, or any judgment, award or decree applicable to Buyer or Guarantor, or (b) conflict with, result in a breach of, constitute (with or without due notice or lapse of time or both) a default under, result in the acceleration of, create in any party the right to accelerate, terminate, modify or cancel, or require any notice, consent or waiver under, or result in any loss of any benefit to which Buyer or Guarantor is entitled under any Guarantor Material Contract.
- 4.4 <u>Approvals</u>. Except for filings with the Securities and Exchange Commission, no Approval or Order or action of or filing with any Governmental Entity is required to be obtained or made by Buyer or Guarantor in connection with the execution and delivery by each of Buyer and Guarantor of the Transaction Documents to which it is a party or the consummation of the Transactions, including, without limitation, payment of the Purchase Price and the Earnout Amount.
- 4.5 <u>Sufficient Funds</u>. Buyer and Guarantor have, or will have at the time for payment thereof pursuant to the terms of this Agreement, sufficient available funds to pay the Purchase Price and the Earnout Amount.

ARTICLE 5 ADDITIONAL COVENANTS

- 5.1 <u>Confidentiality</u>. All information disclosed (i) by Buyer, its Affiliates or representatives to Seller or the Partners (or their respective representatives) or (ii) by Seller or the Partners to Buyer or its Affiliates (or their respective representatives) whether before or after the date hereof, in connection with the Transactions or the discussions and negotiations preceding this Agreement shall be subject to the terms of the Mutual Non-Disclosure Agreement between Seller and Guarantor dated July 23, 2007 (the "<u>Mutual NDA</u>"), which Mutual NDA shall survive with respect to Seller as set forth in Section 10(b) thereof. Further, Seller and the Partners, and any of such Persons' representatives or Affiliates shall not, at any time, make use of, divulge or otherwise disclose, directly or indirectly, any trade secret or other proprietary data (including, without limitation, any Customer list, record or financial information) concerning the Assets or the Business. In addition, Seller and the Partners, and any of such Persons' representatives or Affiliates shall not make use of, divulge or otherwise disclose, directly or indirectly, to Persons other than Buyer and its representatives, any confidential information concerning the Assets or the Business.
- 5.2 <u>Audit Cooperation</u>. After the Closing, Seller and the Partners shall, and shall cause their representatives to, cooperate fully in the preparation of all financial statements determined by Buyer to be necessary to meet its financial reporting and tax obligations in connection with the consummation of the Transactions. Seller and the Partners shall, and shall cause their representatives to, provide, or cause to be provided to Buyer any records and other information requested by Buyer in connection therewith and available to or accessible by such person. None of Seller or any Partner shall be required to incur any out-of-pocket costs (including fees and expenses of attorneys, accountants or other professionals) in taking any action specifically requested by Buyer to be taken by them pursuant to this Section 5.2.

5.3 <u>All Reasonable Efforts</u>. If at any time after the Closing any further action is necessary or desirable to carry out the purposes of the Transaction Documents, including, without limitation, the execution of additional documents or instruments, the parties to the Transaction Documents shall take all such necessary action.

5.4 Post-Closing Cooperation Relating to Assets and Excluded Assets.

- (a) Seller and the Partners agree that they will cooperate with Buyer, at the sole expense of Buyer, in enforcing the terms of any agreements between Seller and any third party involving the activities associated with the Assets, including without limitation terms relating to confidentiality and the protection of any and all rights related to the Intellectual Property ("Intellectual Property Rights"). In the event that Buyer is unable to enforce its Intellectual Property Rights against a third party as a result of a rule or law barring enforcement of such rights by a transferee of such rights, Seller and the Partners agree to reasonably cooperate with Buyer, at the sole expense of Buyer, by assigning to Buyer such rights as may be required by Buyer to enforce its Intellectual Property Rights in its own name.
- (b) Seller and the Partners agree further that, if reasonably requested by Buyer, Seller and the Partners shall cooperate fully with Buyer to provide reasonable access to records and personnel of Seller and the Partners to the extent Buyer finds such access necessary in order to transition the Assets (including the Intellectual Property) to Buyer.
- (c) Notwithstanding anything to the contrary herein, Seller and the Partners agree that they will cooperate fully with Buyer (without the obligation to expend any funds, except as expressly set forth in the penultimate sentence of this subsection) in any Action to challenge, cancel or oppose any third party's use of or trademark registration or application for any Seller Mark or any Trademark confusingly similar thereto, including by assigning Buyer or an Affiliate of Buyer any rights as may be necessary or advisable for Buyer to enforce Seller's rights in any Seller Mark in its own name and providing full access to the records and personnel of Seller and the Partners to the extent Buyer reasonably determines is necessary or advisable to support any Action by Buyer or an Affiliate of Buyer to challenge, cancel or oppose any use of or trademark registration or application for any Seller Mark or any Trademark confusingly similar thereto by any third party and to secure statutory rights and registration for Buyer or an Affiliate of Buyer to any Seller Mark. Seller and the Partners agree that they will reimburse Buyer or, upon Buyer's request, an Affiliate of Buyer, for fifty percent (50%) of the amount of any fees and expenses (including, without limitation, attorneys' fees) incurred by Buyer and/or an Affiliate of Buyer in connection with any Action initiated by Buyer and/or any Affiliate of Buyer to challenge, cancel or oppose any use of, or trademark registration or application for the Disputed Mark; provided, that the aggregate amount paid by Seller and the Partners pursuant to this Section 5.4(c) shall not exceed \$100,000. Nothing in this Section 5.4(c) shall limit the rights of Buyer pursuant to Article 9 hereof with respect to any action initiated by any Person other than Buyer or any Affiliate of Buyer with respect to any Seller Mark or any Trademark confusingly similar thereto.

- (d) Seller and the Partners agree that Seller and the Partners (i) shall be solely responsible for any inquiries, complaints, requests and the like from all former and current Customers and other third parties relating to Seller's services which arise with respect to facts and circumstances arising at any time at or prior to the Closing Date, (ii) shall provide to Buyer at or before the Closing appropriate contact information including, without limitation, a phone number, fax number, email address and physical address for each Customer, (iii) consent to references to Seller's name and contact information on websites controlled by Buyer and its Affiliates, including the websites acquired by Buyer from Seller and (iv) consent to Buyer referring inquiries, complaints, requests and the like from Customers to Seller. In the event that the provided contact information for Seller or any Partner shall be changed, Seller or the Partner shall inform Buyer of such change in writing at least ten (10) days prior to such change.
 - (e) Seller at its expense shall take all actions necessary to ensure Buyer has full right and title to and control over each A/R Bank Account.
- 5.5 No Post-Closing Retention of Copies. Immediately after the Closing, Seller shall deliver to Buyer or destroy copies of Assets in Seller's possession that are in addition to copies delivered to Buyer as part of the Closing, whether such copies are in paper form, on computer media or stored in another form; provided, however, that Seller may retain and use copies of financial books and records relating to the activities associated with the Assets.
- 5.6 <u>Post-Closing Tax Matters</u>. Seller acknowledges that it (and, as applicable, the Partners acknowledge that they) shall be solely responsible for the resolution of any Tax audits or other proceedings relating to Seller or the Business and arising after the Closing Date with respect to any period (or portion thereof) ending on or prior to the Closing Date. However, to the extent Buyer could be reasonably expected to have successor liability for any Taxes of Seller or any Partner, Buyer reserves the right to participate in Tax audits with respect to such Taxes. Nothing in this Section 5.6 shall limit Buyer's indemnification rights under Article 9 of this Agreement or the rights of any Buyer Indemnified Party under Section 9.5. The parties agree to cooperate reasonably in providing information and documentation relating to Tax matters relating to the Business and the Assets, including by providing reasonable access to Tax Returns and supporting workpapers to the extent necessary for the preparation of Tax Returns relating to the Business and the Assets following the Closing.
- 5.7 <u>Continued Existence of Seller</u>. From the date hereof until the end of the Claims Period, Seller and the Partners shall not terminate or otherwise dissolve the legal existence of Seller.

5.8 <u>Use of Purchase Price; Employee Matters</u>.

(a) Subject to Section 5.8(b) below, Seller hereby agrees to pay, or cause to be paid, from the Purchase Price, prior to the distribution or dividend of any portion of the Purchase Price to any of the Partners (or any other security holder of Seller), all of its liabilities and obligations outstanding as of the Closing Date which become due and payable at or prior to the Closing Date, related to any employment, compensation or other related taxes, wages and other compensation and benefits required to be paid to all current and former Seller Employees by applicable Law, including, without limitation, any required tax withholding amounts related thereto,

any 401(k) Matching Payment, any amounts payable or other liabilities due as a result of the assignment of any Contracts required hereunder, all credit card debt, and the Transaction Expenses (except to the extent that the Transaction Expenses are otherwise deducted from the Purchase Price pursuant to the terms hereof). Subject to Section 5.8(b) below, Buyer shall not be obligated to provide any severance, separation pay benefits or any other compensation to any Seller Employee or service provider of Seller on account of any termination of such Person's employment or service relationship with Seller on or before or after the Closing Date, and such amounts and/or benefits (if any) shall be payable by Seller. At or immediately after the Closing, except for the amounts set forth on Schedule 5.8(b) hereto, Seller shall pay or cause to be paid to each Seller Employee, or other service provider of Seller, all sums that are due as of the Closing Date for any accrued but unpaid salary, wages, 401(k) Matching Payment, and other compensation or benefits (if any) as required by applicable Law. Additionally, after the Closing, Seller shall contribute or cause to be contributed all 401(k) Matching Payments to Seller's 401(k) plan, with such payments allocated to the accounts of all applicable Seller Employees, in each case in accordance with the terms of Seller's 401(k) plan and Seller's past practice.

- (b) At the Closing, Buyer shall assume Seller's liabilities and obligations for the Healthcare Employee Payroll Amount and all Undetermined Commissions set forth on Schedule 5.8(b) hereto, and shall pay the Healthcare Employee Payroll Amount in accordance with Seller's regular payroll cycle for Healthcare Employees and the Undetermined Commissions pursuant to the terms of the applicable Seller Benefit Arrangements and the terms hereof. Schedule 5.8(b) hereto sets forth (i) the name of each Healthcare Employee to whom any Healthcare Employee Payroll Amount is payable, and (ii) the name of each Seller Employee who may be entitled to an Undetermined Commission and the maximum potential dollar amount of the Undetermined Commission which may become payable to each such Seller Employee. Notwithstanding the foregoing, in no event does Buyer assume liability for, and Buyer shall have no obligation to pay, (A) any Undetermined Commission amount to any Seller Employee in excess of the maximum potential dollar amount set forth on Schedule 5.8(b) hereto with respect to such Seller Employee, or (B) any salary or wages to any Seller Employee that accrued before the Closing Date other than the Healthcare Employee Payroll Amount. For purposes of clarification, except pursuant to this Section 5.8(b) and as set forth on Schedule 1.3 hereto, Buyer shall only be responsible for and shall only pay liabilities related to events or circumstances involving the Continuing Employees that first occur or arise on or after the Closing Date. In the event Buyer or its Affiliate is the sponsor of a 401(k) plan, Buyer shall arrange for its own 401(k) plan to accept rollovers of distributions from Seller's 401(k) plan with respect to any Continuing Employee who is or becomes eligible to participate in Buyer's 401(k) plan.
- (c) Seller shall not enforce against Buyer, Guarantor or any Continuing Employee any confidentiality, non-compete, non-solicit or similar contractual obligations, or otherwise assert with respect to any such Continuing Employee or Buyer or Guarantor or any of their respective affiliates claims that would otherwise prohibit or place conditions on any such Continuing Employee's acceptance of employment with Buyer or any of its affiliates, any such Continuing Employee's employment by Buyer or any of its affiliates, or any actions taken by any such Continuing Employee as an employee of Buyer or any of its affiliates
- (d) From and after the Closing Date, the Buyer shall recognize the prior service with Seller of each Continuing Employee in connection with any paid time off policy in which any Continuing Employee is eligible to participate following the Closing Date and each such Continuing Employee shall be credited with any accrued paid time off or debited any used paid time off for purposes of their service with Buyer following the Closing that has been accrued or used during their prior service with Seller prior to the Closing.

- (e) To the extent "continuation coverage" (within the meaning of Code Section 4980B) is required to be provided, Buyer will assume responsibility for any "M&A qualified beneficiaries" (within the meaning of Code Section 4980B) in connection with the Transactions.
- 5.9 No Continuing Business Operations; Noncompetition. For a period commencing from the Closing Date and ending on the five year anniversary thereof, neither Seller nor any Partner shall, anywhere within the Territory, directly or indirectly own, manage, operate, control, enable (whether by license, sublicense, assignment or otherwise) or otherwise be affiliated in any manner with or engage or participate in (which shall include, without limitation, marketing of any of Seller's current services) any business or activity whose services or activities compete in whole or in part with the Business, unless otherwise approved in writing by Buyer in its sole discretion. Seller and each Partner hereby acknowledges that a violation of this section shall cause such damage to Buyer as will be irreparable and the exact amount of which will be impossible to ascertain, and for that reason agree that Buyer shall be entitled, as a matter of right, to an injunction from any court of competent jurisdiction restraining any further violation hereunder by Seller or any such Partner. Nothing herein contained shall be construed as prohibiting Buyer from pursuing any other remedies available to it for such breach or threatened breach, including, without limitation, the recovery of damages from Seller or any Partner who is in breach or threatening breach.
- 5.10 <u>Continuing Employees Confidentiality, Non-Competition and Non-Solicitation Agreements</u>. Seller shall use commercially reasonable efforts to obtain from each Continuing Employee an executed Confidentiality, Non-Competition and Non-Solicitation Agreement and to coordinate delivery of all such executed agreements to Buyer at the Closing.
- 5.11 Release of Guarantees. Schedule 5.11 hereto identifies each guarantee executed by any Partner with respect to the obligations of Seller and any Seller Employee related to the Business or the Assets (the "Partner Guarantees"). Seller and the Partners have delivered to Buyer true, correct and complete copies of each Partner Guarantee. Seller and the Partners hereby represent and warrant that any and all Debt incurred on or before the Closing Date related to each Partner Guarantee has been paid by Seller on or before the Closing Date or is included in Section 3.10(d) of the Disclosure Schedule. Subject to termination or cancellation and closure, as applicable, of any line of credit or credit card account with respect to which Seller or any Partner has a guarantee obligation and the accuracy of Sellers and the Partners' representations and warranties in the immediately preceding sentence, Buyer agrees to use commercially reasonable efforts to assist Seller and the Partners in securing releases of each Partner Guarantee following the Closing.
- 5.12 Nonassignable Contracts. With respect to any Contract of Seller that is part of the Assets and that is not capable of being delegated, assigned or transferred to Buyer without the consent or waiver of another party to such Contract, or under which any delegation, assignment or purported assignment to Buyer would constitute a breach or default, Seller shall obtain such requisite

consents or waivers and, to the extent such consents and approvals are not obtained and in full force and effect as of the Closing Date, Seller shall (a) cooperate with Buyer to establish lawful arrangements which result in the transfer to Buyer of all benefits and obligations under such Contracts from and after the Closing Date in a manner that is consistent with the purpose and intention of this Agreement and (b) at Seller's expense, obtain any such consents and waivers, and enforce Seller's rights under such Contracts on behalf of and for the benefit of, and as directed by, Buyer. Schedule 5.12 hereto sets forth a list of those Contracts of Seller that are identified in Section 3.5 of the Disclosure Schedule and for which Seller has obtained and delivered to Buyer all requisite consents and waivers prior to the Closing.

5.13 <u>Professional Liability Insurance</u>. Seller shall not terminate the PL Insurance Tail (or any underlying policy) at any time prior to the day after the third anniversary of the Closing Date and shall use its best efforts to maintain the PL Insurance Tail during such three (3)-year period.

ARTICLE 6 THE CLOSING – SELLER AND PARTNER DELIVERABLES AND ACTIONS

At the Closing, Seller and the Partners are delivering to Buyer each of the following documents and taking each of the following actions:

6.1 <u>Performance</u>. Seller shall have delivered to Buyer a certificate executed by each Limited Partner, dated the Closing Date, certifying that Seller and the Partners have performed and complied with all of their agreements, covenants and conditions required by this Agreement to be performed or complied with by it and them, respectively, prior to or at the Closing Date.

6.2 Employees.

- (a) The Employment Agreement (and all exhibits thereto) shall have been executed by Landry Seedig and delivered to Buyer; and
- (b) Not less than ninety percent (90%) of the Continuing Employees who are Corporate Employees shall have executed and delivered to Buyer a Confidentiality, Non-Competition and Non-Solicitation Agreement.
- 6.3 <u>Bank of Texas Pay-Off.</u> Seller shall have delivered to Buyer a copy of a letter executed by the Bank of Texas, N.A. ("<u>Bank of Texas</u>") setting forth the amount of principal, interest and other amounts necessary to repay in full any and all amounts outstanding under the Business Loan Agreement (Asset Based) dated July 15, 2005 between Seller and Bank of Texas on the Closing Date (the "<u>Pay-Off Letter</u>"), and, contingent upon payment by or on behalf of Seller to Bank of Texas of the aggregate amount set forth in the Pay-Off Letter, shall include an agreement by Bank of Texas to file, or an authorization for Seller to file, a UCC termination statement with the Secretary of State of Texas releasing Bank of Texas' security interest in any and all of the Assets promptly after receipt of the amounts set forth in the Pay-Off Letter.
 - 6.4 <u>Liabilities Schedule</u>. Seller shall have delivered <u>Schedule 2.2(b)</u> to Buyer setting forth the Liability Payments as of the Closing Date.

- 6.5 <u>Seller Closing Debt</u>. As of the Closing Date, except for the amounts set forth in the Pay-Off Letter, the Seller Closing Debt listed in <u>Schedule 2.2(b)</u> hereto and the Assumed Working Capital Obligations, Seller shall not have any Debt.
- 6.6 <u>Partner Independent Consultant Agreements</u>. On or before the Closing, each of the Limited Partners shall have executed and delivered to Buyer an Independent Consultant Agreement, each in the form attached hereto as <u>Exhibit B</u> (each, a "<u>Partner Independent Consultant Agreement</u>").
- 6.7 Other Independent Consultant Agreements. On or before the Closing, Seller shall have delivered to Buyer copies of the Independent Consultant Agreement entered into between Seller and each of Barrett Alley, Wil Batchler, Carlos Mendez and Verve Communications, each in the form attached hereto as Exhibit C.
- 6.8 <u>Insurance; Insurance Tail</u>. As of the Closing, Seller shall have obtained, and delivered to Buyer proof of, a fully paid-up three (3) year extended reporting period professional liability insurance policy providing for professional liability insurance from an insurance carrier with not less than an "A" rating from A.M. Best with respect to claims arising from facts or events that occurred on or before the Closing that (a) provides for not more than a \$25,000 self-insured or deductible amount pre claim and (b) is reasonably acceptable to Buyer (the "<u>PL Insurance Tail</u>").
- 6.9 <u>Transfer of A/R Bank Accounts</u>. On or before the Closing, Seller shall have taken all actions necessary to assign and transfer to Buyer the A/R Bank Accounts (including, without limitation, obtaining any signature guarantees required by any financial institution that maintains an A/R Bank Account).
 - 6.10 <u>Seller Contracts Workbook</u>. On or before the Closing, Seller shall have delivered or caused to be delivered to Buyer the Seller Contracts Workbook.
 - 6.11 Additional Closing Deliverables and Actions. Seller shall execute and deliver, or cause to be executed and delivered to Buyer:
 - (a) the Bill of Sale and Assignment of Contracts;
 - (b) the Assignment and Assumption of Office Lease;
 - (c) the Assignment and Assumption of Office Sublease;
 - (d) the Assignment of Intellectual Property;
 - (e) the Assignment and Assumption of Temporary Storage Agreement;
- (f) a certificate of Aunkst and Bomar, dated the Closing Date, certifying (i) Seller's certificate of limited partnership and the Limited Partnership Agreement and (ii) resolutions of the General Partner and the Limited Partners approving and adopting this Agreement and the Transactions;

- (g) a certificate of the manager or managers of the General Partner, dated the Closing Date, certifying (i) the General Partner's articles of organization and regulations and (ii) resolutions of the General Partner approving and adopting this Agreement and the Transactions;
- (h) certificates, or other evidence reasonably acceptable to Buyer, from each of the Secretary of State and the Comptroller of Public Accounts of the State of Texas and the Secretary of State (or other applicable Governmental Entity) of each other state or jurisdiction in which Seller is qualified to do business certifying that Seller is in good standing and that all applicable Taxes and fees of Seller incurred through and including the Closing Date have been paid;
 - (i) an opinion dated the Closing Date of Baker & McKenzie LLP, counsel to Seller, in the form attached hereto as Exhibit D;
- (j) (i) copies of UCC-2 or UCC-3 termination statements duly executed by each Person (other than Bank of Texas) holding a security interest in any Asset and (ii) evidence reasonably satisfactory to Buyer that all Encumbrances of record or that are otherwise known to Seller to which any of the Assets are subject or by which any of the Assets are bound have been released prior to, or shall be released simultaneously with, the Closing;
 - (k) the Seller Financial Condition Certificate;
 - (l) the books, records and other documentation of Seller relating to the Assets; and
- (m) a Consent of Spouse, in the form attached hereto as <u>Exhibit E</u>, executed and delivered to Buyer by the spouse of each Partner, as applicable. <u>Schedule 6.11(m)</u> hereto sets forth the name of the spouse of each Partner or, if a Partner is not married, indication of that fact.

ARTICLE 7 THE CLOSING – BUYER DELIVERABLES AND ACTIONS

At the Closing, Buyer is delivering to Seller each of the following documents and taking each of the following actions:

7.1 <u>Performance</u>. Buyer shall have delivered to Seller a certificate of an officer of Buyer, dated the Closing Date, that Buyer has performed and complied with all of its agreements, covenants and conditions required by this Agreement to be performed or complied with by it prior to or at the Closing Date.

7.2 Additional Closing Deliverables and Actions.

(a) On the Closing Date, Buyer shall have delivered to Seller the Initial Cash Payment via wire transfer to an account of Seller, which account, along with wire transfer instructions thereto, shall be specified by Seller in writing at least three (3) Business Days before the Closing;

- (b) On the Closing Date, Buyer shall have paid the Liability Payments in the amounts and to the Persons identified on Schedule 2.2(b) hereto.
- (c) Buyer shall have executed and delivered each Partner Independent Consultant Agreement;
- (d) Buyer shall have executed and delivered the Assignment and Assumption of Office Lease;
- (e) Buyer shall have executed and delivered the Assignment and Assumption of Office Sublease; and
- (f) Buyer and Guarantor shall have delivered a certificate of an officer of each of Buyer and Guarantor, dated the Closing Date, certifying (i) each of Buyer's and Guarantor's certificate of formation and bylaws and (ii) resolutions of the boards of directors of each of Buyer and Guarantor approving and adopting this Agreement and the Transactions.

ARTICLE 8 SURVIVAL

8.1 <u>Survival of Seller's and the Partners' Representations and Warranties</u>. The representations and warranties of Seller and the Partners contained in this Agreement, the Disclosure Schedule, the other Transaction Documents and the certificates of Seller delivered pursuant to Article 6 shall survive the Closing and remain in full force and effect, regardless of any investigation made by or on behalf of any of the parties to this Agreement, until the close of business on the eighteen (18) month anniversary of the Closing Date ("<u>Expiration Date</u>"); <u>provided, however</u>, that the representations and warranties of Seller contained in each of Section 3.1 (Organization, Power and Authority), Section 3.3 (Authorization and Approval of Agreements), Section 3.4 (Effect of Agreement), Section 3.6 (Capitalization), Section 3.9 (Tax Matters) and Section 3.17 (Employees and Other Compliance) shall remain operative and in full force and effect until all applicable statutes of limitations shall have expired, in each case, regardless of any investigation made by or on behalf of any of the parties to this Agreement; <u>provided further</u>, however, that no right to indemnification pursuant to Article 9 in respect of any breach of a representation or warranty that is set forth in a Notice of Claim delivered prior to the applicable expiration date of such representation or warranty shall be affected by the expiration of such representation or warranty; <u>provided, however</u>, that the right of indemnification shall be unaffected only with respect to that portion of the applicable representation or warranty that is subject of the Notice of Claim; and <u>provided, further</u>, that such expiration shall not affect the rights of any Buyer Indemnified Party (as defined below) under Article 9 or otherwise to seek recovery of Losses arising out of any fraud, willful breach or intentional misrepresentation by Seller or the Partners until the expiration of the applicable statute of limitations with respect thereto.

8.2 <u>Survival of Buyer's and Guarantor's Representations and Warranties</u>. The representations and warranties of Buyer and Guarantor contained in this Agreement shall survive the Closing and remain in full force and effect until the close of business on the Expiration Date; <u>provided</u>, <u>however</u> that the representations and warranties of Buyer and Guarantor contained in Section 4.5

(Sufficient Funds) of this Agreement shall survive the Closing and remain in full force and effect until the applicable statutes of limitations shall have expired; provided further, however, that no right to indemnification pursuant to Article 9 in respect of any breach of a representation or warranty of Buyer or Guarantor contained in this Agreement that is set forth in a Notice of Claim delivered prior to the applicable expiration date of such representation or warranty shall be affected by the expiration of such representation or warranty; provided, however, that the right of indemnification shall be unaffected only with respect to that portion of the applicable representation or warranty that is subject of the Notice of Claim; and provided, further, that such expiration shall not affect the rights of any Seller Indemnified Party (as defined below) under Article 9 or otherwise to seek recovery of Losses arising out of any fraud, willful breach or intentional misrepresentation by Buyer or Guarantor until the expiration of the applicable statute of limitations with respect thereto.

8.3 <u>Survival of Post-Closing Covenants</u>. All covenants and agreements to be performed by any party pursuant to the terms of this Agreement after the Closing Date shall continue indefinitely to the extent necessary to give effect to their respective terms.

ARTICLE 9 INDEMNIFICATION; HOLDBACK

- 9.1 <u>Obligations of Seller and the Partners</u>. Subject to the limitations set forth herein, Seller and each Partner (individually, a "<u>Seller Indemnifying Party</u>" and, together, the "<u>Seller Indemnifying Parties</u>") agrees to indemnify and hold harmless Buyer and its directors, officers, employees, Affiliates, agents, advisors, representatives, stockholders and assigns (each hereinafter referred to individually as a "<u>Buyer Indemnified Party</u>," and, collectively, as the "<u>Buyer Indemnified Parties</u>") from and against any and all Losses directly or indirectly arising out of, resulting from or in connection with:
- (a) any inaccuracy in or breach of any representation or warranty of Seller or the Partners made in this Agreement, including the Disclosure Schedule and the other exhibits and schedules hereto, and in any other Transaction Document delivered hereunder, as of the Closing Date (except in the case of any individual representation and warranty which by its terms speaks only as of a specific date or dates), disregarding for purposes of this subsection the Disregarded Disclosures;
- (b) any failure of any certification made by Seller or any Partner pursuant to Section 2.3 or Article 6 of this Agreement to be true and correct as of the Closing Date or, if different, as of the date such certificate is delivered to Buyer;
 - (c) any breach of or failure of performance of any covenant of Seller made in this Agreement;
 - (d) any assertion of claims made against the Assets sold hereunder or against Buyer by creditors of Seller under any bulk sales Law;
 - (e) the Excluded Liabilities (including, without limitation, any liability or obligation in respect of Transaction Expenses);

- (f) the Excluded Assets;
- (g) any professional liability incurred, whether known or unknown as of the Closing Date, with respect to any employee or service provider placed by Seller at any time at or prior to the Closing Date which liability first arose at or prior to the Closing Date and which relates to any specialty or discipline for which Seller has placed any such employee or service provider at any time at or prior to the Closing Date; and
 - (h) any Net Working Capital Decrease and amount deemed to be Losses pursuant to Sections 2.3 and 2.4(b)(ii).
- 9.2 Obligations of Buyer. Subject to the limitations set forth herein, Buyer (for purposes of this Article 9, the "Buyer Indemnifying Party," and, together with the Seller Indemnifying Parties, the "Indemnifying Parties") agrees to indemnify and hold harmless Seller, the Partners and their respective directors, officers, employees, Affiliates, agents, advisors, representatives, members, managers and assigns (each hereinafter referred to individually as a "Seller Indemnified Party," and, collectively, as the "Seller Indemnified Parties," and, collectively with the Buyer Indemnified Parties, the "Indemnified Parties") from and against any and all Losses directly or indirectly arising out of, resulting from or in connection with (i) any inaccuracy in or breach of the representations and warranties of Buyer or Guarantor made in Article 4 of this Agreement, (ii) any failure of any certification made by Buyer or Guarantor pursuant to Article 7 of this Agreement to be true and correct as of the Closing Date or, if different, as of the date such certificate is delivered to Seller, (iii) any breach or failure of performance of any covenant of Buyer or Guarantor made in this Agreement, (iv) the Assumed Liabilities (including, without limitation, any liability or obligation with respect to Liability Payments to those persons and in those amounts as are set forth on Schedule 2.2(b) hereto), and (v) any failure of Seller to (A) terminate in writing the employment of any Continuing Employee's employment with Seller, in each case at the request of Buyer.

9.3 Limitations.

(a) Notwithstanding anything contained herein to the contrary, no Buyer Indemnified Party may receive any portion of the Holdback Amount in respect of any claim for indemnification that is made pursuant to Section 9.1(a) or (b) above (other than claims which result from any inaccuracy in or breach of any representation or warranty made by Seller or any Partner in Section 3.1 (Organization, Power and Authority), Section 3.3 (Authorization and Approval of Agreements), Section 3.4 (Effect of Agreement), Section 3.6 (Capitalization), Section 3.7 (Intellectual Property), Section 3.9 (Tax Matters) and Section 3.17 (Employees and Other Compliance), or that involve fraud, willful breach or intentional misrepresentation by Seller or any Partner), unless and until Losses in an aggregate amount greater than \$25,000 (the "Basket") have been incurred, paid or accrued, in which case the Buyer Indemnified Party may make claims for indemnification and may receive cash from the Holdback Amount for all indemnifiable Losses (including the amount of the Basket).

- (b) The aggregate amount recoverable by all Buyer Indemnified Parties under this Article 9 from the Seller Indemnifying Parties, including, without limitation, in connection with any representation, warranty, covenant or agreement herein shall not (when aggregated with all other amounts recoverable by such parties hereunder) exceed the Holdback Amount, except for Losses arising out of or related to the following (in which cases, when aggregated with all other amounts recoverable by such parties, the amount recoverable shall not exceed the applicable amounts, if any, set forth in (i) - (iv) of this subsection 9.3(b)): (i) fraud, willful breach or intentional misrepresentation of Seller or any Partner, in which case there shall be no limit to the amount recoverable by all Buyer Indemnified Parties against the party or parties that committed the fraud, willful breach and/or intentional misrepresentation, (ii) any inaccuracy in or breach of any representation or warranty set forth in Section 3.7 (Intellectual Property), in which case the amount recoverable by all Buyer Indemnified Parties shall not exceed the amount equal to the Holdback Amount or the remaining portion thereof, if applicable, plus \$1,000,000 (with the entire \$1,000,000 available for Claims related to Section 3.7 (Intellectual Property) and Section 3.17 (Employees and Other Compliance)) (for the avoidance of doubt, the maximum amount recoverable pursuant to the foregoing is limited to the Holdback Amount plus \$1,000,000 in the aggregate), (iii) any inaccuracy in or breach of any representation or warranty set forth in Section 3.17 (Employees and Other Compliance), in which case the amount recoverable by all Buyer Indemnified Parties shall not exceed the amount equal to the Holdback Amount or the remaining portion thereof, if applicable, plus \$1,000,000 (with the entire \$1,000,000 available for Claims related to Section 3.7 (Intellectual Property) and Section 3.17 (Employees and Other Compliance)) (for the avoidance of doubt, the maximum amount recoverable pursuant to the foregoing is limited to the Holdback Amount plus \$1,000,000 in the aggregate), (iv) Sections 9.1(c) through 9.1(h), in each case for which there shall be no limit to the amount recoverable by the Buyer Indemnified Parties, and (v) any inaccuracy in or breach of any representation or warranty set forth in Section 3.1 (Organization, Power and Authority), Section 3.3 (Authorization and Approval of Agreements), Section 3.4 (Effect of Agreement), Section 3.6 (Capitalization) and Section 3.9 (Tax Matters), for which the aggregate amount recoverable by all Buyer Indemnified Parties shall not exceed the amount of the Purchase Price plus the Earnout Amount payable to Seller.
- (c) Notwithstanding anything contained herein to the contrary, the indemnification obligations of the Partners hereunder shall be several and proportionate (based on each Partner's Pro Rata Interest), and not joint, except in the case of any Loss arising out of or related to fraud, willful breach or intentional misrepresentation of any Partner, in which case (i) the liability of the Partner who has committed such fraud, willful breach or intentional misrepresentation shall be unlimited and (ii) the liability of each Partner who did not commit the fraud, willful breach or intentional misrepresentation shall not exceed the amount equal to such Partner's Pro Rata Interest in the amount of the Purchase Price plus the Earnout Amount payable to Seller. With respect to Losses arising out of or related to fraud, willful breach and/or intentional misrepresentation of any Partner, the Buyer Indemnified Parties shall first seek indemnification for such Losses against the Partner or Partners who committed the fraud, willful breach and/or intentional misrepresentation and, thereafter, shall seek indemnification for such Losses against any other Partner and/or Seller.
- (d) Notwithstanding anything contained herein to the contrary, the aggregate amount recoverable by all Seller Indemnified Parties under this Article 9 from Buyer in connection with any representation, warranty, covenant or agreement herein shall not exceed the Holdback Amount plus the Earnout Amount payable to Seller.

9.4 Notice of Claim.

- (a) As used herein, the term "Claim" means a claim for indemnification under this Article 9 by any Indemnified Party for Losses which have been or may be incurred or accrued by such Indemnified Party.
- (b) An Indemnified Party may give notice of a Claim under this Agreement by written notice executed by an officer of the Indemnified Party (if the Indemnified Party is not a natural person) or by the Indemnified Party or its authorized representative (if the Indemnified Party is a natural person) (a "Notice of Claim"), to the Indemnifying Party promptly after the Indemnified Party becomes aware of the existence of any potential claim for indemnification under this Article 9, arising from or relating to:
 - (i) any matter specified in Section 9.1 or Section 9.2; or
- (ii) the assertion, whether orally or in writing, against the Indemnified Party of a claim, demand, suit, action, arbitration, investigation, inquiry or proceeding brought by a third party against the Indemnified Party (in each such case, a "<u>Third-Party Claim</u>") that is based on, arises out of or relates to any matter specified in Section 9.1 or Section 9.2.
- (c) A delay on the part of the Indemnified Party in giving the Indemnifying Party a Notice of Claim shall relieve the Indemnifying Party from any of its obligations under this Article 9 but only to the extent that the Indemnifying Party is materially prejudiced thereby.
- (d) All Notices of Claims and other copies or correspondence required to be delivered to any Seller Indemnifying Party or Seller Indemnified Party shall be delivered to Seller and all actions to be taken by any Seller Indemnifying Party or Seller Indemnified Party shall be taken by Seller on behalf of any such Seller Indemnifying Party or Seller Indemnified Party.
- (e) The period during which claims may be initiated (the "Claims Period") for indemnification shall commence at the Closing and terminate on the Expiration Date; provided, however, that (i) the Claims Period for indemnification from and against Losses arising out of, resulting from or in connection with (A) fraud, willful breach or intentional misrepresentation of Seller or any Partner, (B) Losses arising out of, resulting from or in connection with Sections 9.1(c) through 9.1(f), and (C) any inaccuracy in or breach of any representation or warranty set forth in Section 3.1 (Organization, Power and Authority), Section 3.3 (Authorization and Approval of Agreements), Section 3.4 (Effect of Agreement), Section 3.6 (Capitalization), Section 3.9 (Tax Matters) and Section 3.17 (Employees and Other Compliance) shall commence on the Closing Date and terminate upon the expiration of the applicable statute of limitations, and (ii) the Claims Period for indemnification from and against Losses arising out of or relating to (A) fraud, willful breach or intentional misrepresentation of Buyer or Guarantor, (B) Losses arising out of, resulting from or in connection with Sections 9.2(iii) and 9.2(iv), and (C) any inaccuracy in or breach of any representation or warranty set forth in Section 4.5 (Sufficient Funds) shall commence on the Closing Date and terminate upon the expiration of the applicable statute of limitations.

Notwithstanding anything contained herein to the contrary, any Claims for Losses specified in any Notice of Claim delivered to an Indemnifying Party prior to expiration of the applicable Claims Period with respect to facts and circumstances existing prior to expiration of the applicable Claims Period shall remain outstanding until such Claims for Losses have been resolved or satisfied, notwithstanding the expiration of such Claims Period. For purposes of clarification, with respect to the foregoing sentence, only the items identified in such Notice(s) of Claim shall survive the expiration of the Claims Period.

9.5 Defense of Third-Party Claims.

- (a) The Indemnified Party shall determine and conduct the defense or settlement (subject to Section 9.5(b)) of any Third-Party Claim, and the costs and expenses incurred by the Indemnified Party in connection with such defense or settlement (including reasonable attorneys' fees, other professionals' and experts' fees and court or arbitration costs) shall be included in the Losses for which the Indemnified Party may seek indemnification pursuant to a Claim made by the Indemnified Party hereunder.
- (b) Settlement of any Third-Party Claim shall be determinative of the existence of or amount of Losses relating to such matter without the consent of the Indemnifying Party unless the Indemnifying Party shall have objected within ten (10) days after a written request for such consent by the Indemnified Party.
- (c) Notwithstanding anything herein to the contrary, the Indemnifying Party shall have the right to monitor the defense and settlement of any Third-Party Claim and to receive copies of pleadings and other documents related to any Third Party Claim as may be reasonably requested by the Indemnifying Party from time to time, but only to the extent that such monitoring activities, including the receipt of any documents, does not affect any privilege relating to the Indemnified Party or result in the violation or breach by any Indemnified Party of any confidentiality agreement between such Indemnified Party and any other Person. The Indemnifying Party shall bear the entire cost (including the fees and expenses of any attorneys, experts or other service providers) associated with exercising the rights provided in this subsection. For purposes of clarification, this subsection does not give any Indemnifying Party any right to determine or conduct the defense or of any Third-Party Claim or, subject to Section 9.5(b), settlement negotiations with respect to any Third-Party Claim.
 - 9.6 Contents of Notice of Claim. Each Notice of Claim by the Indemnified Party given pursuant to Section 9.4 shall contain the following information:
- (a) that the Indemnified Party has directly or indirectly incurred, paid or properly accrued or, in good faith, reasonably believes it shall have to directly or indirectly incur, pay or accrue, Losses in an aggregate stated amount arising from such Claim (which amount may be an estimated amount and may be the amount of Losses claimed by a third party in an Action brought against the Indemnified Party based on alleged facts, which if true, would give rise to liability for Losses to such Indemnified Party under this Article 9);
- (b) a brief description, in reasonable detail (to the extent reasonably available to the Indemnified Party), of the facts, circumstances or events giving rise to the alleged Losses based on the Indemnified Party's good faith reasonable belief thereof, including the identity and address of any third-party claimant (to the extent reasonably available to the Indemnified Party) and copies of any formal demand or complaint, the amount of Losses (to the extent known), or the basis for such anticipated liability, and the specific nature of the breach to which such item is related;

(c) a statement that: (i) the Claims identified in such Notice of Claim (together with Claims set forth in all Notice of Claims theretofore delivered), individually or in the aggregate, do not satisfy the Basket; (ii) the Claims identified in such Notice of Claim (together with Claims set forth in all Notice of Claims theretofore delivered), individually or in the aggregate do satisfy the Basket, or (iii) that the Basket does not apply to such Claims.

9.7 Resolution of Notice of Claim. Each Notice of Claim given by an Indemnified Party shall be resolved as follows:

- (a) If, within fifteen (15) days after a Notice of Claim is received by the Indemnifying Party, the Indemnifying Party does not contest such Notice of Claim in writing to the Indemnified Party, the Indemnifying Party shall be conclusively deemed to have consented to the recovery by the Indemnified Party of the full amount of Losses (subject to the limits contained in this Article 9) specified in the Notice of Claim in accordance with this Article 9, and, without further notice, to have stipulated to the entry of a final judgment for damages against the Indemnifying Party for such amount in any court having jurisdiction over the matter where venue is proper.
- (b) Following the delivery of a Notice of Claim to the Indemnifying Party, the Indemnifying Party shall be given such access as it may reasonably require during the Indemnified Party's normal business hours (or such other times as the parties may agree) to those books and records of the Indemnifying Party in the possession of, and/or under the control of, the Indemnified Party, and access to such personnel or representatives of the Indemnified Party as they may reasonably require for the purposes of determining whether to contest all or any portion of a Notice of Claim.
- (c) If the Indemnifying Party gives the Indemnified Party written notice contesting all or any portion of a Notice of Claim (a "Contested Claim") within the fifteen (15) day period specified in Section 9.7(a) above, then such Contested Claim shall be resolved by either (i) a written settlement agreement or memorandum executed by the Indemnified Party and the Indemnifying Party or (ii) in the absence of such a written settlement agreement within thirty (30) days following receipt by the Indemnified Party of the written notice from the Indemnifying Party, by binding arbitration between the Indemnified Party and the Indemnifying Party in accordance with the terms and provisions of Section 9.7(d) below.
- (d) If no such agreement can be reached after good faith negotiation, either Buyer or Seller may, by written notice to the other, demand arbitration of the matter before arbitrators from Judicial Arbitration and Mediation Services ("JAMS"), unless the amount of the damage or loss is at issue in pending litigation with a third party, in which event arbitration shall not be commenced until such amount is ascertained or both parties agree to arbitration; and in either such event the matter shall be settled by arbitration conducted by three (3) arbitrators. Within fifteen (15) days after such written notice is sent, Buyer (on the one hand) and Seller (on the other hand) shall each select one arbitrator from JAMS, and the two (2) arbitrators so selected shall select a third arbitrator from

JAMS. The decision of the arbitrators as to the validity and amount of any claim in any disputed Notice of Claim shall be binding and conclusive upon the parties to this Agreement, and, if applicable, Buyer shall be entitled to act in accordance with such decision and withhold from any Applicable Earnout Amount owed to Seller pursuant to Section 2.4(c), if any, in accordance with such decision.

- (e) Judgment upon any award rendered by the arbitrators may be entered in any court having jurisdiction. Any such arbitration shall be held in Dallas, Texas under the commercial rules then in effect of JAMS. The non-prevailing party to an arbitration shall pay its own expenses, the fees of each arbitrator, the administrative fee of JAMS, and the expenses, including, without limitation, the reasonable attorneys' fees and costs, incurred by the prevailing party to the arbitration. The arbitration panel shall be authorized to determine which party to the arbitration is the prevailing party and which party is the non-prevailing party.
- 9.8 <u>Payment of Indemnification Claims</u>. Any amounts owed to any Buyer Indemnified Party following the resolution of a Claim, as determined by Section 9.7 above, shall be satisfied first from the Holdback Amount and the Earnout Amount, if any. In the event that the aggregate value of such Claims exceeds the Holdback Amount and, following the date that the determination of the Applicable Earnout Amount shall become conclusive and binding on the parties to this Agreement, the Earnout Amount (if any), Seller and the Partners must pay any remaining amounts owed to any Buyer Indemnified Party within fifteen (15) Business Days following the resolution of each Claim. Any amounts owed to any Seller Indemnified Party following the resolution of such Claim.
- 9.9 <u>Release of Remaining Holdback Amount</u>. Within ten (10) Business Days following the close of business on the Expiration Date, Buyer shall deliver to Seller all of the remaining Holdback Amount in excess of any Holdback Amount that is necessary to satisfy all unresolved, unsatisfied or disputed claims for Losses specified in any Notice of Claim delivered to Seller before the Expiration Date. If any such Claims are unresolved, unsatisfied or disputed as of the expiration of the Claims Period, then Buyer shall retain possession and custody of that portion of the Holdback Amount that equals the total maximum amount of Losses then being claimed by Buyer Indemnified Parties in all such unresolved, unsatisfied or disputed Claims, and as soon as all such Claims have been resolved, Buyer shall deliver to Seller all of the remaining Holdback Amount (if any) not required to satisfy such Claims.
- 9.10 <u>Tax Consequences of Indemnification Payments</u>. All payments and offsets (if any) made or received by Buyer pursuant to any indemnification obligations under this Article 9 will be treated as adjustments to the purchase price for tax purposes and such agreed treatment will govern for purposes of this Agreement, unless otherwise required by law.
- 9.11 <u>Appointment of Seller as Attorney-In-Fact</u>. By execution of this Agreement, each Partner shall be deemed to have irrevocably appointed and designated Seller as the representative of the Partners and as the attorney-in-fact and agent for and on behalf of each Partner with respect to claims for indemnification under this Article 9, the taking by Seller of any and all actions set

forth in Section 2.4 and the taking by Seller of any and all actions and the making of any decisions required or permitted to be taken by Seller under this Agreement, including the exercise of the power to: (a) give and receive notices and communications to or from Buyer (on behalf of itself or any other Buyer Indemnified Party) relating to this Agreement or any of the transactions and other matters contemplated hereby (except to the extent that this Agreement expressly contemplates that any such notice or communication shall be given or received by such Partners individually); (b) authorize the withholding by Buyer of all or a portion of the Applicable Earnout Amount in satisfaction of indemnification claims by Buyer or any other Buyer Indemnified Party pursuant to this Article 9 (including by not objecting to such claims); (c) agree to, object to, negotiate, resolve, enter into settlements and compromises of, demand arbitration or litigation of, and comply with orders of arbitrators or courts with respect to, (i) indemnification claims by Buyer or any other Buyer Indemnified Party pursuant to this Article 9 or (ii) any dispute between any Buyer Indemnified Party and any such Partner, in each case relating to this Agreement; (d) receive notices and take such actions as may be necessary or desirable pursuant to Section 2.4 and this Article 9; and (e) take all actions necessary or appropriate in the judgment of Seller for the accomplishment of the foregoing. Seller shall have authority and power to act on behalf of each Partner with respect to the disposition, settlement or other handling of all claims under this Article 9 and all rights or obligations arising under Section 2.4 or this Article 9. The Partners shall be bound by all actions taken and documents executed by Seller in connection with Section 2.4 or this Article 9, and Buyer and other Buyer Indemnified Parties shall be entitled to rely on any action or decision of Seller. No bond shall be required of Seller, and Seller shall receive no compensation fo

9.12 Insurance Proceeds. Notwithstanding anything contained herein to the contrary, the amount of any Claims incurred or suffered by an Indemnified Party shall be calculated after giving effect to (a) any insurance proceeds actually received by the Indemnified Party (or any of its Affiliates) with respect to such Claims, net of any costs or expenses incurred by such Indemnified Party in making the claim for such proceeds, and (b) any recoveries obtained by the Indemnified Party (or any of its Affiliates) from any other third person; provided, however, that no Indemnified Party shall have any obligation to pursue such insurance proceeds or recovery from third persons. If any such proceeds or recoveries are received by an Indemnified Party (or any of its Affiliates) with respect to any Claims after an Indemnifying Party has made a payment to the Indemnified Party with respect to such Claim, the Indemnified Party (or such Affiliate) shall pay to the Indemnifying Party the amount of such proceeds or recoveries (up to the amount of the Indemnifying Party's payment with respect to such Claim). As a courtesy to the Indemnifying Party, if requested by the Indemnifying Party following the delivery of a Notice of Claim, the Indemnified Party will endeavor to indicate to the Indemnifying Party whether the Losses set forth in such Notice of Claim are covered under the Indemnified Party's insurance policies; provided, however, that in no event shall the Indemnified Party have any obligation to pursue recovery under any such insurance policy provided, further, however, that in no event shall the Indemnified Party to provide any such indication as to availability of insurance under any insurance policy or to pursue recovery under any insurance policy affect any rights of the Indemnified Party to indemnification for Losses under this Article 9.

ARTICLE 10 TERMINATION

- 10.1 <u>Termination by Mutual Consent</u>. This Agreement may be terminated at any time prior to the Closing Date by the mutual written consent of Buyer and Seller.
- 10.2 <u>Procedure and Effect of Termination</u>. In the event of termination of this Agreement by Buyer and Seller pursuant to Section 10.1, this Agreement shall forthwith become void and there shall be no liability or obligation on the part of any party hereto or their respective officers, directors, stakeholders or Affiliates; <u>provided, however</u>, that (i) the provisions of Section 5.1 (Confidentiality) (with respect to Seller and the Partners only), this Section 10.2 (Procedure and Effect of Termination) and Article 11 (General) shall remain in full force and effect and survive any termination of this Agreement and (ii) nothing herein shall relieve any party hereto from liability in connection with any material breach of any of such party's representations, warranties or covenants contained herein.

ARTICLE 11 GENERAL

- 11.1 <u>Amendments</u>; <u>Waivers</u>. This Agreement and any Schedule or Exhibit attached hereto may be amended only by agreement in writing of all parties thereto. No waiver of any provision nor consent to any exception to the terms of this Agreement shall be effective unless in writing and signed by the party to be bound and then only to the specific purpose, extent and instance so provided.
- 11.2 <u>Schedules; Exhibits; Integration</u>. Each Schedule and Exhibit delivered pursuant to the terms of this Agreement shall be in writing and shall constitute a part of this Agreement, although such Schedules and Exhibits need not be attached to each copy of this Agreement. This Agreement, together with such Schedules and Exhibits, constitutes the entire agreement among the parties pertaining to the subject matter hereof and supersedes all prior agreements and understandings of the parties in connection therewith.
- 11.3 Governing Law; Submission to Jurisdiction. This Agreement shall be governed by, and construed in accordance with, the internal laws of the State of Delaware without regard to the choice of law principles thereof. Except with respect to the dispute resolution mechanisms set forth in Sections 2.3 and 2.4 and Article 9, each of the parties hereto irrevocably submits to the exclusive jurisdiction of the courts of the State of Delaware located in Newcastle County and the United States District Court for the District of Delaware for the purpose of any suit, action, proceeding or judgment relating to or arising out of this Agreement and the transactions contemplated hereby. Service of process in connection with any such suit, action or proceeding may be served on each party hereto anywhere in the world by the same methods as are specified for the giving of notices under this Agreement. Each of the parties hereto irrevocably consents to the jurisdiction of any such court in any such suit, action or proceeding and to the laying of venue in such court. Each party hereto irrevocably waives any objection to the laying of venue of any such suit, action or proceeding brought in such courts and irrevocably waives any claim that any such suit, action or proceeding brought in an inconvenient forum.

- 11.4 <u>No Assignment</u>. Neither this Agreement nor any rights or obligations under it are assignable except that Buyer may assign without the consent of any other party its rights hereunder to any subsidiary of Buyer or to any post-Closing purchaser of a substantial part of the Assets.
- 11.5 <u>Headings</u>. The descriptive headings of the Articles, Sections and subsections of this Agreement are for convenience only and do not constitute a part of this Agreement.
- 11.6 <u>Counterparts</u>. This Agreement and any amendment hereto or any other agreement (or document) delivered pursuant hereto may be executed in one or more counterparts and by different parties in separate counterparts. All of such counterparts shall constitute one and the same agreement (or other document) and shall become effective (unless otherwise provided therein) when one or more counterparts have been signed by each party and delivered to the other party.
- 11.7 <u>Publicity and Reports</u>. Seller and the Partners shall not release, issue or permit to be made or issued any press release, public statement or other public notice relating to this Agreement, or the transactions contemplated by this Agreement, without obtaining the prior consent of Buyer. Seller and the Partners shall obtain the prior consent of Buyer to the form and content of any application or report made to any Governmental Entity or which relates to this Agreement, which consent shall not be unreasonably withheld.
- 11.8 Remedies. Except in the case of fraud or intentional misrepresentation, the rights and remedies set forth in Article 9 of this Agreement shall be the exclusive remedy of the parties to this Agreement to recover for breach of this Agreement or any other Transaction Document. In the case of fraud or intentional misrepresentation of a party, all rights and remedies existing under this Agreement shall be cumulative to, and not exclusive of, any rights or remedies otherwise available to the injured party and the rights and remedies set forth in Article 9 of this Agreement shall not be deemed to preclude or otherwise limit in any way the exercise of any other rights or pursuit of other remedies.
- 11.9 Parties in Interest. This Agreement shall be binding upon and inure to the benefit of each party, and nothing in this Agreement, express or implied, is intended to confer upon any other Person any rights or remedies of any nature whatsoever under or by reason of this Agreement. Nothing in this Agreement is intended to relieve or discharge the obligation of any third Person to any party to this Agreement.
- 11.10 Notices. All notices and other communications required or permitted between or among the parties under this Agreement shall be in writing and shall be either delivered in person, sent by facsimile, sent by certified or registered first-class mail, postage pre-paid, sent by nationally recognized express courier service, or, if notice or other communication is to Seller, by electronic mail. Such notices, consents and other communications shall be effective upon receipt if delivered in person or sent by facsimile, three (3) Business Days after mailing if sent by mail, and one (1) Business Day after dispatch if sent by express courier, to the following addresses, or such other addresses as any party may notify the other parties in accordance with this Section 11.10:

If to Buyer, addressed to:

c/o AMN Healthcare, Inc. 12400 High Bluff Drive San Diego, CA 92130 Attention: General Counsel

Fax No.: (858) 509-3587

With a copy (which shall not constitute notice) to:

Heller Ehrman LLP 275 Middlefield Road Menlo Park, CA 94025 Attention: Steven J. Tonsfeldt Fax No.: (650) 324-0638

If to Seller, addressed to:

Platinum Select, L.P. 325 N. St. Paul, Suite 4200 Dallas, TX 75201

Attention: Kristi Bomar Email: kbomar99@yahoo.com

With a copy (which shall not constitute notice) to:

Baker & McKenzie LLP 2300 Trammell Crow Center 2001 Ross Avenue Dallas, Texas 75201 Attention: Ted Schweinfurth

Attention: Ted Schweinfurth Fax No.: (214) 965-5914

If to the Partners, addressed to such Partner:

Patrick Aunkst 2318 Throckmorton Dallas, TX 75219

Kristi Bomar 2650 Lakeforest Ct. Dallas, TX 75214

Stephanie Houston 6159 Belmont Ave. Dallas, TX 75214 Robert Quigley 1243 Waterside Circle Dallas, TX 75218 Lyle Seedig 1600 Harper Lane Carrollton, TX 75010

or to such other address or to such other person as either party shall have last designated by such notice to the other party (including, without limitation, a designated representative of the Partners).

- 11.11 Expenses and Attorneys Fees. Each party shall be responsible for its own expenses and attorneys fees incurred in negotiating, executing, preparing and delivering the Transaction Agreements, including, without limitation, all legal, accounting, broker, finder and financial advisor fees. Should suit be brought to enforce or interpret any part of this Agreement, the prevailing party shall be entitled to recover, as an element of the costs of suit and not as damages, reasonable attorneys' fees to be fixed by the court (including costs, expenses and fees on any appeal). The prevailing party shall be entitle to recover its costs of suit, regardless of whether the suit proceeds to final judgment.
- 11.12 <u>Specific Performance</u>. Seller and the Partners each acknowledge that, in view of the uniqueness of the Business and the transactions contemplated by this Agreement, each party would not have an adequate remedy at law for money damages in the event that this Agreement has not been performed in accordance with its terms, and therefore agrees that the other party shall be entitled to specific enforcement of the terms hereof in addition to any other remedy to which it may be entitled, at law or in equity.
- 11.13 <u>Guaranty</u>. Guarantor hereby unconditionally guarantees in favor of Seller and the Limited Partners the full performance by Buyer of each of its obligations relating to the payment of the Purchase Price (including, for avoidance of doubt, the Liability Payments, the Holdback Amount (if applicable) and the Net Working Capital Increase (if applicable)), the Earnout Amount (if any), the Healthcare Employee Payroll Amount and all Undetermined Commissions, each pursuant to the terms and conditions of this Agreement and the performance of Buyer's other obligations under this Agreement.

ARTICLE 12 DEFINITIONS

For all purposes of this Agreement, except as otherwise expressly provided,

- (a) the terms defined in this Article 12 have the meanings assigned to them in this Article 12 and include the plural as well as the singular,
- (b) all accounting terms not otherwise defined herein have the meanings assigned under GAAP,

- (c) all references in this Agreement to designated "Articles," "Sections" and other subdivisions are to the designated Articles, Sections and other subdivisions of the body of this Agreement,
 - (d) pronouns of either gender or neuter shall include, as appropriate, the other pronoun forms, and
- (e) the words "herein," "hereof" and "hereunder" and other words of similar import refer to this Agreement as a whole and not to any particular Article, Section or other subdivision.

As used in this Agreement and the exhibits and schedules delivered pursuant to this Agreement, the following definitions shall apply.

- "401(k) Matching Payment" shall mean any amount required to be paid by Seller as a matching contribution pursuant to the terms of Seller's 401(k) plan to any Seller Employee's account under Seller's 401(k) plan.
- "A/R Bank Account" means any account (including, for avoidance of doubt, "lock-box" and other deposit accounts) of Seller into which accounts receivable collections related to the Business and the services rendered by Seller in connection with the Business prior to the Closing Date or by Buyer in connection with the Business after the Closing Date are deposited.
- "Action" means any claim, complaint, action, petition, investigation, suit, mediation, arbitration or other proceeding, whether civil or criminal, in law or in equity, or before any arbitrator or Governmental Entity.
- "Affiliate" means a Person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, a specified Person.
- "Applicable Earnout Amount" means such portion of the Earnout Amount as shall be payable to Seller pursuant to Section 2.4 hereof and as calculated in accordance with the terms set forth on Schedule 2.4 hereto.
- "Approval" means any approval, authorization, consent, qualification or registration, or any waiver of any of the foregoing, required to be obtained from, or any notice, statement or other communication required to be filed with or delivered to, any Governmental Entity or any other Person.
 - $\begin{tabular}{l} {\bf ``Assignment\ of\ Intellectual\ Property''} means\ the\ Assignment\ of\ Intellectual\ Property\ in\ the\ form\ of\ \underline{Exhibit\ F}\ hereto. \end{tabular}$
 - "Assignment and Assumption of Office Lease" means the Assignment and Assumption of Office Lease agreement in the form of Exhibit G hereto.
 - "Assignment and Assumption of Office Sublease" means the Assignment and Assumption of Office Sublease agreement in the form of Exhibit H hereto.

- "Assignment and Assumption of Temporary Storage Agreement" means the Assignment and Assumption of Temporary Storage Agreement in the form of Exhibit I hereto.
 - "Assumed Seller Benefit Arrangements" means the Seller Benefit Arrangements identified as such on Schedule 1.1(h) hereto.
- "<u>Assumed Working Capital Obligations</u>" means those categorical working capital obligations of Seller set forth on <u>Schedule 12-1</u> hereto, each in an amount that does not exceed the amount of such obligation in Seller's ordinary course business consistent with past practices.
 - "Balance Sheet Date" means December 31, 2007.
 - "Bill of Sale and Assignment of Contracts" means the Bill of Sale and Assignment of Contracts agreement in the form of Exhibit J hereto.
- "Business" means the business of Seller, which includes recruiting and providing temporary and permanent healthcare professionals, placements, staffing, workforce management and educational services to healthcare providers and other healthcare organizations.
- "Business Day" means any day (A) other than Saturday or Sunday and (B) other than a day on which the United States federal government is closed for the observance of any holiday.
 - "Code" means the Internal Revenue Code of 1986, as amended, or as hereafter amended.
 - "Continuing Employees" mean those Seller Employees who become employed by Buyer in connection with the Closing.
- "Contract" means any contract, agreement, lease, sub-lease, license, sales order, purchase order, or other legally binding commitment or instrument, whether or not in writing.
- "Confidentiality, Non-Competition and Non-Solicitation Agreement" means the Confidentiality, Non-Competition and Non-Solicitation Agreement in the form of Exhibit K hereto.
 - $\hbox{$\stackrel{``}{\hbox{\it Corporate Employee}}$" means any current Seller Employee who is not a Healthcare Employee.}$
- "Customer" means any customer or client of the Business, including, without limitation, the individual healthcare professionals who contract with Seller to provide services at third-party healthcare facilities and the companies, organizations or associations that operate or manage the staffing of healthcare facilities where Seller staffs such healthcare professionals and any individual or entity whose name and/or contact information was collected, processed and/or stored through use of TKO software. Such individual healthcare professionals include, without limitation, radiology, oncology and cardiology technicians, physical, respiratory, occupational and speech therapists, nurses (including LVNs, LPNs and RNs), nurse practitioners, physician assistants, pharmacists and medical laboratory technologists.

- "Debt" of Seller means all indebtedness for borrowed money (excluding accounts payable, but including long-term debt and any current portion of long-term debt and any amounts outstanding under any line of credit), obligations with respect to capital leases and obligations or indebtedness related to the factoring of receivables, including, in each case any interest accrued thereon and any fees, charges or penalties payable by Seller in connection with the pre-payment or repayment of such indebtedness and/or obligations.
 - "Disclosure Schedule" means the Disclosure Schedule dated as of the date hereof, delivered by Seller and the Partners to Buyer and attached hereto.
 - "Disputed Mark" means the Trademark PLATINUM HEALTHCARE STAFFING, including standard format and/or with design.
- "<u>Disregarded Disclosures</u>" means the statements and information appearing in the section (or subsection) of the Disclosure Schedule that corresponds to such section (or subsection) of Article 3 hereof and which are identified as "Disregarded Disclosures."
 - "Domain Names" means all Internet domain name registrations and applications therefor.
 - "Earnout Amount" means an amount up to \$17,445,000, payable to Seller, if at all, pursuant to and in accordance with the provisions of Section 2.4 hereof.
- "<u>Earnout Period EBITDA</u>" means net income before interest, tax, depreciation and amortization attributable to the Assets and the Business as operated by Buyer during the Earnout Period as determined in accordance with GAAP.
- "<u>Earnout Period Total Revenue</u>" means the total revenue generated by the Assets and the Business as operated by Buyer during the Earnout Period as determined by Buyer pursuant to Section 2.4 hereof.
- "EBITDA" means net income before interest, tax, depreciation and amortization of Seller (or of the Business if relating to the Earnout Period), as determined in accordance with GAAP.
- "<u>Encumbrance</u>" means any claim, charge, easement, encumbrance, lease, covenant, security interest, lien, option, pledge, rights of others, or restriction (whether on voting, sale, transfer, disposition or otherwise), whether imposed by agreement, understanding, law, equity or otherwise.
 - "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.
- "ERISA Affiliate" means any entity which is a member of (A) a "controlled group of corporations," as defined in Section 414(b) of the Code, (B) a group of entities under "common control," as defined in Section 414(c) of the Code, or (C) an "affiliated service group," as defined in Section 414(m) of the Code, or treasury regulations promulgated under Section 414(o) of the Code, any of which includes Seller.
 - "GAAP" means generally accepted accounting principles in the United States, as in effect from time to time.

"Governmental Entity" means any government or any agency, district, bureau, board, commission, court, department, official, political subdivision, tribunal or other instrumentality of any government (including quasi-governmental agencies), whether federal, state, local, provincial, domestic or foreign, including, without limitation, the United States Trademark Trial and Appeal Board.

"Guarantor Material Contract" means any "material contract," within the meaning of Item 601(b)(10) of Regulation S-K under the Securities Exchange Act of 1934, as amended, with respect to Guarantor.

"<u>Healthcare Employee</u>" means any Seller Employee who is a healthcare professional who provides services at third-party healthcare facilities. Such Persons include, without limitation, radiology, oncology and cardiology technicians, physical, respiratory, occupational and speech therapists, nurses, nurse practitioners, physician assistants, pharmacists and medical laboratory technologists.

"Healthcare Employee Payroll Amount" means the aggregate dollar amount of Seller's Healthcare Employee payroll balance (including any related employer withholding obligations), accrued in the ordinary course of Seller's business consistent with past practice during the four (4) day period of February 11, 2008 through February 14, 2008. For avoidance of doubt, the Healthcare Employee Payroll Amount shall not include any amount due and payable to any Corporate Employee.

"Knowledge of Seller," or words of similar import, means the knowledge of a particular fact, circumstance, event or other matter in question by any individual who is an officer, manager or Partner of Seller (collectively, the "Seller Representatives"). Any such Seller Representative will be deemed to have knowledge of a particular fact, circumstance, event or other matter if (a) such Seller Representative has actual knowledge of the fact, circumstance, event or other matter in question or (b) such Seller Representative could obtain such knowledge of the fact, circumstance, event or other matter in question by reasonable inquiry under the circumstances.

"Law" means any constitutional provision, statute or other law, rule, regulation, or interpretation of any Governmental Entity and any Order.

"<u>Legal Requirements</u>" means any United States federal, state, municipal or local or foreign order, judgment, writ, injunction, decree, law, statute, standard ordinance, code, resolution, promulgation, rule, regulation or any similar provision having the force or effect of law.

"<u>Limited Partnership Agreement</u>" means the Amended and Restated Agreement of Limited Partnership of Platinum Select, L.P., dated as of July 6, 2004, by and among the General Partner and each of the Limited Partners.

"Logo" means the Seller Mark set forth on Schedule 12-2 hereto.

"Loss" means any action, cost, damage, disbursement, expense, liability, loss, deficiency, obligation, penalty or settlement of any kind or nature, whether foreseeable or unforeseeable (including, without limitation, interest or other carrying costs, penalties, reasonable legal, accounting and other professional fees and expenses) actually incurred or properly accrued in accordance with GAAP in the investigation, collection, prosecution and defense of claims and amounts paid in settlement, that are imposed on or otherwise incurred or suffered by the specified person.

- "Material Adverse Change" and "Material Adverse Effect" when used in connection with an entity means any change, event, circumstance, condition or effect that is or is reasonably likely to be, individually or in the aggregate, materially adverse in relation to the condition (financial or otherwise), capitalization, properties, products, prospects, assets (including intangible assets), Intellectual Property, liabilities, business, operations or results of operations of such entity and its subsidiaries, taken as a whole.
- "Net Working Capital" means the dollar value as of a particular date of (a) the sum of Seller's (i) cash, (ii) accounts receivable, (iii) accrued revenue and (iv) prepaid expenses, less (b) the sum of Seller's (i) accounts payable and (ii) accrued expenses.
- "Net Working Capital Decrease" means the amount by which the Net Working Capital is less than the sum of (a) seven million, three hundred thousand dollars (\$7,300,000) minus (b) \$25,000.
- "Net Working Capital Increase" means the amount by which the Net Working Capital is greater than the sum of (a) seven million, three hundred thousand dollars (\$7,300,000) plus (b) \$25,000.
 - "Order" means any decree, injunction, judgment, order, ruling, assessment or writ.
- "Permit" means any license, permit, franchise, certificate of authority, or order, or any waiver of the foregoing, required to be issued by any Governmental Entity.
 - "Permitted Encumbrance" means liens for current Taxes not yet due and payable and any Encumbrance specifically identified on Schedule 12-3 hereto.
- "Person" means an association, a corporation, an individual, a partnership, a trust, a limited liability company or any other entity or organization, including a Governmental Entity.
 - "Primary Mark 'A'" means the Seller Mark PLATINUM SELECT, including standard format and/or with design.
 - "Primary Mark 'B'" means the Seller Mark PLATINUM SELECT STAFFING, including standard format and/or with design.
 - "Primary Marks" means, collectively, Primary Mark "A," Primary Mark "B" and the Logo.
 - "Pro Rata Interest" means each Partner's interest in Seller (expressed as a percentage set forth on Exhibit A to the Limited Partnership Agreement).

"Seller Balance Sheet" means Seller's unaudited balance sheet as of the Balance Sheet Date.

"Seller Benefit Arrangement" means each "employee benefit plan" (within the meaning of Section 3(3) of ERISA, including, without limitation, multiemployer plans within the meaning of Section 3(37) of ERISA), and all equity purchase, option to acquire equity, severance, employment, change-incontrol, fringe benefit, vacation, personal time off, collective bargaining, bonus, incentive, deferred compensation, employee loan and all other employee benefit plans, Contracts, programs, policies or other arrangements, whether or not subject to ERISA, whether formal or informal, oral or written, legally binding or not, under which (i) any Seller Employee has any present or future right to benefits and which are contributed to, sponsored by or maintained by Seller or (ii) Seller has any present or future liability.

"Seller Closing Debt" means the Debt of Seller as of the Closing Date other than the Assumed Working Capital Obligations.

"Seller Contracts Workbook" means, collectively, the electronic files of Seller containing a complete list of all Contracts pursuant to which Seller has any continuing obligation to any third party as of the Closing Date or pursuant to which any third party has any continuing obligation to Seller as of the Closing Date, which contracts are copied onto a compact disk for delivery to Buyer

"Seller Employee" means any current or former Partner, officer, manager, employee, consultant or other independent contractor (including, without limitation, healthcare professionals) of Seller.

"Seller Financial Condition Certificate" means a certificate executed by the Chief Financial Officer of Seller, Aunkst and Bomar, dated as of the Closing Date, certifying (A) that the Year End Financial Statements attached to such Seller Financial Condition Certificate are true, correct and complete as of December 31, 2007, (B) the amount of the Seller Closing Debt, (C) the amount of Seller as of December 31, 2007 (as reflected in the Year End Financial Statements), and (D) an itemization of all of the Transaction Expenses paid or payable at any time prior to, at or following the Closing Date, and the amounts of such Transaction Expenses that remain unpaid by Seller immediately prior to the Closing, it being the expressed intent of Buyer, Seller and the Partners that to the maximum extent possible all unpaid Transaction Expenses be deducted from the Purchase Price.

"Seller Financial Statements" means (A) Seller's unaudited balance sheets dated December 31, 2005 and December 31, 2006, (B) Seller's unaudited statements of income and statements of cash flows for the years ended December 31, 2005 and December 31, 2006, and (C) the Year End Financial Statements.

"<u>Software</u>" means all computer programs (whether in source code or object code form), data bases, compilers, compilations, software libraries, source code annotations, software architecture designs, layouts and development tools and the programmers' notes or logs, user, operator and training manuals, and instructions related to any of the foregoing.

"<u>Tangible Personal Property</u>" means all machinery, equipment, tools, furniture, office equipment, computer hardware, supplies, materials, vehicles and other items of tangible personal property of every kind owned or leased by Seller (wherever located and whether or not carried on Seller's books), together with any express or implied warranty by the manufacturers or sellers or lessors of any item or component part thereof and all maintenance records and other documents relating thereto.

"Tax" (and, with correlative meaning, "Taxes") means (A) all federal, state, local or foreign taxes, charges, fees, imposts, levies or other assessments, including, without limitation, all income, gross receipts, capital, sales, use, ad valorem, value added, transfer, franchise, profits, inventory, license, withholding, payroll, employment, social security, unemployment, excise, severance, stamp, occupation, property and estimated taxes, customs, duties, fees, additions to Tax or additional amounts (whether disputed or not) imposed by a Tax authority, and (B) any liability for the payment of any amounts of the type described in clause (A) of this sentence as a result of being a transferee of or successor to any Person or as a result of any express or implied obligation to indemnify any other Person.

"Tax Return" means a declaration, statement, report, return or other document or information required to be filed or supplied with respect to Taxes.

"Territory" means the United States and its territories.

"<u>Trademarks</u>" means all United States and foreign trademarks, service marks, names, trade names, words, designs, logos, trade dress, slogans, symbols, and combinations of the same, and any other general intangibles of like nature, used or intended to be used to identify, distinguish or indicate the source of goods or services, together with all goodwill appurtenant thereto, and all rights to apply for or register, applications for, and registrations of, any of the foregoing.

"Trade Secrets" means all United States and foreign trade secrets and other proprietary or confidential information or technology, in any format, whether tangible or intangible, whether patentable or unpatentable, and whether or not reduced to practice, including, without limitation, confidential business information, know-how, research and development information and data, experimental results and data, tests and all other tangible or intangible proprietary information, financial, marketing and business data, pricing and cost information, business and marketing plans and Customer and supplier lists and information. For example and without any limitation, Trade Secrets includes all subject matter that derives independent economic value, actual or potential, from not being generally known to, and not readily ascertainable without improper means by, other parties.

"<u>Transaction Documents</u>" means this Agreement, the Employment Agreement, each Partner Independent Consultant Agreement, the Bill of Sale and Assignment of Contract Rights, the Assignment and Assumption of Office Lease, the Assignment and Assumption of Office Sublease, the Assignment of Intellectual Property and each Confidentiality, Non-Competition and Non-Solicitation Agreement.

"<u>Transaction Expenses</u>" means all out-of-pocket costs and expenses incurred by, or to be paid or reimbursed by, Seller in connection with this Agreement and the Transactions (including any fees and expenses of legal counsel, financial advisors, investment bankers and accountants).

"Transactions" means the transactions contemplated by the Transaction Documents.

"<u>Undetermined Commission</u>" means the dollar amount of any commission, benchmark bonus or perm-placement bonus actually payable to a Seller Employee pursuant to a Seller Benefit Arrangement, which amounts shall not exceed those set forth opposite each such Seller Employee's name on <u>Schedule 5.8(b)</u> hereto.

"Year End Financial Statements" means Seller's (A) unaudited balance sheet dated December 31, 2007 and (B) unaudited statement of income and statement of cash flows for the year ended December 31, 2007.

INDEX OF OTHER DEFINED TERMS

Defined Term	Section Reference
A/R Bank Accounts	Section 5.4(e)
Agreement	Introduction
Assets	Section 1.1
Assumed Liabilities	Section 1.3
Aunkst	Introduction
Balance Sheet Dispute Notice	Section 2.3
Bank of Texas	Section 6.3
Basket	Section 9.3(a)
Bomar	Introduction
Buyer	Introduction
Buyer Indemnified Party	Section 9.1
Buyer Indemnifying Party	Section 9.2
Claim	Section 9.4(a)
Claims Period	Section 9.4(e)
Closing	Section 2.1
Closing Balance Sheet	Section 2.3
Closing Balance Sheet Certificate	Section 2.3
Closing Date	Section 2.1
COBRA	Section 3.17(i)
Confirmation Certificate	Section 2.3
Contested Claim	Section 9.7(c)
Disputed Mark	Section 5.4(c)
Earnout Period	Section 2.4(a)
Employment Agreement	Recitals
Excluded Assets	Section 1.2
Excluded Liabilities	Section 1.4
Expiration Date	Section 8.1
Facility Contract	Section 3.8
General Partner	Introduction

Defined Term	Section Reference
Guarantor	Introduction
Hazardous Materials	Section 3.19
Holdback Amount	Section 2.2(c)
Houston	Introduction
Indemnified Parties	Section 9.2
Indemnifying Parties	Section 9.2
Independent Accountant	Section 2.3
Initial Cash Payment	Section 2.2(a)
Intellectual Property	Section 1.1(a)
Intellectual Property Rights	Section 5.4(a)
JAMS	Section 9.7(d)
Liability Payments	Section 2.2(b)
Limited Partner	Introduction
Material Contract	Section 3.8
Mutual NDA	Section 5.1
Notice of Claim	Section 9.4(b)
Open Source Materials	Section 3.7(n)
Partner Guarantees	Section 5.11
Partners	Introduction
Partner Independent Consultant Agreement	Section 6.6
Pay-Off Letter	Section 6.3
PL Insurance Policy Tail	Section 6.8
Purchase Price	Section 2.2
Purchase Price Allocation Schedule	Section 2.5(b)
Quigley	Introduction
Rights of Set-Off	Section 2.4(c)
Seedig	Introduction
Seller	Introduction
Seller Expenses	Section 2.4(b)(ii)
Seller Indemnified Party	Section 9.2
Seller Indemnifying Party	Section 9.1
Seller License	Section 3.7(b)
Seller Mark	Section 3.7(d)
Operating and Integration Plan	Section 2.4(d)
Seller Source Code	Section 3.7(f)
Statement No. 5	Section 3.10(b)
Third-Party Claim	Section 9.4(b)(ii)
Transfer Taxes	Section 2.5(a)
Uploaded Materials	Section 3.7(p)
WARN Act	Section 3.17(o)
	FC: () () 1
	1C:t f-111

[Signature page follows]

IN WITNESS WHEREOF, each of the parties hereto has executed this Asset Purchase Agreement, or caused this Asset Purchase Agreement to be executed by its duly authorized officers, as of the day and year first above written.

BUYER:

PLATINUM SELECT HEALTHCARE STAFFING, INC.

By: /s/ Susan R. Nowakowski

Name: Susan R. Nowakowski Title: Chief Executive Officer

GUARANTOR:

AMN HEALTHCARE, INC.

By: /s/ Susan R. Nowakowski

Name: Susan R. Nowakowski
Title: Chief Executive Officer

SELLER:

PLATINUM SELECT, L.P.

By: Platinum Select Management, L.L.C.,

its general partner

By: /s/ Patrick Aunkst
Name: Patrick Aunkst
Title: Owner - Manager

[Signature page to Asset Purchase Agreement. Additional signature page follows.]

PARTNERS:

PLATINUM SELECT MANAGEMENT, L.L.C.

By: /s/ Patrick Aunkst

Name: Patrick Aunkst

Owner - Manager

/s/ Patrick Aunkst

PATRICK AUNKST

/s/ Kristi Bomar

KRISTI BOMAR

/s/ Robert Quigley

ROBERT QUIGLEY

/s/ Stephanie Houston

STEPHANIE HOUSTON

/s/ Lyle Seedig

THIRD AMENDMENT TO SECOND AMENDED AND RESTATED CREDIT AGREEMENT

THIS THIRD AMENDMENT TO SECOND AMENDED AND RESTATED CREDIT AGREEMENT (this "<u>Amendment</u>"), dated as of April 28, 2008, is entered into among **AMN HEALTHCARE**, **INC.**, a Nevada corporation (the "<u>Borrower</u>"), **AMN HEALTHCARE SERVICES**, **INC.**, a Delaware corporation (the "<u>Parent</u>"), the Subsidiary Guarantors from time to time parties thereto, the lenders identified on the signature pages hereto (the "<u>Lenders</u>") and **BANK OF AMERICA**, **N.A.**, as Administrative Agent.

WITNESSETH

WHEREAS, the Borrower, the Parent, the Subsidiary Guarantors, the Lenders party thereto, and the Administrative Agent entered into that certain Second Amended and Restated Credit Agreement dated as of November 2, 2005, as amended by that certain First Amendment to Second Amended and Restated Credit Agreement dated as of May 1, 2006 and as amended by that certain Second Amendment to Second Amended and Restated Credit Agreement dated as of July 6, 2007 (the "Existing Credit Agreement");

WHEREAS, the Borrower has requested that the Lenders amend certain provisions of the Existing Credit Agreement;

WHEREAS, the Lenders have agreed to amend the Existing Credit Agreement in accordance with such request and as provided herein; and

NOW, THEREFORE, in consideration of the agreements hereinafter set forth, and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the parties hereto agree as follows:

PART 1 DEFINITIONS

SUBPART 1.1 Certain Definitions. Unless otherwise defined herein or the context otherwise requires, the following terms used in this Amendment, including its preamble and recitals, have the following meanings:

"Amended Credit Agreement" means the Existing Credit Agreement as amended hereby.

"Third Amendment" has the meaning set forth in Subpart 3.1.

SUBPART 1.2 Other Definitions. Unless otherwise defined herein or the context otherwise requires, terms used in this Amendment, including its preamble and recitals, have the meanings provided in the Existing Credit Agreement.

PART 2 AMENDMENTS TO EXISTING CREDIT AGREEMENT

Effective on (and subject to the occurrence of) the Third Amendment Effective Date, the Existing Credit Agreement is hereby amended in accordance with this Part II.

- **SUBPART 2.1** <u>Amendment to Section 1.1</u>. Subsection (viii) of the definition of "Permitted Lien" found in Section 1.1 is hereby amended as follows: (viii) Liens securing Indebtedness permitted by <u>Sections 8.1(g)</u> and <u>8.1(k)</u>;
- **SUBPART 2.2** Amendment to Section 8.1. Section 8.1 is hereby amended by (i) deleting the "and" before existing clause (k), (ii) replacing "(k)" in
- **SUBPART 2.2** <u>Amendment to Section 8.1</u>. Section 8.1 is hereby amended by (i) deleting the "and" before existing clause (k), (ii) replacing "(k)" in existing clause (k) with "(l)" and (iii) adding the following as new clause (k):
 - (k) other secured Indebtedness of the Borrower or any of its Subsidiaries in an aggregate principal amount not to exceed \$15,000,000 at any one time outstanding; provided that any such secured Indebtedness shall be subject to an intercreditor agreement on terms and conditions reasonably acceptable to the Administrative Agent; and
- **SUBPART 2.3** Amendment to Section 8.7. Subsection (k) of Section 8.7 of the Existing Credit Agreement is hereby amended in its entirety to read as follows:
 - (k) the repurchase, redemption or other acquisition or retirement for value of any Capital Stock of the Parent or any option to acquire Capital Stock of the Parent in an aggregate cash amount not to exceed \$50,000,000 in the aggregate; provided that (x) no Default or Event of Default exists either before or after giving effect to such Restricted Payment and (y) such repurchase, redemption or other acquisition or retirement occurs on or before March 31, 2009, and
 - SUBPART 2.4 Amendment to Section 8.14. Section 8.14 of the Existing Credit Agreement is hereby amended in its entirety to read as follows:

8.14 Capital Expenditures.

The Credit Parties will not permit Consolidated Capital Expenditures for any fiscal year to exceed \$15,000,000 plus the unused amount available for Consolidated Capital Expenditures under this Section 8.14 for the immediately preceding fiscal year (excluding any carry forward available from any prior fiscal year).

PART 3 CONDITIONS TO EFFECTIVENESS

SUBPART 3.1 Third Amendment Effective Date. This Amendment shall be and become effective as of the date hereof when all of the conditions set forth in this Part 3 shall have been satisfied, and thereafter this Amendment shall be known, and may be referred to, as the "Third Amendment".

SUBPART 3.2 Execution of Counterparts of Amendment. The Administrative Agent shall have received counterparts of this Amendment, which collectively shall have been duly executed on behalf of each of the Borrower, the Parent, the Subsidiary Guarantors, the Requisite Lenders, Lenders (other than Defaulting Lenders) holding in the aggregate at least a majority of the Revolving Commitments and the Administrative Agent.

SUBPART 3.3 Fees and Expenses. The Administrative Agent shall have received from the Borrower (i) the aggregate amount of fees and expenses payable to the Administrative Agent and Banc of America Securities LLC in connection with the consummation of the transactions contemplated hereby, (ii) for the account of each Lender approving this Amendment by 12:00 p.m. Eastern Standard Time on April 25, 2008, an amendment fee equal to 25 basis points on such Lender's Commitment and (iii) all reasonable and documented out-of-pocket costs and expenses of the Administrative Agent in connection with the preparation, execution and delivery of this Amendment, including without limitation the reasonable fees and expenses of Moore & Van Allen PLLC, special counsel to the Administrative Agent.

PART 4 MISCELLANEOUS

SUBPART 4.1 Representations and Warranties. The Borrower hereby represents and warrants to the Administrative Agent and the Lenders that, after giving effect to this Amendment, (a) no Default or Event of Default exists under the Existing Credit Agreement, after giving effect to this Amendment and (b) the representations and warranties set forth in Section 6 of the Existing Credit Agreement are, subject to the limitations set forth therein, true and correct in all material respects as of the date hereof (except for those which expressly relate to an earlier date, in which case, they are true and correct in all material respects as of such earlier date).

SUBPART 4.2 Cross-References. References in this Amendment to any Part or Subpart are, unless otherwise specified, to such Part or Subpart of this Amendment.

SUBPART 4.3 <u>Instrument Pursuant to Existing Credit Agreement</u>. This Amendment is executed pursuant to the Existing Credit Agreement and shall (unless otherwise expressly indicated therein) be construed, administered and applied in accordance with the terms and provisions of the Existing Credit Agreement.

SUBPART 4.4 References in Other Credit Documents. At such time as this Amendment shall become effective pursuant to the terms of Subpart 3.1, all references to the "Credit Agreement" shall be deemed to refer to the Existing Credit Agreement as amended by this Amendment.

SUBPART 4.5 Counterparts. This Amendment may be executed by the parties hereto in several counterparts, each of which shall be deemed to be an original and all of which shall constitute together but one and the same agreement. Delivery of executed counterparts of the Amendment by facsimile or electronic mail shall be effective as an original and shall constitute a representation that an original shall be delivered.

SUBPART 4.6 Governing Law. THIS AMENDMENT SHALL BE DEEMED TO BE A CONTRACT MADE UNDER AND GOVERNED BY THE LAWS OF THE STATE OF NEW YORK.

SUBPART 4.7 Acknowledgment. The Guarantors acknowledge and consent to all of the terms and conditions of this Amendment and agree that this Amendment does not operate to reduce or discharge the Guarantors' obligations under the Existing Credit Agreement as amended by this Amendment or the other Credit Documents. The Guarantors further acknowledge and agree that the Guarantors have no claims, counterclaims, offsets, or defenses to the Credit Documents and the performance of the Guarantors' obligations thereunder or if the Guarantors did have any such claims, counterclaims, offsets or defenses to the Credit Documents or any transaction related to the Credit Documents, the same are hereby waived, relinquished and released in consideration of the Lenders' execution and delivery of this Amendment. Each Guarantor also hereby confirms and agrees that notwithstanding the effectiveness of this Amendment, the Collateral Documents to which each of the undersigned is a party and all of the Collateral described therein do, and shall continue to, secure the payment of all of the Credit Party Obligations.

SUBPART 4.8 Binding Effect. This Amendment, the Existing Credit Agreement as amended by this Amendment and the other Credit Documents embody the entire agreement between the parties and supersede all prior agreements and understandings, if any, relating to the subject matter hereof. These Credit Documents represent the final agreement between the parties and may not be contradicted by evidence of prior, contemporaneous or subsequent oral agreements of the parties. Except as expressly modified and amended in this Amendment, all the terms, provisions and conditions of the Credit Documents shall remain unchanged and shall continue in full force and effect.

SUBPART 4.9 <u>Successors and Assigns</u>. This Amendment shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns.

SUBPART 4.10 General. Except as amended hereby, the Existing Credit Agreement and all other credit documents shall continue in full force and effect.

SUBPART 4.11 Severability. If any provision of this Amendment is determined to be illegal, invalid or unenforceable, such provision shall be fully severable and the remaining provisions shall remain in full force and effect and shall be construed without giving effect to the illegal, invalid or unenforceable provisions.

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, the parties hereto have executed this Amendment to the Credit Agreement as of the date first above written.

By:

BORROWER: AM

AMN HEALTHCARE, INC.

/s/ David C. Dreyer

Name: David C. Dreyer
Title: Chief Financial Officer

<u>PARENT</u>:

AMN HEALTHCARE SERVICES, INC.

By: /s/ David C. Dreyer

By: /s/ David C. Dreyer

Name: David C. Dreyer

Title: Chief Financial Officer

SUBSIDIARY
GUARANTORS:

AMN SERVICES, INC. (formerly known as Worldview Healthcare, Inc.)

By: /s/ David C. Dreyer
Name: David C. Dreyer
Title: Chief Financial Officer

O'GRADY-PEYTON INTERNATIONAL (USA), INC.

By: /s/ David C. Dreyer
Name: David C. Dreyer
Title: Chief Financial Officer

INTERNATIONAL HEALTHCARE RECRUITERS, INC.

By: /s/ David C. Dreyer
Name: David C. Dreyer
Title: Chief Financial Officer

AMN STAFFING SERVICES, INC.

By: /s/ David C. Dreyer
Name: David C. Dreyer
Title: Chief Financial Officer

THE MHA GROUP, INC.

By: /s/ David C. Dreyer
Name: David C. Dreyer
Title: Chief Financial Officer

MERRITT, HAWKINS & ASSOCIATES

By: /s/ David C. Dreyer
Name: David C. Dreyer
Title: Chief Financial Officer

MED TRAVELERS, INC.

By: /s/ David C. Dreyer
Name: David C. Dreyer
Title: Chief Financial Officer

RN DEMAND, INC.

By: /s/ David C. Dreyer
Name: David C. Dreyer
Title: Chief Financial Officer

STAFF CARE, INC.

By: /s/ David C. Dreyer
Name: David C. Dreyer
Title: Chief Financial Officer

MHA ALLIED CONSULTING, INC.

By: /s/ David C. Dreyer
Name: David C. Dreyer
Title: Chief Financial Officer

MED TRAVELERS, LLC

By: /s/ David C. Dreyer
Name: David C. Dreyer
Title: Chief Financial Officer

LIFEWORK, INC.

By: /s/ David C. Dreyer
Name: David C. Dreyer
Title: Chief Financial Officer

PHARMACY CHOICE, INC.

By: /s/ David C. Dreyer
Name: David C. Dreyer
Title: Chief Financial Officer

RX PRO HEALTH, INC.

By: /s/ David C. Dreyer
Name: David C. Dreyer
Title: Chief Financial Officer

PLATINUM SELECT HEALTHCARE STAFFING, INC.

By: /s/ David C. Dreyer
Name: David C. Dreyer
Title: Chief Financial Officer

[Signatures Continued]

ADMINISTRATIVE AGENT:

BANK OF AMERICA, N. A.,

in its capacity as Administrative Agent

By: /s/ Alysa Trakas
Name: Alysa Trakas
Title: Vice President

LENDERS:

BANK OF AMERICA, N. A.,

in its capacity as Lender, Issuing Lender and Swingline Lender

By: /s/ Alysa Trakas
Name: Alysa Trakas
Title: Vice President

[Signatures Continued]

Certification Pursuant To Rule 13a-14(a) of the Securities Exchange Act of 1934

- I, Susan R. Nowakowski, certify that:
 - 1. I have reviewed this report on Form 10-Q of AMN Healthcare Services, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting;
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 9, 2008

/s/ SUSAN R. NOWAKOWSKI

Name: Title:

Susan R. Nowakowski President and Chief Executive Officer (Principal Executive Officer)

Certification Pursuant To Rule 13a-14(a) of the Securities Exchange Act of 1934

I, David C. Dreyer, certify that:

- 1. I have reviewed this report on Form 10-Q of AMN Healthcare Services, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting;
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 9, 2008

/s/ DAVID C. DREYER

Name: Title:

David C. Dreyer Chief Accounting Officer, Chief Financial Officer and Treasurer (Principal Accounting and Financial Officer)

AMN Healthcare Services, Inc.

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of AMN Healthcare Services, Inc. (the "Company") on Form 10-Q for the period ended March 31, 2008 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Susan R. Nowakowski, President and Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 9, 2008

/s/ Susan R. Nowakowski

Susan R. Nowakowski President and Chief Executive Officer (Principal Executive Officer)

AMN Healthcare Services, Inc.

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of AMN Healthcare Services, Inc. (the "Company") on Form 10-Q for the period ended March 31, 2008 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, David C. Dreyer, Chief Accounting Officer and Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 9, 2008

/s/ DAVID C. DREYER

David C. Dreyer
Chief Accounting Officer,
Chief Financial Officer and Treasurer
(Principal Accounting and Financial Officer)